

Review

Contents lists available at ScienceDirect

## Biomedicine & Pharmacotherapy



journal homepage: www.elsevier.com/locate/biopha

# Therapeutic potential and limitations of curcumin as antimetastatic agent

Petr Dytrych<sup>a</sup>, Zdeněk Kejík<sup>b,c</sup>, Jan Hajduch<sup>b</sup>, Robert Kaplánek<sup>b</sup>, Kateřina Veselá<sup>b,c</sup>, Kateřina Kučnirová<sup>b,c</sup>, Markéta Skaličková<sup>b,c</sup>, Anna Venhauerová<sup>b,c</sup>, David Hoskovec<sup>a</sup>, Pavel Martásek<sup>c,\*</sup>, Milan Jakubek<sup>b,c,\*\*</sup>

<sup>a</sup> 1st Department of Surgery-Department of Abdominal, Thoracic Surgery and Traumatology, First Faculty of Medicine, Charles University and General University Hospital, U Nemocnice 2, 121 08 Prague, Czech Republic

<sup>b</sup> BIOCEV, First Faculty of Medicine, Charles University, Průmyslová 595, 252 50 Vestec, Czech Republic

<sup>c</sup> Department of Paediatrics and Inherited Metabolic Disorders, First Faculty of Medicine, Charles University and General University Hospital in Prague, Ke Karlovu 455/

2, 128 08 Prague, Czech Republic

ARTICLE INFO

Keywords: Curcumin Metastasis Migrastatic agents Tumour targeting Nanoparticles

## ABSTRACT

Treatment of metastatic cancer is one of the biggest challenges in anticancer therapy. Curcumin is interesting nature polyphenolic compound with unique biological and medicinal effects, including repression of metastases. High impact studies imply that curcumin can modulate the immune system, independently target various metastatic signalling pathways, and repress migration and invasiveness of cancer cells. This review discusses the potential of curcumin as an antimetastatic agent and describes potential mechanisms of its antimetastatic activity. In addition, possible strategies (curcumin formulation, optimization of the method of administration and modification of its structure motif) to overcome its limitation such as low solubility and bioactivity are also presented. These strategies are discussed in the context of clinical trials and relevant biological studies.

## 1. Introduction

Treatment of metastases is a major challenge in cancer therapy. It was proved that most deaths in oncology patients are not caused by the primary tumours but by their metastasis [1]. Nevertheless, therapeutic strategy of classical neoadjuvant therapeutic regimens is primarily focused on the reduction of tumour mass by using cytostatic drugs and less focused on the repression of metastasis [2]. In the present time, a new therapeutic method that targets metastasis formation via the use of migrastatic drugs (inhibitors of cell migration) instead of reducing the tumour mass are intensively studied and developed [3,4]. In the case of nonmetastatic castration-resistant prostate cancer, some androgen receptor inhibitors (e.g., apalutamide, enzalutamide, and darolutamide) can delay metastases formation [5]. Besides, the application of some therapeutic agents such as tyrosine kinase inhibitors and immuno-checkpoint inhibitors or inhibitors of IL-6 signalling could be used for the metastasis suppression. [6-9] Gkountela et al. published that Na<sup>+</sup>/K<sup>+</sup> ATPase inhibitors ouabain and digitoxin (FDA approved agents) induce dissociation of circulating tumour cell (CTC) clusters into single cells and thereby repress metastatic formation [10]. After entry

into the bloodstream, CTCs (known as inherent metastatic factors) can infiltrate healthy tissues that are distant from the tumour [11]. Nevertheless, CTC clusters display metastatic potential of 20- to 100-fold greater metastatic potential than single CTCs [12–14]. Other possible targets of migrastatic agents are mechanisms of cell mobility (e.g., cytoskeletal dynamics and cell contractility) and energy provisions (e.g., ATP availability, mitochondrial metabolism) [15].

However, oncological diseases display high heterogeneity and oncogenic signalling pathways, or factors are at least partially substitutability in the metastasis formation [16,17]. Which both lead to a loss of response to the therapy. However, high impact works suggest, that multifunctional agents have a better chance of avoiding the development of resistance than focusing on the single molecular target [18–21].

Curcumin (polyphenol product turmeric from Curcumin longa) [22] and other natural and synthetic curcuminoids display many antitumour and antimetastatic effects. Structure of curcumin and other natural turmeric curcuminoids with similar anticancer effects is showed on the Fig. 1.

Curcumin targets activity/expression of oncogenic signalling pathways and factors (e.g., NF- $\kappa$ B, EGFR, PI3K/Akt/mTOR, Wnt/ $\beta$ -catenin,

https://doi.org/10.1016/j.biopha.2023.114758

Received 21 March 2023; Received in revised form 18 April 2023; Accepted 20 April 2023

<sup>\*</sup> Corresponding author.

<sup>\*\*</sup> Corresponding author at: BIOCEV, First Faculty of Medicine, Charles University, Průmyslová 595, 252 50 Vestec, Czech Republic. *E-mail addresses:* pavel.martasek@lfl.cuni.cz (P. Martásek), milan.jakubek@lfl.cuni.cz (M. Jakubek).

JAK/STAT-3, Hedgehog and Hippo signalling pathways, MAPK, AR, IL-6, IL-8, TNF- $\alpha$ , IL-1 $\beta$ , VEGF) [23–28]. It was proved, that curcumin application decreases migration and invasiveness of cancer cells, quantity of stem cancer cells, angiogenesis, recruitment and functionality of tumour associated cells, tumour development and metastatic activity. Its possible effect on the tumour growth and development are showed on the Fig. 2.

The clinical potential of curcumin is shortly introduced below. Section 2 summarizes results from in vitro and in vivo studies of curcumin with emphasis on its antimetastatic effects. Section 3 discusses possible approaches used in the curcumin formulation with emphasis on clinical trials. Section 4 is focused on the usability of the various strategies of curcumin administration such as oral, inhalation and intratumoural routes. Section 5 summarizes limitations of curcumin application and provides possible solutions to overcome them. Section 6 introduces studied strategies for improvement of curcumin antimetastatic efficacy (e.g., advanced nanoformulation and structural derivatization) and discusses its possible effect on the intratumoural microbiota.

In clinical trials, it was observed that curcumin induces the conversion of Treg cells into Th1 cells and increases IFN- $\gamma$  level [29,30]. In oncological patients, Treg cell levels can be correlated with the abundance of CTCs, and higher CD8 + T cell and IFN- $\gamma$  levels can induce a decrease in CTC count [31,32]. The application of curcumin in combination with other nutrients (garlic, green tea, grape seed extract, modified citrus pectin, and medicinal mushroom) decreased CTC count in patients with oncological diseases [33].

Curcumin is also a very suitable agent for boosting other used therapeutic modalities [27,34,35]. For example, NF- $\kappa$ B (strongly repressed by curcumin) is deeply associated with metastatic activity and resistance against therapeutic modalities such as chemotherapy, radiotherapy, immunotherapy, and photodynamic therapy [36–39]. Long-term use of anticancer drugs can induce multidrug resistance and limit therapeutic efficiency. Repression of drug resistance by curcumin application was observed in many high impact studies representing various types of serious cancers including liver, pancreatic, lung, cervical, prostate, breast cancers, leukemic diseases and so on. Besides, curcumin in combination with multiple anticancer drugs (e.g., doxorubicin, 5-fluorouracil, paclitaxel, berberine, docetaxel, metformin, gemcitabine) displays potent synergic effects in the inhibition of proliferation, invasion and metastasis [40].

In addition, curcumin toxicity is selective for cancer cells and tissue and displays protective effects against healthy tissue and normal cells. Its applications display only mild side effects and can alleviate side effects caused by other therapeutic methods. For example, its incorporation with chemotherapy and radiotherapy can strongly improve quality of life of patients with solid tumours [41].

# 2. Curcumins antimetastatic effects: results from in vitro and in vivo studies

Metastatic formation is a complex process and curcuminoids can

suppress its individual parts. Firstly, cancer cells change their cellular phenotype (from epithelial to mesenchymal, called epithelialmesenchymal transition). This process is supported by tumour associated cells (TAM M2 phenotype, CAFs, and TECs) which cause polarity loss and cell/matrix adhesion of cancer cells, and aids in the digestion of the extracellular matrix. Subsequently cancer cells travel up the bloodstream and are transported to the metastatic sites. In the bloodstream or at the site of metastasis, cancer cells revert their phenotype (mesenchymal-epithelial reverting transition (MErT)) and acquire increased cell adhesion and infiltrate the target tissue [42]. Primary tumours besides CTCs can liberate exosomes (nano-sized membranous structures liberated from the cells) into distant organs [43]. They can transport various factors (e.g., RNA, DNA, miRNA, and lipids) [44], which can increase the chance of acceptance of the cancer cell into target organs [43].

In the primary tumour, curcumin can supress EMT transition and migration of the cancer cells and recruitment of tumour associated cells. In the mice model, it was observed that curcumin repolarizes the phenotype of tumour associated macrophage from M2 to M1 (protumour to antitumour). [45,46] Suppression by curcumin in the cancer associated fibroblast phenotype ( $\alpha$ -SMA and vimentin) was associated with lower EMT and the migration of pancreatic cancer cells and lung metastasis in the mice model [47]. In the coculture tumour endothelial cells and colon cancer cells, curcumin supressed TEC transition, migration and phenotype [48]. In this case of the patients with colon and lung cancer, curcumin administration resulted in the conversion of the Treg cells (immune suppressive cells) into Th1 cells [29,30]. In the mice model with Ehrlich's carcinoma, this effect (induced by curcumin) was associated with the increase of IFN- $\gamma$ -secreting CD4 + and CD8 + T cells (antitumour) in the circulation and at the tumour site [49].

Curcumin could also effectively target CTC in the bloodstream. Mirza et al. reported that curcumin (10 µM) displayed potent cytotoxicity against circulating metastatic primary adenocarcinoma cells [50]. In comparison, gemcitabine (100 µM) did not display any significant cytotoxicity. In this section, it should be mentioned, that curcumin could also decrease levels of IL-8 [48,51-53]. Some high impact studies suggest, that higher IL-8 levels could be associated with higher activity and viability of CTC and IL-8 targeting could supress CTC spreading [54-56]. For example, Arnoletti et al. reported, that the CTC model derived from patients with pancreatic ductal adenocarcinoma expressed higher IL-8 RNA [56]. Application of IL-8 antibodies lead to impaired CTC cluster formations and increased CTC apoptosis. On the other hand, relevant clinical trials did not find any correlation between IL-8 and CTC levels in breast and prostate cancer patients [57,58]. However, triple negative breast cancer patients with liver metastasis against nonmetastatic ones display increase in CTC levels and inflammatory markers (IL-6, IL-8 and C-reactive protein) [59]. Similarly, in patients with osteosarcoma, IL-8 levels significantly correlated with the Enneking stage and metastasis [55]. In accordance with the presented hypothesis, in a mouse model, CTC implantation was strongly associated with increased IL-8 levels and its suppression of primary and metastatic tumours and CTC seeding. This



Fig. 1. Turmeric curcuminoid structure.



**Fig. 2.** Anti-tumour effect of curcumin. Curcumin display inhibition effects of proliferation cancer cells and induce their apoptosis. Curcumin represses angiogenesis via downregulation of VEGF. All these effects lead to decrease tumour growth and mass. Curcumin is also potent anti-metastatic/migrastatic agents via inhibition of oncogenic signalling (e.g., IL-6, IL-8, EGF and others) and matrix metalloprotease expression decrease migration and invasion of cancer cells and thereby CTC level formation CTC cluster. Curcumin application thank inhibition of NF-κB and HIF-1α activity decrease resistance of cancer cells against therapy. The Figure was partly generated using Servier Medical Art, provided by Servier, licensed under a Creative Commons Attribution 3.0 unported license.

suggests that curcumin application could also significantly repress CTC infiltration into the premetastatic site. Palange et al. reported, that curcumin significantly decreased cell adherence of MDA-MB-231 by almost half [60]. In addition, curcumin (NANOCurc<sup>TM</sup>) also repressed TNF- $\alpha$  activation of the HUVECs (in vitro model of inflamed endothelial cells). Combined exposition of both HUVECs and MDA-MBA-231 cells resulted in a 70% reduction of tumour cell adherence.

Curcumin can target infiltrated cancer and suppress micrometastases development. In the survival of dormant cancer cells, hypoxia plays a significant part, and their activation is controlled by the Wnt/ $\beta$ -catenin signalling pathway [61,62]. Curcumin can effectively target cancer cells hypoxia phenotype [63,64] and is a potent inhibitor of Wnt/ $\beta$ -catenin signalling [65–69]. Curcumin could also suppress oxidative stress (essential factor for metastasis formation) [70] in the affected tissue [71].

Some recent studies also imply, that curcumin could also suppress formation of premetastatic niches in the distant organs via tumour exosomes. Their application did not lead to a decrease in exosome formation but stimulated their production by cancer cells. Nevertheless, their biological effect is the opposite and displays an antitumour and metastatic properties, for example the activation of NK cells, suppression of angiogenesis and recruitment of cancer associated cells [72–74]. The effect of curcumin on the metastatic process is shown in more detail in Fig. 3.

The presented model clearly shows that curcumin is a potent anticancer agent with strong and robust antimetastatic effects in many oncological diseases. However, is it sufficiently supported by relevant studies? We can say that results obtained from the numerous experimental studies are consistent with this hypothesis. Numerous in vitro studies have shown that curcumin suppresses migration and invasiveness in various cancer cell models (e.g., breast, lung, colorectal, thyroid, pancreatic, liver and others) (Table 1).

Some studies imply (Table 1), that curcumin effects on cell metastatic activity are significantly higher than on cell proliferation. For example, 20  $\mu$ M curcumin has no significant effect on MCF-7 mammosphere proliferation, but 15  $\mu$ M curcumin reduces cell migration by up to a quarter [86]. In this case of MDA-MB-231, the IC<sub>50</sub> value was 28.7  $\mu$ M, but application of 20  $\mu$ M curcumin reduced cell invasion by approximately up to 10% of the original value [79]. Similarly, also in this case, the effect on cell proliferation was sometimes lower than the effect on cell migration and invasiveness in SCC25 head and neck cancer cell lines [112–114]. In this case of SCOV3 spheroids, a reduction of proliferation by up to half was observed at relatively high levels of curcumin (60  $\mu$ M), however a reduction in cell adhesion and invasion by up to a quarter and less than half was found for the half-dose of curcumin [96].

Some published results suggest, that curcumin could effectively, or preferentially target cancer stem cells (CSCs) like phenotype [93], drug resistance, [100,101] hypoxia [63,64,106] or activated cancer cells (e. g., TPA, phthalate, lipopolysaccharide, neurotensin and autocrine growth hormone signalling) [65,66,82,85,102,118]. In this case of primary human pancreatic cancer cells, curcumin had a significantly higher effect on the inhibition of cell invasiveness, which was observed in cells under hypoxia conditions compared to cells under normoxia [64]. These cell phenotypes display higher metastatic activity and lower sensitivity against typically used drugs. It is well known that tumours that display strong heterogeneity and mechanisms of metastatic activity are associated with displaying similar mechanisms of drug resistance [130,131]. It could be suggested that curcumin can selectively target cells with high metastatic activity and thereby significantly suppress tumour metastatic activity.

Significant antimetastatic effects were also observed in in vivo studies (Table 2). Curcumin applications lead to the suppression of metastatic tumours (e.g., decrease in the number of metastatic nodules, or metastatic mass and volume) and to the down expression of metastatic factors in the primary and metastatic tumour.

Results from various cancer models (e.g., melanoma, colorectal and breast cancer) clearly indicate that curcumin can also target metastasis formation, especially in lung tissue, in addition to suppressing primary tumours. Some studies suggest that curcumin applications could at least decrease development of bone metastasis (incurable in the present time).

Nevertheless, some recent studies have shown possible limitations of curcumin anticancer/antimetastatic effectivity. Prakobwong et al. reported, that during the late stages (5–6 months) of the study, curcumin lost inhibition against NF- $\kappa$ B (golden hamsters infected by *Opisthorchis viverrini* and treated with N-nitrosodimethylamine) [142]. It could imply, that curcumin and other curcuminoids are mostly effective in the early stages of cancer, or as preventive agents. Numerous in vivo studies reported that curcumin application suppresses carcinogen induced



Fig. 3. Curcumin anticancer effect with emphasis on antimetastatic activity. Metastasis formation is a complex process. Potentially metastatic cancer cells are activated by stimuli produced by the tumour microenvironment. Tumour associated cells can induce the mesenchymal phenotype (migrated and metastatic phenotype of cancer cells). In addition, tumour associated cells can migrate with circulating tumour cells and protect them from the immune system and drugs or protect them against Anoikis apoptosis caused by stress in the bloodstream. On the other hand, cancer with the mesenchymal phenotype displays more potent abilities in recruitment and activation of cancer associated cells and in suppression of the immune system, higher resistance to drug and greater similarity with the phenotype of cancer stem cells. Primary tumours can also support metastasis formation via preparation of metastatic niches through exosomes and metastatic factors. Tumour microbiota may have a significant role in this. In the present time, it has been shown, that primary and metastatic tumours can contain their own specific microbiota, which stimulate tumour development and progression. Curcumin can target tumour growth and metastases through independent mechanisms. It suppresses the phenotype of tumour associated cells and can activate the immune system to kill cancer cells. Curcumin also displays direct cytotoxic effects on cancer cells and decreases their proliferation, migration, and invasiveness. It decreases tumour growth and angiogenesis enhances sensitivity of the tumour to chemotherapy and other used anticancer regiments. Curcumin application can decrease the subpopulation of stem cancer like cells in the tumour. Some works suggest that curcumin could be hacking the tumours exosome system and inducing the production of exosomes with antitumour and antimetastatic properties. Curcumin also displays potent cytotoxicity against some typical representatives of the tumour microbiota. In the bloodstream, curcumin directly targets CTCs and activates immune cells to kill them. Curcumin decreases the adherence of cancer cells and thereby most probably decreases their infiltration into distant tissues. In the metastatic niche, curcumin reduces survival and activation of inactive tumour cells and suppresses metastasis. The Figure was partly generated using Servier Medical Art, provided by Servier, licensed under a Creative Commons Attribution 3.0 unported license.

cancer. In addition, several clinical trials reported, that curcumin decreased DNA damage after exposition to chemical pollutants [144-146]. Nevertheless, Yan et al. surprisingly reported, that a curcumin rich diet before and after tumour implantation (Lewis lung carcinoma in the mice model) and surgical removal did not display any effect on the primary tumour and increased angiogenesis and metastatic volume [137]. However, in the other studies with the mice model, curcumin displayed potent effectivity against Lewis lung carcinoma, including metastatic activity [147-150]. Fan et al. reported, that although curcumin induced the HIF1 $\alpha$ /mTOR/VEGF/VEGFR cascade in ischemic tissue (obtained from mice bearing Lewis lung carcinoma), its efficacy in cancer tissue was reversed [148]. A possible explanation could also be that curcumin resistant cells represent more aggressive cancer lines. Nevertheless, Yan et al. reported that curcumin-surviving subpopulation of Levis lung cancer cells display lower expression of ALDH1A1 and NF-KB and their metastatic activity (after mice implantation) is low [151].

In short, it could be said, that curcumin and most probably also other curcuminoids could suppress various stages of the metastatic process by various independent mechanisms (Fig. 1). On the other hand, it is appropriate to admit here that the presented model is mostly based on the results of in vitro and in vivo studies (Tables 1 and 2, respectively) and its verification or rebuttal is not possible clinically without

appropriate clinical studies.

## 3. Curcumin formulations: data from clinical trials

Some relevant clinical trials have shown indisputable limits of possible curcumin application [152]. For example, incorporation of curcumin powder into docetaxel therapy of patients with metastatic castration-resistant prostate cancer did not lead to any positive results and the clinical trial had to be terminated [153]. A possible explanation could be based on the low solubility ( $456 \mu g/L$ ) [154,155] and biostability of natural curcuminoids and thereby their limited bioactivity. Healthy volunteers' serum curcumin levels were 57.6 ng/mL two hours after oral application of 12 g curcumin (standardized powder) [156]. To improve curcuminoid solubility and biostability, other curcuminoid formulations of applications (nano/micro) are being intensively studied.

Unlike classical materials, nanoparticles can display significantly different properties such as higher solubility and bioactivity [157]. In this case, bulk particles < 1% of contained atoms co are localized on the surface. Whereas > 80% of total atoms are on the surface of nanoparticles because of their small size, large surface area and polydispersity.

In accordance with the above, suitable nanoformulation can strongly boost therapeutic effectivity of curcumin [158]. Paradoxically, first

 
 Table 1

 Examples of antimetastatic effects of curcumin with emphasis on migration and
 inva

nvasiveness of tumour cells in in vitro studies.		Model	Curcumin effect
Model	Curcumin effect Assay (value of tested effects in curcumin presence/original value		effects in curcumin presence/original value of tested effects <sup>a</sup> ; dose/ time (µM/h) <sup>b</sup>
	of tested effects <sup>a</sup> ; dose/ time (μM/h) <sup>b</sup>	MDA-MB-231[77]	Cell (
Breast carcinoma			AXL, <b>0</b> Twist,
EMT 6[75] MCF-7[76]	Scratch (1/3; 30/24) Cell (		↑ZEB1 and ZEB2,
	↓NF-кВ,		∱Fibronectin,
	↓uPA) <b>Proliferation</b> (63%; 20/ 48), <b>Adhesion</b> (3/4; 20/ 1; 20 min pretreatment) and <b>Invasion</b> (1/2; 20/ 12; 24 h pretreatment)	MDA-MB-231[79]	↓Vimentin, 0EHZ and 0STAT-3, 0Notch1) Migration (1/4; 30/48) and Invasion (1/7; 30/ 48) Cell (
CF-10 F[77] normal immortalized breast epithelial cell line	Cell (		Cli1 and 2
inc	↓AXL,		PTCH1 and
	↓Twist,		
	∱Notch1		<b>Proliferation</b> (IC <sub>50</sub> ; 28.7/24) Invasion (1/
	∱Fibronectin,		10; 20/24 h) and <b>Scratch</b> (1/5; 20/24 h)
	$\uparrow$ Vimentin) <b>Proliferation</b> $\ge$ 30/48; inhibition), <b>Migration</b>	TGF-β induced MDA-MB-231[80]	Cell ( ↓p38,
Tumour? from Alako E coll line, and intigation transformed)	(1/3; 30/48) and Invasion (1/4; 30/48)		↓Smad2/3,
[77]	↓AXL,		↓PTHrP) <b>Proliferation</b> (50%; 24/ 24)
	↓Twist,	MDA-MB-231[67]	Cell (
	ZEB2 and		↓CD44 <sup>+</sup> CD24 <sup>-</sup> subpopulation,
	STAT3		↓β-catenin,
	¢¢		Vimentin,
	Notch1 EZH2		Fibronectin,
	Fibronectin,		↓Oct4,
	Vimentin) Migration (4/5; 30/48) and Invasion (1/2: 30/		∱E-cadherin,
MDA-MB-231[78]	48) MMP3 secretion		↓Nanog,
	Cell migration,		↓Sox2 and
DMC-MDA-MB-231[78]	Proliferation (60–70%; 20/24) and Invasion (2/ 3; 15/8) MMP3 secretion	T47D-wt and T47D-GH <sup>+</sup> (Autocrine GH expression)[65]	mamospheres) <b>Proliferation</b> (50%; ~25/24) and <b>Scratch</b> (2/ 3; 15/24) <b>Cell</b> (
	Cell migration, <b>Proliferation</b> (70–80%;		pSTAT1,
RDMC-MDA-MB-231[78]	20/24) and <b>Invasion</b> (1/ 2; 7.5/8)		p38,
	Cell migration		↓c-jun,
	Proliferation (80–90%;		$\bigvee$ NF- $\kappa$ B activation,
	3; 7.5/8)		↓miRNA-182–96-183, (continued on next page)

## Table 1 (continued)

Model	$\begin{array}{l} \mbox{Curcumin effect}\\ \mbox{Assay} (value of tested \\ \mbox{effects in curcumin} \\ \mbox{presence/original value} \\ \mbox{of tested effects}^{3}; \mbox{dose/} \\ \mbox{time} (\mu M/h)^{b} \end{array}$	Model	Curcumin effect Assay (value of tested effects in curcumin presence/original value of tested effects <sup>a</sup> ; dose/ time (µM/h) <sup>b</sup>
	↑vimentin,		↓Vimentin,
	↓β-catenin, ↓MMP-2		GH) <b>Proliferation</b> (50%; ~20 and 25/24), <b>Migration</b> (2)(1) 12, 22 (20)
	↓N-cadherin and		(3/4 and 1/2; 20/ 24) and <b>Invasion</b> s (1/4 and 1/2; 20/ 24)
	↑E-cadherin)	MDA-MB-231-wt/MDA-MB-231-GH <sup>+</sup> [66]	Cell (
	<b>Proliferation</b> (50%; ~20 and 25/48), <b>Scratch</b> (1/3		↓STAT3,
	and <1/10; 20/24) Migration (1/2 and 1/2; 20/ 48) and Invasion (1/		MMP-2, Slug,
BHMC-MDA-MB-231[81]	3 and 2/3; 20/2) Cell (		↓β-catenin,
			GH)
	MMP-9) <b>Proliferation</b> (50%; ~50/ 24), <b>Scratch</b> (2/3; 12.5 / 24), <b>Invasion</b> (1/2;12.5		Proliferation (50%; ~ 20 and 25/ 24), Migration (1/3 and ¼; 20/24) and Invasion (1/2 and
	/ 24) and Migration	MDA MD 021[02]	<sup>1</sup> / <sub>2</sub> ; 20/ 24)
MCF-7 (normal and indudced12-O-	(1/2; 12.5 / 24) Cell (	MDA MB-231[83]	Cell
tetradecanoylphorbol-13-acetate (TPA))[82]			↑ROS.
	↓p-38,		↑E-Cadherin,
	JNK,		1
	↓IkBa,		↓STAT3, VEGF), Paclitaxel
	JKKab,		synergy Proliferation (41% and
	↓IKKb,		63%; 10 µg/mL (27uM) curcumin and curcumin
	<b>↓</b> p65,		with (5 and 5 µg/mL) paclitaxel/36),
	<b>↓</b> p50,		Migration (1/2 and 1/5; 10 µg/mL curcumin and
	↑P-c-Jun,		(5 and 5 $\mu$ g/mL) /36) and Scratch (4/5 and 1/2;
	$\mathbf{P}\mathbf{K}\mathbf{C}\alpha$ )		10 µg/mL curcumin and
	Proliferation (~90%; 30/24: normal) and		curcumin with paclitaxel
	Invasion (1/4;30/24; TPA induced)	4T1[83]	(5 and 5 μg/mL) /36) Cell (
WGF-/ AHU WGF-/-GFI [00]			↑ROS,
	JMMP-2,		∱E-Cadherin,
	↓Slug,		↓STAT3,
	av 11 -		VEGF) Paclitaxel
	↓N-cadherin,		synergy
	GH)		Proliferation (40% and
	▶ Proliferation (50%; ~20		50%; 10 μg/mL (2/uM) curcumin and curcumin
	and 25 / 24), Migration		with paclitaxel (5 and
	(1/4 and 1/2; 20 / 24)		5 μg/mL)), Migration
	and <b>Invasion</b> $(1/2 \text{ and} 1/4 \cdot 20/24)$		(3/5 and 1/3;10 10 μg/
MDA-MB-453-wt/MDA-MB-453-GH <sup>+</sup> [66]	Cell (		mL curcumin and
	STAT3, Akt,		(5 and 5 $\mu$ g/mL)/36)
	▼ 1 1		0 and Scratch (1/3 and
	MMP-2,		1/2;10 10 µg/mL
			(continued on next page)

Table 1 (continued)		Table 1 (continued)	
Model	<b>Curcumin effect</b> Assay (value of tested effects in curcumin presence/original value of tested effects <sup>a</sup> ; dose/ time (μM/h) <sup>b</sup>	Model	Curcumin effect Assay (value of tested effects in curcumin presence/original value of tested effects <sup>a</sup> ; dose/ time (µM/h) <sup>b</sup>
	curcumin and curcumin with paclitaxel (5 and 5 ug/mL))		↓Steam like phenotype (CD44 <sup>+</sup> CD24 <sup>-</sup> ),
LPA induced MCF-7[84]	Cell (		↓microtentacles and
	RhoA,		reattachment) Proliferation
	↓P MVDT1		(insignificant;0–50 μM/ 6 h)
	↓ <sup>MIPII</sup> ,	801D[88]	Cell (
	$\downarrow$ ROCK 1 and 2,		Cdc42, PAK1,
	↓MMP-2 and 9) <b>Proliferation</b>		↑E-cadherin,
	(60–70%; 75/24) and <b>Invasion</b> (2/3; 15/24)		↓actin cytoskeleton reorganization)
MCF-7(normal and LPS induced)[85]	Cell (		Invasion (4/5; 5/ 24)
	↓EMT,	A549 and H226[89]	↑E-cadherin,
	∱E-Cadherin,		↓vimentin,
	↓Vimentin,		↓TCF8,
	↓Snail,		↓Snail,
	<b>↓</b> p65)		↓Slug,
	<b>Proliferation</b> (70–80%; 20/24; normal) and		↓TLR4/MyD88 and
MDA-MB-231 (normal and LPS induced)[85]	Invasion (1/2 and 1/3; 20/48; 1 h pretreatment) Cell (		↓EGFR <b>Proliferation</b> (IC <sub>50</sub> ;
	<b>↓</b> EMT,		19.71 and 30.88 / 72), Colony formation (2/3 and 1/3; 5/ 2 and 3
	∱E-Cadherin,	4549[90]	weeks) and <b>Migration</b> (1/3 and 5/6; 5/48 h) <b>Cell</b> (
	↓Vimentin,	100550	DISK
	↓Snail,		
	↓p65) <b>Proliferation</b> (70–80%;		VECE
	20/24; normal) and <b>Invasion</b> (2/3 and 1/3;		VEGF,
Mammosphere of T47D[86]	20/48; 1 h pretreatment) <b>Proliferation</b>		↓ FAR,
	(75–100%;15/24) and <b>Migration</b> (1/3; 15/24)		↓ <sup>1(35</sup> ,
Mammosphere of MCF-7[86]	Cell (		MMD 2 and 0
	↑E-Cadherin) <b>Proliferation</b>		Dho A and
	(Insignificant; (0–20 $\mu$ M)/24),		
	Adhesion (1/4; 15/1) and Spreading (3/1; 15/ 3 h), Migration (1/4;		↓FAK) <b>Migration</b> (1/3 and 1/ 10; 10/24 and 48) and
	15/24) and <b>Three-</b> dimensional invasion	A549[91]	<b>Invasion</b> (>5% and 1/ 10; 10/24 and 48) <b>Cell</b> (Clobal change in
Lung carcinoma	(1/2, 13/40)	1012[21]	the mRNA profile)
H1299[75] Mammospheres BT-549[87]	Scratch (1/3; 30/ 24) Cell (		Proliferation (90%; 10/ 48), Scratch (6/10; 10/ 24) and Invasion (1/3;
			10/24) (continued on next page)

## Table 1 (continued)

Table 1 (continued)		Table 1 (continued)	
Model	Curcumin effect Assay (value of tested effects in curcumin presence/original value of tested effects <sup>a</sup> ; dose/ time (µM/h) <sup>b</sup>	Model	Curcumin effect Assay (value of tested effects in curcumin presence/original value of tested effects <sup>a</sup> ; dose/ time (µM/h) <sup>b</sup>
A549[92]	Cell ( ↓MMP2 and 9, ↓p21)		(50%;30–40/12), Adhesion (83%; 10/6 h), Migration (76%; 10/6 h) and Invasion (75; 10/ 6 h)
	Proliferation (50–75%; 20/24), Scratch (1/4; 10/24) and Invasion (1/ 4; 10/24)	Colorectal carcinoma HCT-116 and LoVo[98]	<b>Proliferation</b> (IC <sub>50</sub> ; 10 and 20/48), <b>Scratch</b> (88% and 72%; IC <sub>50</sub> (10
A549 (CD166 +/EPCAM+ and CD166-/EPCAM-)[93]	Cisplatin sensitivity <b>Proliferation</b> ( $IC_{50}$ ; 40/ 48) and <b>Scratch</b> (1/2 and 2/4: 40/48) (1/3 and $\frac{1}{2}$ :	HCT-116 cells[99]	and 20)/24) and <b>Invasion</b> (1/7 and 1/7 IC <sub>50</sub> (10 and 20)/24) <b>Cell</b> (
H2170(CD166 +/EPCAM+ and CD166-/EPCAM-)[93]	40 with 30 μM cisplatin) <sup>↑</sup> Cisplatin sensitivity <b>Proliferation</b> (IC <sub>50</sub> ; 30/		↓ММР-9, ↓NF-кВ,
	<ul> <li>48)</li> <li>Scratch (1/2 and 3/4;</li> <li>30/48), (1/10 and 1/7;</li> <li>30/48 with 7uM</li> </ul>		↓claudin-3, ↑FAS.
BDMC-TGF-β1 induced 95D[94]	cispiatin) Cell ( ^E-Cadherin, ^WIF-1		<ul> <li>↑E-cadherin)</li> <li>Proliferation (50%;</li> <li>10–20/24 h), Colon</li> <li>formation (50%; 10/48)</li> </ul>
	↓Vimentin,	HCT-116 (normal and 5-FU resistant)[100] in alginate beads culture	and Migration (55%; 10/24) Cell (
	↓Snail,		↓CXCR4, ↓MMP9,
<b>Ovarian</b> DU145[95]	$ ightarrow \beta$ -catenin nuclear translocation) Scratch (>10%; 5/48) and Invasion (1/2; 5/48) Cell (	SW480 and 5-FU-Resistant SW480[101]	$\bigvee$ NF- $\kappa$ B) Proliferation (IC <sub>50</sub> ; 9 and 5/14 days) and Invasion (1/3 and 1/5; 10/28 day) Cell (
	↓HES-1,		↓MYC,
SKOV3 spheroid[96]	↓MMP-2, ↓MT1-MMP) Proliferation (50%; 25/ 48 h) and Scratch (1/2; 25/24) Cell ( ↓ALDH1A1)	Neurotensin activated HCT-116[102]	↓insulin and IGF-1 receptors) Proliferation (IC <sub>50</sub> ; 20 and 17/ 72 h), Colony formation (<5% and <5%; 20/ 3 weeks) and Scratch (>95% and 1/3; 20/48) Cell (
	Proliferation (50%; 60/ 48 and; 35/48 (monolayer)), Adhesion		↓AP-1,
	(1/4; 30/48), <b>Invasion</b> , (4/10; 30/48) and <b>Mesothelial clearance</b> <sup>a</sup>		$\bigvee$ NF- $\kappa$ B, Ca <sup>2+</sup> mobilization and
SKOV3[97]	(½;30/48) Cell ( ↓STAT3,	Rko and HCT116[103]	↓IL-8 secretion) and Migration (1/3; 10/16; Neurotensin) Cell (
	↑polygonal Appearance and		↓p-c-Jun and
	↓filopodia formation) <b>Proliferation</b>		c-Fos,

(continued on next page)

Table 1 (continued)		Table 1 (continued)	
Model	<b>Curcumin effect</b> Assay (value of tested effects in curcumin presence/original value of tested effects <sup>a</sup> ; dose/ time (μM/h) <sup>b</sup>	Model	Curcumin effect Assay (value of tested effects in curcumin presence/original value of tested effects <sup>a</sup> ; dose/ time (μM/h) <sup>b</sup>
	<sup>↑</sup> Pdcd4) <b>Proliferation</b> (IC <sub>50</sub> ; 10 and 16/49) Migration	K1[107,108]	and <b>Invasion</b> (4/5; 25/ 24; 1 h pretreatment) <b>Cell</b> (
	$(1/3 \text{ and } 1/3; \text{ IC}_{50} (10 \text{ and } 16)/13; 24 \text{ h}$		<b>↓</b> MMP-9,
	preretirement) and <b>Invasion</b> (1/2 and ¼; IC <sub>50</sub> (10 and 16)/13; 24 h		↑E-cadherin,
Cervical	pretreatment)		phenotype (
Bisdemethoxycurcumin <sup>c</sup> -Hela[104]	Cell (		↑round and
	Vimentin,		(prominent))
	JRAS,		↓Cell spreading, <b>Proliferation</b> (90%;
	RHO A,		12.5/24), Attachment (47%;
	↓ Cadherin,		12.5/24; 24 h pretreatment), <b>Scratch</b> (1/2; 12.5/
	↑NF-Kβ,		6; in presence VEGF; 6 h pretreatment),
	Snail) <b>Proliferation</b> (50%; 10-15/24), <b>Scratch</b> (3/4 and $\frac{4}{5}$ (24 and 48)		Migration (87%; 12.5/3 h; 24 h pretreatment) and Invasion (84%; 12.5/4 h)
	<b>Migration</b> (1/5 and 4/5; 5/24 and 48) and	Pancreatic carcinoma Primary human cancer cells (normoxia and hypoxia)	Cell (
Thrusid	<b>Invasion</b> (4/5 and ½; 5/ 24 and 48)	[04]	¢-SMA,
BCPAP[105]	Cell (		↓IL-6 secretion,
	∱E-Cadherin,		<b>↓</b> MMP-9,
	↓Vimentin,		∱E-cadherin,
	$\int$ TGF- $\beta$ 1 induced (		vimentin
	Smad2 and		↓IL-6/ERK/NF-κB,
	↓Smad3		↓EMT,
	MMP2 and 9),		↓hypoxia phenotype) <b>Proliferation</b> (50%; 20/
	Cell spreading) <b>Proliferation</b> (~50%; 50/24), <b>Attachment</b> (53%; 12.5/2) and	BxPC-3(normoxia and hypoxia)[64]	48 (normoxia) and Scratch (1/2 and ½; 20/ 24 h) Cell (
K1- hypoxia phenotype[106]	Scratch (66%; 12.5/24; 6 h pretreatment) Cell (		ERK,
KI Nypoku pichotype[100]	HIF-1α		↓Vimentin,
	↓ ↓MMP-9,		<b>↓</b> MMP9,
	↓ ↑E-Cadherin,		↓IL-6 signalling) <b>Invasion</b> (½ and 1/3; 20/ 48; pancreatic stellate
	↓BNIP3) Proliferation	Panc-1(normoxia and hypoxia)[64]	cells conditioned media) CM induced cell (
	24, <b>Scratch</b> (80%; 25/ 24; 1 h pretreatment)		↓erk,
			(continued on next page)

## Table 1 (continued)

Table 1 (continued)		Table 1 (continued)	
Model	Curcumin effect Assay (value of tested effects in curcumin presence/original value of tested effects <sup>a</sup> ; dose/ time (μM/h) <sup>b</sup>	Model	Curcumin effect Assay (value of tested effects in curcumin presence/original value of tested effects <sup>a</sup> ; dose/ time $(\mu M/h)^{b}$
BxPC-3 and Panc-1[109]	VF-kB) Invasion (½ and 3/4; 20/ 48; pancreatic stellate cells conditioned media) Cell ( ↑ROS, H <sub>2</sub> O <sub>2</sub> induced factors (	Oral squamous cell carcinoma SCC25[112]	(4/5; 10/48) and Invasion (3/4; 10/48) ↓Cellular cohesion in spheroids Proliferation (10–20%; 5/24) and Migration (3/ 4; 2/24) Cell(
	↓MMP-2 and 9, ↓ERK, ↓NF-ĸB)) <b>Proliferation</b> (50%;		↓Snail, ↓Twist, ↑p53,
Panc-1(normoxia and hypoxia)[63]	20/24), Scratch (4/ 5 and $\frac{3}{4}$ ; 5/24; in the presence of H <sub>2</sub> O <sub>2</sub> ) and Invasion (2/3 and $\frac{3}{4}$ ; 5 / 48; in the presence of H <sub>2</sub> O <sub>2</sub> ) Hypoxia cell (		↑E-Cadherin ↓MMP-9 and ↓MMP-2) <b>Proliferation</b> (50%; 10–15/24) and <b>Invasion</b>
	↓vimentin, ↓N-cadherin, ↑E-cadherin	SCC-25[114]	(5%; 10/24) Cell ( ↓uPA, ↓uPAR,
	↓SHH, ↓SMO and		↓MMP-9 and 2, ↓EGFR, ↓p-Akt,
TGF-β1 activated PANC-1[110]	GLI1) <b>Proliferation</b> (~75% and 69%; 20/72) <b>Scratch</b> (9/10 and 2/3; 20/24) and <b>Invasion</b> (7/8 and 2/3; 20/48) <b>Cell</b> ( <b>Shh</b> ,	5–8 F[115]	↓ERK1/2, ↓STAT3) Proliferation (70–80%; 40/24) and Invasion (1/ 2; 40/24; 24 h pretreatment) Proliferation (90%; 50/
	↓GLI1, ↑E-cadherin, ↓vimentin)	Liver carcinoma HepG2[116]	3 n) and Scratch (insignificant; 50/24) Cell ( ↓HSP70 and
Nasopharyngeal carcinoma	48), Scratch (1/3;30/24; 48), Scratch (1/3;30/24; 48 h pretreatment) and Invasion (1/2;30/48; 48 h pretreatment) Cell (	HepG2TT (thermal tolerance HepG2)[116]	Proliferation (50%; ~50/48), Scratch (1/7 and <5%; 50 and 80/24), Migration (22% and 2% and 50 and 80/24) Cell (
	↑p53, ↑p21, ↓MMP-2 and 9,	Hca-F[117]	<pre></pre>
	FAK) <b>Proliferation</b> (IC <sub>50</sub> ; 31/ 48), <b>Colony formation</b> (3/4; 10/24), <b>Scratch</b>		↓Cav-1, ↓CD147,

(continued on next page)

## Table 1 (continued)

Table 1 (continued)		Table 1 (continued)	
Model	Curcumin effect Assay (value of tested effects in curcumin presence/original value of tested effects <sup>a</sup> ; dose/ time (µM/h) <sup>b</sup>	Model	Curcumin effect Assay (value of tested effects in curcumin presence/original value of tested effects <sup>a</sup> ; dose/ time (µM/h) <sup>b</sup>
	1		∱Vimentin,
	EGFR,		N-Cadherin,
	↓EGF,		TC adh arim)
	p38MAPK,		Proliferation (50% ~20;
Huh7 (normal and phthalate induced)[118]	p-44/42MAPK) Migration (1/2;10/12) and Invasion (1/6; 10/ 12)16 h Cell (		and 30/72), Colony formation (½ and 2/3; 10/ 2 weeks), Scratch (2/3 and 2/3; 5 / 24), Migration (1/3 and 1/3; 10 / 2) and Invasion (½ and ½; 10 / 12)
	AhR,	5637[121]	Cell (
	ERK,		MMP-2 and 9,
	↓SKI,		ROS induced HO-
	↓N-cadherin (only for induced cells),		1) <b>Proliferation</b> (86%; 10/24), <b>Invasion</b> (93% and 62%; 5 and 10/24)
	fe-cadherin) <b>Proliferation</b> (Insignificant (0–5 $\mu$ M)/24),	Prostrate Carcinoma	<b>Invasion</b> (1/2 and 1/5; 5 and 10/24; HO-1 knockdown)
	Migration (9/10 and ½;5/24) and Invasion	DU145[75] DU-145[122]	Scratch (1/3; 30/24) Cell (
PLC/PRF/5 (normal and phthalate induced)[118]	(Insignificant and ½; 5/48) <b>Proliferation</b> (Insignificant (0–5 μM)/ 24), <b>Migration</b> (Insignificant and 2/3:5/	PC-3[123]	MMP 2 and 9) <b>Proliferation</b> (~50%; 81/24) and <b>Invasion</b> (1/ 2;13.6/24) <b>Cell</b> (
	(Insignificant and ½; 5/ (Insignificant and ½; 5/		↓CCL2 and
Hep3B[119]	Cell (		↓MMP-9)
	↓mTOR,		Adhesion (2/3;30/0.5; CCL2), Migration (2/3 and ½; 30/18; normal and CCL2) and Invasion
	Alet		(4/5 and 2/3; 30/ 18; normal and CCL2)
	TAKI,	Osteosarcoma	Cell (
	<pre>pGSK3β) Proliferation (50%; 20/ 24) and Scratch (significant: 20/48)</pre>		↓β-catenin nuclear translocation)
CB0140C12[120]	Cell (		<b>Proliferation</b> (60–80%; 20/48) and <b>Invasion</b> (1/ 5 and 1/2; 10/48; normal
	•		and β-catenin plasmid; 24 h pretreatment)
	stress fiber formation) Invasion (2/3;1/6 h), Adhesion (3/4; 1/1) and	Brain tumour N18[124]	Cell (
	Haptotactic migration (1/2: 1/3: to fibronectin		$\bigvee$ MMP-2 and 7,
	and laminin, without change of integrin		↓FAK,
Bladder carcinoma	expression)		↓Rho A,
T24 and 5637 cancer[69]	Cell (		↓ROCK-1,
	<b>∫</b> β-catenin,		<b>↓</b> COX-2,

(continued on next page)

### Table 1 (continued)

Model

A172	[125]

Melanoma

Osteopontin induced B16F10[126]

B16F-10[127]

B16F10[128]

#### B16F10[129]

of tested effects<sup>a</sup>; dose/ time (µM/h)<sup>b</sup> ↓iNOS, ↓NF-kB p65, ↓ERK1/2, ↓MKK7 and

Curcumin effect

Assay (value of tested

presence/original value

effects in curcumin

↓MMP-2 and -7) Migration (2/3 and 1/3; 7.5/24 and 48) and Invasion (1/2 and 1/3; 7.5/24 and 48 h) Cell (

> ↓PI3K, 0mTOR, 0AKT Autophagy factors (

↑C3B/LC3A

LAMP-1 and

Iocalisation) **Migration** (1/2; 25/16; OPN) **and Invasion** (1/2; 25/16; OPN) **Invasion** (45% and 2.2%;14 and 27 /24), (N. D. and 30.9%; 14 and 27/ 24; 24 pretreatment) **Cell** (

### ↑ROS) Proliferation (IC<sub>50</sub>; 38/ 24), Invasion (1/5; 12.5/24) and Migration (1/2; 12.5/24) Cell (

MMP 2 and 9) **Proliferation** (IC<sub>50</sub>; 41/ 24) and **Scratch** (1/6; 41/24)

 $\alpha$ -SMA, Alpha smooth muscle actin; AhR, Aryl hydrocarbon receptor; ALDH1A1, Retinal Dehydrogenase 1; AP-1, Activator protein 1; Akt, Protein kinase B; Atg protein, autophagy-related protein; AXL, Tyrosine-protein kinase receptor UFO; BNIP3, BCL2 Interacting Protein 3; C3B, Complement component 3b; CCL2, chemokine (C-C motif) ligand 2; CXCR4, C-X-C chemokine receptor type 4; COX-2, Cyclooxygenase-2; iNOS, Inducible nitric oxide synthase; c-Fos, AP-1 transcription factor subunit; EGFR, Epidermal growth factor receptor; EGF, Epidermal growth factor; ERK 1/2, Extracellular signal-regulated kinase 1/2; EZH2, Enhancer of zeste homolog 2; EMT, Epithelial-mesenchymal transition; FAK, Focal adhesion kinase; GH, Growth hormone; GLI1, GLI Family Zinc Finger 1; HES-1, Hairy and enhancer of split-1; HIF-1a, Hypoxia-inducible factor 1alpha; HO-1, Heme oxygenase 1; HSP70, Heat shock proteins 70; JNK, c-Jun N-terminal kinase; IGF-1, Insulin-like growth factor 1; IL-6, Interleukin 6; IL-8, Interleukin 8; LAMP-1, Lysosomal-associated membrane protein 1; LC3A, Light Chain 3A; IkBa, nuclear factor of kappa light polypeptide gene enhancer in B-cells inhibitor, alpha; IKKaβ, IkappaB kinase alfa and beta subunits; IKKβ, inhibitor of nuclear factor kappa-B kinase subunit beta; PI3K, phosphatidylinositol 3-kinase; PKCa, Protein kinase C alpha; MAPK, Mitogen-activated protein kinase; MyD88, Myeloid differentiation primary response 88; MKK7, Mitogen-activated protein kinase 7; MMP-2, Matrix metalloproteinase-2; MMP3, Matrix metalloproteinase-3; MMP-9, Matrix metalloproteinase-9; MT1-MMP, Membrane type 1 matrix metalloproteinase; mTOR, Mammalian target of rapamycin; MYPT1, Myosin phosphatase target subunit 1; NF-KB, Nuclear factor kappa-light-chain-enhancer of activated B cells; NOTCH1, Notch Receptor 1; PAK1, P21 (RAC1) Activated Kinase 1; pGSK3β, Glycogen Synthase Kinase 3 Beta; PTCH1, Protein patched homolog 1; PTHrP, Parathyroid hormone-related protein; ROS, Reactive oxygen species; ROCK 1, Rho-associated, coiled-coilcontaining protein kinase 1; uPA, Urokinase-type plasminogen activator; SHH, Sonic Hedgehog; SMO, Smoothened, Seven Transmembrane Spanning Receptor; STAT1, Signal transducer and activator of transcription 1; STAT3, Signal transducer and activator of transcription 3; TCF8, Zinc finger E-box-binding homeobox 1; TGF-β1, Transforming growth factor β; TLR4, Toll-like receptor 4; VEGF, vascular endothelial growth factor; WIF-1, Wnt inhibitory factor 1; ZEB1, Zinc Finger E-Box Binding Homeobox 1; ZEB2, zinc finger E-box binding homeobox 2;  $\uparrow$  = curcumin activation/induction;  $\downarrow$  = curcumin suppression/inhibition; **0** = without change.

<sup>a</sup> If a fraction was stated, the value was not explicitly stated in the article, but was subtracted from the presented graph.

<sup>b</sup> Unless other units are specified.

<sup>c</sup> Nature derivative of curcumin contained in turmeric extract.

nanoformulations probably were traditional applications of curcuminoids in Ayurvedic medicine in the form of golden milk (cow's milk, black pepper and honey). Milk constituents (e.g., lipids) strongly enhance curcumin solubility and stability [159]. Although, in the original recipe there was cow's milk, it can be easily replaced by other types, including plant ones. For example, Zheng et al. simply prepared a curcumin formulation with soy milk ( $\sim$ 0.2 mg /mL) which had good stability (4 and 20 °C, 37 days) [160]. Microscopic analysis of the microstructures of soymilk curcumin loaded into small lipid nanoparticles ( $\sim$  400 nm) was performed.

Piperine, highly abundant in black pepper, strongly increases curcumin bioactivity [161]. Piperine inhibits and suppresses enzymes in metabolic transformation of curcumin [162,163]. Another study (Patil et al., 2016) suggests, that piperidine could directly interact with curcumin and suppress its aggregation [164]. A combination of curcumin and piperine (2:1) sometimes display an increase in the cytotoxicity against Caco-2 cells [165]. Shoba et al. reported, that in the healthy volunteer, co-application of piperine (20 mg) in combination with curcumin (2 g) resulted in an increase of curcumin biovailability by more than three orders of magnitude [166]. Nevertheless, results of another study were not convincing enough. [167] On the other hand, some in vivo studies suggest that piperine can have a strong potent protective effect on curcumin against carcinogens [168-171]. It cannot also be neglected that piperine suppresses the expression of IL-1 $\beta$  in gastric cancer cells induced by H. pylori and inflammation in pyloric mucosa of Mongolian gerbils after infection [172]. In short, we can say piperine most probably boosts therapeutic potential of curcumin by various mechanisms; however, further clinical studies are needed for a definitive assessment.

In the case of natural honey, numerous antitumoural effects were also observed [173]. In addition, however, its benefit could lie in

## Table 2

Curcumin effects on tumours in animal models with emphasis on antimetastatic activity.

				comunitie
Model	Experimental condition	Results		
Implanted Melanoma			Mice with bone implanted MDA-MB-231;[136]	IV; borte (1 mg) a
C57BL/6 mice with B16F10 cells[127,132]	Orally; 200 nmol/kg in gum acacia; 10	↓Serum sialic acid level		(1.337 n /7 day, 3
	alternate days.	(23%),	Liver carcinoma	0 11 1
		lung	Metastasis the liver	200 mg/
		collagenhydroxyproline content (30%),	B6C3F1 mice[120]	20days
		↓lung tumour nodule formation (10%) and	Phthalates induced Huh7 [118]	Intraperi injection (50 mg)
		survival (+144%)		ethylhex phthalat
Mice with B16F10[129]	Orally; 3 and 6 mg/	↓lung metastasis (slight		/kg/2 da
	kg/day for 14 days	reduction)	Lung carcinoma	Orally: S
Bladder carcinoma Nude mice with MB49	Intravesical: (25 mg/	10 1 protein expression in	[137] s tumour removed	diet, 5 w
[121]	kg and 50 mg/kg) in	the lung tissue, lung nodule	after 10 days	LLC impl
	(23% DMSO and 30% propylene glycol in PBS)	numbers (insignificant)		20 days
Breast carcinoma Nude mice with MDA-MB- 435LVB[133]	2% curcumin in the diet; 5 days after	Lung metastasis (		
	tumour removal, and/or paclitaxel (i.	<b>p</b> 65,		
	p., 10 mg/kg, cremophor vehicle);	MMP 9 and		
	10, 17 and 24 days after tumour removal	paclitaxel induced COX-2)		
Nude mice with MDA-MB-	Intravenously (into	Lytic bone lesion area (57%		
231[80]	fat); 25 and 50 mg/ kg/ 2days for 21 days	and 51%) bone-resorbing osteoclasts at the bone-tumour interface (48% and 53%) tumour mass	Prostrate carcinoma SCID mice with DU-145 [122]	Orally; 5 0.5% methylce
		(insignificant)		three tin
Mice with 4T1[83]	IP; curcumin (50 mg)	Tumour (		for 10 w
	or curcumin and paclitaxel (25 and	STAT3,	Gastric carcinoma BALB/c nude mice with	intraperi
	25 mg) /daiy for 21 days	↓MMP2, 3 and 9,	346-7901[136]	160 mg/ weeks
		↓cyclin D,		
		↓VEGF),		
		tumour volume and		
		↓neoplasticity (zero in the combination with paclitaxel) in mammary and hepatic		
Mice with 4T1[134]	Oral: (50 mg/kg)	tissues		
	/2 day for 16 days	↓ <sup>1</sup> unioui voiunie,	Colon carcinoma	
		pulmonary metastatic	Chicken embryos with Rko [103]	Intraven and 14th
BALB/c mice with 4T1	IV; 30 mg/kg /day	↑OS (37 vs 28),		
[199]	IOL IO Udys	↓tumour volume and mass (1/2)	Chicken embryos with HCT116[103]	Intraven and 14th
		$\uparrow$ apoptosis index (2x) and	Head neck carcinoma Mice with patient OSCC	injected
		anogiogenesis	biopsy fragment[112]	tumour; /weak fo
		(microvehicles per field 1/		/ weak IC

2),

Leukemie

## Table 2 (continued)

Model

Experimental condition	Results
IV; bortezomib (1 mg) and curcumin (1.337 mg) /kg, /7 day, 35 days	umber and mass of lung metastases (1/2) <b>Bone</b> : (Slightly reduction in macraphage infiltration tumour volume)
Orally; 100 and 200 mg/kg/day for 20days Intraperitoneal injection; curcumin (50 mg) and bis(2- ethylhexyl) phthalate (60 mg) /kg/2 day for month	Number of intrahepatic metastases (1/3% and 5%) Volume of primary tumour (1/4 and 1/6) Tumour volume, Jung nodules
Orally; 2% and 4%	Plasma (
diet, 5 weeks before LLC implantation and 20 days after	↑angiogenin,
	↑bFGF,
	∱VEGF,
	↑PDGF-BB,
	↑Щ-1β,
	↑MCP-1)),
	↑metastasis volume
Orally; 5 mg/kg in 0.5% methylcellulose and 0.1% Tween 80, three times per week	Pulmonary metastasis (from 90% to 60%), No. of metastatic nodules (from 11.0 to 1.2), <b>tumour tissue</b> (
for 10 weeks	MMP2 and 9)
intraperitoneal injections; 40, 80, or	<b>Lymphatic vessel (0</b> VEGFR- 3, <b>0</b> Podoplanin, <b>0</b> Prox-1),
weeks	Jymphatic vessel density,
	↓tumour volume and gastric tumour (
	↓VEGFR-3,
	↓Podoplanin,
	Prox-1,
	LYVE-1)
Intravenously; 12th and 14th day	Liver and lung metastases (11% and 24%), <b>tumour</b> (
Intravenously; 12th and 14th day	pre and miR-21) (Liver and lung metastases 55% and 53%), tumour ( pre and miR-21)
injected around the tumour; 70 mg/kg /weak for 4 weeks	↑Tumour differentiation, ↓inflammatory infiltrate

(continued on next page)

### Table 2 (continued)

Model	Experimental condition	Results
BALB/cJGpt- Prkdcem <sup>1Cd561</sup> /Gp mice with SHL-1[130]	Intraperitoneal injections; 15 and 30 mg /kg/day in	Tumour (
with SHI-1[155]	olive oil for 15 days	p38, ↑.INK.
		ERK and
		NF-кB,
		MMP 9 and 2),
Induced		tumour volume
Long-Evans Cinnamon rats [140]	0.5% curcumin in the diet for life	↓Median survival time (from 88.7 to 78.1 days),
MTX treated mice with MNNG/HOS/MTX[141]	Intraperitoneal; Curcumin (5 mg/ kg) and MTX (0.4 mg/ kg)/weakly for 4	Metastases (18% vs 0%) P-gp expression (1/3), primary tumour volume (22%), cases of lung metastasis (from 3 to 0)
Viverrini infection and N-	weeks Orally; 65 mg/day	Nuclear extract form liver (
nitrosodimethylamine exposed golden hamsters[142]	for 6 months	↓STAT-3,
		↓c-fos,
		<b>₽</b> 65,
		TRAF-1 and
		JAK2) tumour (
		↓COX-2,
Tobacco smoke exposed	Orally; 50 or	↓iNOS), Tumour volume and development, 5–6 month- no effect of curcumin on NF-κB activation Lung (
DALB/C IIICe[143]	12 weeks	∱E-cadherin,
		↑ZO-1,
		↓vimentin,
		↓N-cadherin,
		Jun,
		FOS and
		↓ERK 1/2)

Akt, Protein kinase B; bFGF, Basic fibroblast growth factor; c-Fos, AP-1 transcription factor subunit; COX-2, Cyclooxygenase-2; Extracellular signal-regulated kinase 1/2; iNOS, Inducible nitric oxide synthase; HO-1, Heme oxygenase 1; IL-1 $\beta$ , Interleukin-1 beta; JAK2, Janus kinase 2, JNK, c-Jun N-terminal kinase; Jun, Fos-binding protein **p39**; GSK3 $\beta$ , Glycogen Synthase Kinase 3 Beta; LYVE 1, Lymphatic Vessel Endothelial Hyaluronan Receptor 1; MCP-1, Monocyte chemoattractant protein-1; MMP-2, Matrix metalloproteinase-2; MMP3, Matrix metalloproteinase-3; MMP-9, Matrix metalloproteinase-9; mTOR, Mammalian target of rapamycin; NF- $\kappa$ B, nuclear factor kappa-light-chain-enhancer of activated B cells; PDGF-BB, Platelet-derived growth factor – two B subunits; PROX1, Prospero homeobox protein 1; STAT3, Signal transducer and activator of transcription 3; TRAF1, TNF Receptor Associated Factor 1; VEGF, vascular endothelial growth factor; VEGFR-3, Vascular endothelial growth factor to receptor 3; ZO-1, Tight junction protein-1

 $\uparrow$  = curcumin activation/induction/enchant;  $\downarrow$  = curcumin repression/inhibition; **0** = without change.

increasing the bioactivity of curcumin. Mor et al. reported that sucrose in the presence of piperine and high fat milk significantly increased the release of curcumin from its alginate deposit [159]. Honey has shown to have massive amounts of saccharides. There is an abundance of glucose and fructose and a smaller amount of sucrose in honey [174]. In this context, it is interesting that some curcumin formulations with enhanced bioactivity and bioavailability, such as Theracurmin and Neocurcumin, contain disaccharide maltase.

In accordance with the above, the application of solid lipid nanoparticles (similar nanostructure as milk) resulted in a significant increase of bioactivity of curcuminoids. In the case of Meriva (phytosomal formulation of curcuminoids), this formulation displayed an increase by more than an order of magnitude increase in maximum and total curcuminoid concentrations (0.2 and 1.3  $\mu$ g/mL) in plasma of a healthy volunteer [175]. In oncological patients with solid tumours, Meriva applications lead to decreased levels of metastatic factors (interleukin 6 (IL-6), TGF- $\beta$ , and tumour necrosis factor alpha (TNF- $\alpha$ )) [41]. Similarly, it was reported, that curcumin nanoformulaton sometimes lead to an increase in antimetastatic activity. [83,176] Promising pharmacokinetic results were also observed in other oral curcumin formulations (Table 3), for example after application of BCM-95 value of c<sub>max</sub> and AUC<sub>0-inf</sub> was 0.5 and 3  $\mu$ g/mL (1.3 and 8.1  $\mu$ M), respectively [177].

However, the question is to what extent the potential therapeutic efficacy of curcumin is caused by an increase of its concentration in the blood and how many other factors can increase curcumin concentration in the tumour tissue. Curcumin itself probably does not significantly accumulate in the tumour [146]. Nevertheless, nanoparticles (20-500 nm) and macromolecules (40 kDa and higher) accumulate strongly in tumour tissue due to the EPR effect (e.g., "enhanced permeability and retention") [194]. They are too big for the diffusion through capillary walls and their half-life in plasma is higher compared to other small molecules (e.g., classical drugs and curcuminoids). On the other hand, a characteristic feature of tumour tissue is that it leaks through pores (100 nm to 1 µm, depending on the tumour type) or by fenestration and there is a limited functionality of the tumour lymphatic system. Nevertheless, larger nanoparticles (d>100 nm) can be transported by sinusoids in the spleen and fenestrae of the liver, which are approximately 150-200 nm in diameter [194].

The distribution of nanoparticles can also depend on protein interactions [195]. Numerous proteins can be bound to the surface of nanoparticles and their size can sometimes be larger [176]. Aggregates of proteins and nanoparticles are removed by macrophages from the circulation. [195] Minimizing protein binding (e.g., surface modification) is a necessary condition for the design of long circulating nanoparticles. Stealth polymers are mostly used for these purposes such as poly(ethylene glycol) (PEG), poly(2-oxazoline) (POx).

However, success may not be guaranteed. In the case of the glioblastoma patients treated with NovaSol (micellar curcuminoids formulation,  $3 \times 840$  mg/day), it was found that curcumin serum and intratumoural concentration was 253 ng/mL (0.73  $\mu$ M) and 56 pg/mg (0.15  $\mu$ M), respectively. [196] Nevertheless, more optimistic results were reported by Gálvez et al. in breast cancer patients [193]. Although plasma concentration of free curcumin was 5.26 nmol/l. Its concentration in normal and malignant mammary tissue was approximately 1 and 0.2 mmol/l. However, authors did not test a single formulation of curcumin, but combined extracts (570 mg turmeric extract and ~ 350 mg other polyphenolic compounds per day). A negligible part of this formulation were flavonoids. In this context, it is interesting that some polyphenol compounds such as flavonoids can strongly enhance curcumin bioavailability and bioactivity [48,197–199].

The above suggests that the concentration of curcumin in plasma

## Table 3

Pharmacokinetic parameters of curcumin and its formulations from clinical trials.

(Dose Name; Subject) Lit.	Physiological curcuminoid concentration Curcuminoid (c <sub>max</sub> /ng/mL and AUC /ng.h/mL)
Unformulated	
(2 g; 8)[166]	Free curcumin (c <sub>max</sub> = 6)
(10 and 12 g; 1 and	<b>Free curcumin</b> ( $c_{max} = 50$ and 51, respectively)
$(1425 \text{ mg; } 30 (33.6 \pm 6.79 \text{ years}))$ [167]	free curcumin ( $c_{max} = 18$ and AUC <sub>0-24 h</sub> = 56), curcumin sulphate ( $c_{max} = 59.6$ and AUC <sub>0-24 h</sub> = 618), curcumin glucuronide ( $c_{max} = 34$ and AUC <sub>0-24 h</sub> = 236), Free DMC ( $c_{max} = 3.2$ and AUC <sub>0-24 h</sub> = 7.4), free BDMC ( $c_{max} = 0.7$ and AUC <sub>0-24 h</sub> = 1.3), THC glucuronide ( $c_{max} = 163$ and AUC <sub>0-24 h</sub> = 1230), HHC glucuronide ( $c_{max} = 97.9$ and AUC <sub>0-24 h</sub> = 946), THC sulphate ( $c_{max} = 59.5$ and AUC <sub>0-24 h</sub> = 317), HHC sulphate ( $c_{max} = 128$ and AUC <sub>0-24 h</sub> = 1370), and total curcuminoids ( $c_{max} = 445$ and AUC <sub>0-24 h</sub> = 5080)
(1295 mg curcumin, 396 mg <b>DMC</b> , and 108 mg BDMC; 3 (35 ± 10 years)) [175]	Curcumin ( $c^{max}$ 9 ng/mL AUC = 122.5,), DMC ( $c_{max}$ 4.2 ng/mL and AUC = 55.8) and BDMC ( $c_{max}$ = 2.1 and AUC = 24.6)
$(1800 \text{ mg}; 15 (23.0 \pm 2.4 \text{ years}))[178]$ (1500 mg; 7)[179]	Curcumn ( $c_{max} = 2.3$ and AUC <sub>0-12 h</sub> = 10.8), DMC ( $c_{max} = 1.7$ and = AUC <sub>0-12 h</sub> 18.4), BDMC ( $c_{max} = 1.1$ and AUC <sub>0-12 h</sub> = 9.3), THC ( $c_{max} = 1.1$ and AUC <sub>0-12 h</sub> = 9.1), and total curcuminoids ( $c_{max} = 5.2$ and AUC <sub>0-12 h</sub> = 39.6) Free curcumin ( $c_{max} = 0.154.9$ and AUC <sub>0-24 h</sub> = 0.116), total curcumin ( $c_{max} = 21$ and AUC <sub>0-24 h</sub> = 224), DMC ( $c_{max} = 21$ and AUC <sub>0-24 h</sub> = 189), BDMC ( $c_{max} = 1.1$ and AUC <sub>0-24 h</sub> = 189), BDMC ( $c_{max} = 1.1$ and AUC <sub>0-24 h</sub> = 189), BDMC ( $c_{max} = 1.1$ and AUC <sub>0-24 h</sub> = 189), BDMC ( $c_{max} = 1.1$ and AUC <sub>0-24 h</sub> = 189), BDMC ( $c_{max} = 1.1$ and AUC <sub>0-24 h</sub> = 189), BDMC ( $c_{max} = 1.1$ and AUC <sub>0-24 h</sub> = 189), BDMC ( $c_{max} = 1.1$ and AUC <sub>0-24 h</sub> = 189), BDMC ( $c_{max} = 1.1$ and AUC <sub>0-24 h</sub> = 189), BDMC ( $c_{max} = 1.1$ and AUC <sub>0-24 h</sub> = 189), BDMC ( $c_{max} = 1.1$ and AUC <sub>0-24 h</sub> = 189), BDMC ( $c_{max} = 1.1$ and AUC <sub>0-24 h</sub> = 189), BDMC ( $c_{max} = 1.1$ and AUC <sub>0-24 h</sub> = 189), BDMC ( $c_{max} = 1.1$ and AUC <sub>0-24 h</sub> = 189), BDMC ( $c_{max} = 1.1$ and AUC <sub>0-24 h</sub> = 189), BDMC ( $c_{max} = 1.1$ and AUC <sub>0-24 h</sub> = 189), BDMC ( $c_{max} = 1.1$ and AUC <sub>0-24 h</sub> = 189), BDMC ( $c_{max} = 1.1$ and AUC <sub>0-24 h</sub> = 189), BDMC ( $c_{max} = 1.1$ and AUC <sub>0-24 h</sub> = 189), BDMC ( $c_{max} = 1.1$ and AUC <sub>0-24 h</sub> = 189), BDMC ( $c_{max} = 1.1$ and AUC <sub>0-24 h</sub> = 189), BDMC ( $c_{max} = 1.1$ and AUC <sub>0-24 h</sub> = 189), BDMC ( $c_{max} = 1.1$ and AUC <sub>0-24 h</sub> = 189), BDMC ( $c_{max} = 1.1$ and AUC <sub>0-24 h</sub> = 189), BDMC ( $c_{max} = 1.1$ and AUC <sub>0-24 h</sub> = 189), BDMC ( $c_{max} = 1.1$ and AUC <sub>0-24 h</sub> = 189), BDMC ( $c_{max} = 1.1$ and AUC <sub>0-24 h</sub> = 189), BDMC ( $c_{max} = 1.1$ and AUC <sub>0-24 h</sub> = 189), BDMC ( $c_{max} = 1.1$ and AUC <sub>0-24 h</sub> = 189), BDMC ( $c_{max} = 1.1$ and AUC <sub>0-24 h</sub> = 189), BDMC ( $c_{max} = 1.1$ and AUC <sub>0-24 h</sub> = 180), AUC <sub>0-24 h</sub> =
(500 mg; 12)[180]	$(c_{max} = 8.8 \text{ and } AUC_{0-24 h} = 51.6)$ , <b>IFC</b> $(c_{max} = 87 \text{ and } AUC_{0-24 h} = 962)$ , and <b>total curcuminoids</b> $(c_{max} = 48.4 \text{ and } AUC_{0-24 h} = 470)$ <b>Curcumin</b> $(c_{max} = 43.1 \text{ and } AUC_{0-6 h} = 165)$
(90 mg; 7)[181]	Curcumin ( $C_{max} = 1.8$ and AUC <sub>0-12 h</sub> = 5.75)
Curcumin powder (1800 mg; 24) [182]:	Curcumin ( $c_{max} = 9.85$ and AUC <sub>0-6 h</sub> = 15.4), DMC ( $c_{max} = 3.20$ and AUC <sub>0-6 h</sub> = 6.42), BDMC ( $c_{max} = 2.15$ and AUC <sub>0-6 h</sub> = 4.89), THC ( $c_{max} = 5.38$ and AUC <sub>6-h</sub> = 7.88), and total curcuminoids ( $c_{max} = 15.65$ and AUC <sub>0-6 h</sub> = 33.21)
(1.5 g, 3 g and 6 g; 11)[183]	<b>Curcumin</b> ( $C_{max} = 41.63, 41.20$ and 2.84, respectively); Negative correlation with antioxidative activity
(500 mg; 23)[184] (2000 mg; 4)[177]	$ \begin{array}{l} \textbf{Curcumin} \ (c_{max} = 7.1 \ \text{and} \ AUC_{0-24 \ h} = 65.6 \ \text{nM}), \ \textbf{DMC} \ (c_{max} = 1.3 \ \text{and} \ AUC_{0-24 \ h} = 10 \ \text{nM}), \ \text{and} \ \textbf{BDMC} \ (c_{max} = 0.5 \ \text{and} \ AUC_{0-24 \ h} = 2.4 \ \text{nM}) \\ \textbf{Curcumin} \ (c_{max} = 150 \ \text{and} \ AUC_{0-inf} = 461) \end{array} $
supplement	
(2 g curcumin and piperine 20 mg; 8) [166]	Free curcumin (c <sub>max</sub> 180)
Curcumin C3	$\label{eq:Free curcumin} \textit{Free curcumin} (c_{max} = 12.9 \textit{ and } \textit{AUC}_{0-24 \textit{ h}} = 54.1), \textit{curcumin sulphate} (c_{max} = 42.3 \textit{ and } \textit{AUC}_{0-24 \textit{ h}} = 474), \textit{curcumin glucuronide} (c_{max} = 26 \textit{ and } \textit{AUC}_{0-24 \textit{ h}} = 474), \textit{curcumin glucuronide} (c_{max} = 26 \textit{ and } \textit{AUC}_{0-24 \textit{ h}} = 474), \textit{curcumin glucuronide} (c_{max} = 26 \textit{ and } \textit{AUC}_{0-24 \textit{ h}} = 474), \textit{curcumin glucuronide} (c_{max} = 26 \textit{ and } \textit{AUC}_{0-24 \textit{ h}} = 474), \textit{curcumin glucuronide} (c_{max} = 26 \textit{ and } \textit{AUC}_{0-24 \textit{ h}} = 474), \textit{curcumin glucuronide} (c_{max} = 26 \textit{ and } \textit{AUC}_{0-24 \textit{ h}} = 474), \textit{curcumin glucuronide} (c_{max} = 26 \textit{ and } \textit{AUC}_{0-24 \textit{ h}} = 474), \textit{curcumin glucuronide} (c_{max} = 26 \textit{ and } \textit{AUC}_{0-24 \textit{ h}} = 474), \textit{curcumin glucuronide} (c_{max} = 26 \textit{ and } \textit{AUC}_{0-24 \textit{ h}} = 474), \textit{curcumin glucuronide} (c_{max} = 26 \textit{ and } \textit{AUC}_{0-24 \textit{ h}} = 474), \textit{curcumin glucuronide} (c_{max} = 26 \textit{ and } \textit{AUC}_{0-24 \textit{ h}} = 474), \textit{curcumin glucuronide} (c_{max} = 26 \textit{ and } \textit{AUC}_{0-24 \textit{ h}} = 474), \textit{curcumin glucuronide} (c_{max} = 26 \textit{ and } \textit{AUC}_{0-24 \textit{ h}} = 474), \textit{curcumin glucuronide} (c_{max} = 26 \textit{ and } \textit{AUC}_{0-24 \textit{ h}} = 474), \textit{curcumin glucuronide} (c_{max} = 26 \textit{ and } \textit{AUC}_{0-24 \textit{ h}} = 474), \textit{curcumin glucuronide} (c_{max} = 26 \textit{ and } \textit{AUC}_{0-24 \textit{ h}} = 474), \textit{curcumin glucuronide} (c_{max} = 26 \textit{ and } \textit{AUC}_{0-24 \textit{ h}} = 474), \textit{curcumin glucuronide} (c_{max} = 26 \textit{ and } \textit{AUC}_{0-24 \textit{ h}} = 474), \textit{curcumin glucuronide} (c_{max} = 26 \textit{ and } \textit{AUC}_{0-24 \textit{ h}} = 474), \textitcurcumin glucuronide} (c_{max} = 26 \textit{ and } \textit{AUC}_{0-24 \textit{ h}} = 474), \textitcurcumin glucuronide} (c_{max} = 26 \textit{ and } max), \textitcurcumin glucuronide} (c_{max} = 26 \textit{ and } max), \textitcurcumin glucuronide} (c_{max} = 26 \textit{ and } max)), \textitcurcumin glucuronide} (c_{max} = 26 \textit{ and } max)), \textitcurcumin glucuronide} (c_{max} = 26 \textit{ and } max)), \textitcurcumin glucuronide} (c_{max} = 26 \textit{ and } max)), \textitcurcumin glucuronide} (c_{max} = 26 \textit{ and } max)), \textitcurcu$
Comples® (1425 mg; with 14 mg piperine Sabinsa; 30) TEP	= 187), Free DMC ( $c_{max} = 10.3$ and $AUC_{0-24 h} = 10.7$ ), free BDMC ( $c_{max} = 0.2$ and $AUC_{0-24 h} = 0.6$ ), THC glucuronide ( $c_{max} = 154$ and $AUC_{0-24 h} = 1230$ ), HHC glucuronide ( $c_{max} = 74$ and $AUC_{0-24 h} = 264$ ), THC sulphate ( $c_{max} = 59$ $AUC_{0-24 h} = 317$ ), HHC sulphate ( $c_{max} = 99$ and $AUC_{0-24 h}$ AUC24 = 1230), and total curcuminoids ( $c_{max} = 373$ and $AUC_{0-24 h} = 4380$ )
(2000 mg; lecithin, piperine (Life Extension, USA); 4)[177]	<b>Curcumin</b> ( $c_{max} = 344$ and $AUC_{0-inf} = 624$ )
Silicon dioxide/triaceti	n/Panodan® Micronized powder
(500 mg; 23)[184]	Curcumin ( $c_{max} = 51$ and AUC <sub>0-24 h</sub> = 700 nM), DMC ( $c_{max} = 34$ AUC = 246 nM) and BDMC ( $c_{max} = 5.1$ and AUC <sub>0-24 h</sub> = 27 nM)
Liposomal and micell (165 and 297 mg curcumin, 38 and 68 mg DMC and 6 and 11 mg	ar system: Meriva (lecithin curcuminoid formulation) Curcumin ( $c_{max} = 24.2$ and 50.3 and AUC = 272 and 538), DMC ( $c_{max} = 39.1$ and 134.6 and AUC = 297 and 655), and BDMC ( $c_{max} = 8.8$ and 24.9 and AUC = 70 and 142))
<b>BDMC;</b> 3 and 3 ( $35 \pm 10$ years)	
$(376 \text{ mg } 15 (23.0 \pm 2.4 \text{ years}))[178]$	<b>Curcumin</b> ( $c_{max} = 2.8$ and AUC <sub>0-12 h</sub> = 28.7), <b>DMC</b> ( $c_{max} = 5$ and AUC <sub>0-12 h</sub> = 28.7), <b>BDMC</b> ( $c_{max} = 0.8$ and AUC <sub>0-12 h</sub> = 6.7), <b>THC</b> ( $c_{max} = 0.1$ and AUC <sub>0-12 h</sub> = 1.1), and <b>total curcuminoids</b> ( $c_{max} = 2.3$ and AUC <sub>0-12 h</sub> = 65.3)
(152 mg; 9)[185]	Curcumin ( $C_{max} = 58.8$ )
(180–220 mg; 30) [167]	Free Curcumin ( $c_{max} = 11.3$ and AUC <sub>0-24 h</sub> = 37.6), curcumin sulphate ( $c_{max} = 27.1$ and AUC <sub>0-24 h</sub> = 291), and curcumin glucuronide ( $c_{max} = 25.6$ and AUC <sub>0-24 h</sub> = 142), free DMC ( $c_{max} = 6.4$ and AUC <sub>0-24 h</sub> = 2.2), BDMC ( $c_{max} = 0.2$ and AUC <sub>0-24 h</sub> = 0.2), THC glucuronide ( $c_{max} = 72.7$ and AUC <sub>0-24 h</sub> = 628), HHC ( $c_{max} = 31.7$ and AUC <sub>0-24 h</sub> = 323), THC sulphate ( $c_{max} = 24$ and AUC <sub>0-24 h</sub> = 164), HHC sulphate ( $c_{max} = 48$ and AUC <sub>0-24 h</sub> = 528), and total curcuminoids ( $c_{max} = 209$ and AUC <sub>0-24 h</sub> = 2030)
Liposomal and micellar system:	
Doctor's Best Curcumir	n Phytosome (lecitin, 18% curcuminoids)
(500 mg; 15)[186] Lipocurc™ (Lyposomal curcumin 80–400 mg/m <sup>2</sup> ; 50)[187]	Curcumin ( $c_{max} = 69$ and AUC <sub>0-24 h</sub> = 187) Curcumin ( $C_{max} = 42-2358$ ng/mL) and THC ( $C_{max} = 41-265$ )
Intravenous	
Liquid micellar preparation (NOV) c (60 mg; NovaSOL,	Free curcumin ( $c_{max} = 9.1$ and AUC <sub>0-24 h</sub> = 25.2), curcumin sulphate ( $c_{max} = 62.7$ and AUC <sub>0-24 h</sub> = 296), curcumin glucuronide ( $c_{max} = 295$ and AUC <sub>0-24 h</sub> = 491), free DMC ( $c_{max} = 1.3$ and AUC <sub>0-24 h</sub> = 2.6), free BDMC ( $c_{max} = 0.2$ and AUC <sub>0-24 h</sub> = 0.1), THC glucuronide ( $c_{max} = 711$ and AUC <sub>0-24 h</sub> = 3050) HHC ( $c_{max} = 330$ and AUC <sub>0-24 h</sub> = 1790), THC sulphate ( $c_{max} = 24$ and AUC <sub>0-24 h</sub> = 164), HHC sulphate ( $c_{max} = 63$ and AUC <sub>0-24 h</sub> = 296), and total curcuminoids ( $c_{max} = 1760$ and AUC <sub>0-24 h</sub> = 8540)

Ρ.	Dytrych	et	al.
----	---------	----	-----

Table 3 (continued)	
(Dose Name; Subject) Lit.	Physiological curcuminoid concentration <b>Curcuminoid</b> (c <sub>max</sub> /ng/mL and AUC /ng.h/mL)
AQUANOVA; 30)	
[167] BioCurc (lauryl macros	ol-32 glycerides polysorbate-20 DL-alpha-tocopherol) 400 mg
BioCurc (76 mg,	Free curcumin ( $c_{max} = 2 ng/mL$ ), curcumin glucuronide ( $c_{max} 300 ng/mL$ ), curcumin sulphate ( $c_{max} = 20 ng/mL$ ), and total curcumin* ( $c_{max} = 2 ng/mL$ ).
Boston BioPharm	277.24 ng/mL)
(Southlake, TX;	
12)[188]	
(750 mg	Currentian ( $c_{max} = 691$ and AU( $c_{-24}$ h = 1549) <sup>*</sup> , JMC ( $c_{max} = 96.8$ and AU( $c_{-24}$ h = 366), BDMC ( $c_{max} = 24$ and AU( $c_{-24}$ h = 128), and total curcuminoids ( $c_{-24}$ h = 1808)
Pharmako	$(c_{max} - 60)$ and $AO(0-24h - 1696)$
Biotechnologies,	
New South Wales;	
5)[189]	
WDTE60N/	Free curcumin ( $c_{max} = 0.438.9$ and AUC <sub>0-24</sub> h = 0.807), total curcumin ( $c_{max} = 435$ and AUC <sub>0-24</sub> h = 232), DMC ( $c_{max} = 35$ and AUC <sub>0-24</sub> h = 1/1), BDMC ( $c_{max} = 35$ and AUC <sub>0-24</sub> h = 1/1), BDMC ( $c_{max} = 35$ and AUC <sub>0-24</sub> h = 1/1), BDMC ( $c_{max} = 35$ and AUC <sub>0-24</sub> h = 1/1), BDMC ( $c_{max} = 35$ and AUC <sub>0-24</sub> h = 1/1), BDMC ( $c_{max} = 35$ and AUC <sub>0-24</sub> h = 1/1), BDMC ( $c_{max} = 35$ and AUC <sub>0-24</sub> h = 1/1), BDMC ( $c_{max} = 35$ and AUC <sub>0-24</sub> h = 1/2).
(150 mg	$(t_{max} - 1.7 \text{ and } AOC_{0-24}h - 0.2)$ , find $(t_{max} - 0.141 \text{ and } AOC_{0-24}h - 1.150)$ , and total curcummonds $(t_{max} - 0.24 \text{ and } AOC_{0-24}h - 425.1)$
curcuminoids,	
Inventia	
Healthcare Ltd.,	
India; 7)[179]	Computing $(a_{1} - 2704)$ and AUC = 12147 pND DMC $(a_{2} - 440)$ and AUC = 1224 pMD and RDMC $(a_{2} - 10.4)$ and AUC = 27 pMD
(500 mg: 23)[184]	Curcumm $(t_{max} = 3/04 \text{ dnu} AOC_{0.24} \text{ h} = 12147 \text{ mv})$ , Dive $(t_{max} = 440 \text{ dnu} AOC_{0.24} \text{ h} = 1224 \text{ mv})$ and Dive $(t_{max} = 10.4 \text{ dnu} AOC_{0.24} \text{ h} = 37 \text{ mv})$
Turmeric matrix	
formulation	
Volatile oils of	$\textbf{Curcumin} (c_{max} = 0.5 \text{ and } AUC_{0-12 h} = 5.8), \textbf{DMC} (c_{max} = 0.2 \text{ and } AUC_{0-12 h} = 2.2), \textbf{BDMC} (c_{max} = 0.3 \text{ and } AUC_{0-12 h} = 2.6), \textbf{THC} (c_{max} = 0.0 \text{ and } AUC_{0-12 h} = 2.6), \textbf{THC} (c_{max} = 0.6), \textbf{THC} (c_$
turmeric rhizome	0.3), and total curcuminoids ( $c_{max} = 0.1$ and AUC <sub>0-12 h</sub> = 10.9)
oils of turmeric	
rhizome DolCas	
Biotech 15 (23.0	
± 2.4 years))[178]	
BCM-95®CG (Biocurcus	maxIM) $= 47.54$ and AUC $= 117$
BCM-95 (500 mg:	Curcumm $(c_{max} = 47.54 \text{ and } AOc_{0-24} \text{ h} = 117)$
Volatile oil	
formulation; 15)	
[186]	
BCM-95 (279 mg	Curcumin ( $C_{max} = 45.0$ )
turmeric essential	
oils; 9)[185]	
(2000 mg; 8)[177]	Curcumin ( $c_{max} = 553$ and AUC <sub>0-inf</sub> = 3050)
Cureit <sup>™</sup> /Acumin	<b>Curcumin</b> ( $c_{max} = 170$ and AUC <sub>0-24 h</sub> = 824.9)
(500 mg; Aurea Biolab C	
completely natural	
turmeric matrix	
formulation NTM	
formulation;15)	
Cureit Cansules	Curcumin ( $c_{max} = 434$ and AUC <sub>0.9.b</sub> = 904)
(500 mg; 12)[180]	
(376 mg, hydrophilic	$\textbf{Curcumin} \ (c_{max} = 273 \ \text{and} \ \text{AUC}_{0-12 \ h} = 307), \ \textbf{DMC} \ (c_{max} = 5.4 \ \text{and} \ \text{AUC}_{0-12 \ h} = 54.4), \ \textbf{BDMC} \ (c_{max} = 1.4 \ \text{and} \ \text{AUC}_{0-12 \ h} = 10.2), \ \text{THC} \ (c_{max} = 0.7 \ \text{and} \ \text{AUC}_{0-12 \ h} = 10.2), \ \textbf{C} \ (c_{max} = 1.4 \ \text{and} \ \text{AUC}_{0-12 \ h} = 10.2), \ \textbf{C} \ (c_{max} = 1.4 \ \text{and} \ \text{AUC}_{0-12 \ h} = 10.2), \ \textbf{C} \ (c_{max} = 1.4 \ \text{and} \ \text{AUC}_{0-12 \ h} = 10.2), \ \textbf{C} \ (c_{max} = 1.4 \ \text{and} \ \text{AUC}_{0-12 \ h} = 10.2), \ \textbf{C} \ (c_{max} = 1.4 \ \text{and} \ \text{AUC}_{0-12 \ h} = 10.2), \ \textbf{C} \ (c_{max} = 1.4 \ \text{and} \ \text{AUC}_{0-12 \ h} = 10.2), \ \textbf{C} \ (c_{max} = 1.4 \ \text{and} \ \text{AUC}_{0-12 \ h} = 10.2), \ \textbf{C} \ (c_{max} = 1.4 \ \text{and} \ \text{AUC}_{0-12 \ h} = 10.2), \ \textbf{C} \ (c_{max} = 1.4 \ \text{and} \ \text{AUC}_{0-12 \ h} = 10.2), \ \textbf{C} \ (c_{max} = 1.4 \ \text{and} \ \text{AUC}_{0-12 \ h} = 10.2), \ \textbf{C} \ (c_{max} = 1.4 \ \text{and} \ \text{AUC}_{0-12 \ h} = 10.2), \ \textbf{C} \ (c_{max} = 1.4 \ \text{and} \ \text{AUC}_{0-12 \ h} = 10.2), \ \textbf{C} \ (c_{max} = 1.4 \ \text{and} \ \text{AUC}_{0-12 \ h} = 10.2), \ \textbf{C} \ (c_{max} = 1.4 \ \text{and} \ \text{AUC}_{0-12 \ h} = 10.2), \ \textbf{C} \ (c_{max} = 1.4 \ \text{and} \ \text{AUC}_{0-12 \ h} = 10.2), \ \textbf{C} \ (c_{max} = 1.4 \ \text{and} \ \text{AUC}_{0-12 \ h} = 10.2), \ \textbf{C} \ (c_{max} = 1.4 \ \text{and} \ \text{AUC}_{0-12 \ h} = 10.2), \ \textbf{C} \ (c_{max} = 1.4 \ \text{and} \ \text{AUC}_{0-12 \ h} = 10.2), \ \textbf{C} \ (c_{max} = 1.4 \ \text{and} \ \text{AUC}_{0-12 \ h} = 10.2), \ \textbf{C} \ (c_{max} = 1.4 \ \text{and} \ \text{AUC}_{0-12 \ h} = 10.2), \ \textbf{C} \ (c_{max} = 1.4 \ \text{and} \ \text{AUC}_{0-12 \ h} = 10.2), \ \textbf{C} \ (c_{max} = 1.4 \ \text{and} \ \text{AUC}_{0-12 \ h} = 10.2), \ \textbf{C} \ (c_{max} = 1.4 \ \text{and} \ \text{AUC}_{0-12 \ h} = 10.2), \ \textbf{C} \ (c_{max} = 1.4 \ \text{and} \ \text{AUC}_{0-12 \ h} = 10.2), \ \textbf{C} \ (c_{max} = 1.4 \ \text{and} \ \text{AUC}_{0-12 \ h} = 10.2), \ \textbf{C} \ (c_{max} = 1.4 \ \text{and} \ \text{AUC}_{0-12 \ h} = 10.2), \ \textbf{C} \ (c_{max} = 1.4 \ \text{and} \ \text{AUC}_{0-12 \ h} = 10.2), \ \textbf{C} \ (c_{max} = 1.4 \ \text{and} \ \text{AUC}_{0-12 \ h} = 10.2), \ \textbf{C} \ (c_{max} = 1.4 \ \text{and} \ \text{AUC}_{0-12 \$
carrier, cellulosic	$AUC_{0-12 h} = 7.7$ ), and total curcuminoids ( $c_{max} = 34.9$ and $AUC_{0-12 h} = 380$ )
derivatives and	
antioxidants: (lot	
number	
CU20DNS1-008/	
009 OmniActive	
Technologies: 15	
$(23.0 \pm 2.4 \text{ years})$	
[178]	
Cavacurmin® γ-CD	Curcumin ( $c_{max} = 73$ and AUC <sub>0-12 h</sub> = 328), DMC ( $c_{max} = 12$ and AUC <sub>0-12 h</sub> = 51), BDMC ( $c_{max} = 1.4$ and AUC <sub>0-12 h</sub> = 9.4), and total curcuminoids ( $c_{max} = 1.2$ and AUC <sub>0-12 h</sub> = 9.4), and total curcuminoids ( $c_{max} = 1.2$ and AUC <sub>0-12 h</sub> = 9.4), and total curcuminoids ( $c_{max} = 1.2$ and AUC <sub>0-12 h</sub> = 9.4), and total curcuminoids ( $c_{max} = 1.2$ and AUC <sub>0-12 h</sub> = 9.4), and total curcuminoids ( $c_{max} = 1.2$ and AUC <sub>0-12 h</sub> = 9.4), and total curcuminoids ( $c_{max} = 1.2$ and AUC <sub>0-12 h</sub> = 9.4), and total curcuminoids ( $c_{max} = 1.2$ and AUC <sub>0-12 h</sub> = 9.4), and total curcuminoids ( $c_{max} = 1.2$ and AUC <sub>0-12 h</sub> = 9.4), and total curcuminoids ( $c_{max} = 1.2$ and AUC <sub>0-12 h</sub> = 9.4), and total curcuminoids ( $c_{max} = 1.2$ and AUC <sub>0-12 h</sub> = 9.4), and total curcuminoids ( $c_{max} = 1.2$ and AUC <sub>0-12 h</sub> = 9.4), and total curcuminoids ( $c_{max} = 1.2$ and AUC <sub>0-12 h</sub> = 9.4), and total curcuminoids ( $c_{max} = 1.2$ and AUC <sub>0-12 h</sub> = 9.4), and total curcuminoids ( $c_{max} = 1.2$ and AUC <sub>0-12 h</sub> = 9.4), and total curcuminoids ( $c_{max} = 1.2$ and AUC <sub>0-12 h</sub> = 9.4), and total curcuminoids ( $c_{max} = 1.2$ and AUC <sub>0-12 h</sub> = 9.4), and total curcuminoids ( $c_{max} = 1.2$ and AUC <sub>0-12 h</sub> = 9.4), and total curcuminoids ( $c_{max} = 1.2$ and AUC <sub>0-12 h</sub> = 9.4), and total curcuminoids ( $c_{max} = 1.2$ and AUC <sub>0-12 h</sub> = 9.4), and total curcuminoids ( $c_{max} = 1.2$ and AUC <sub>0-12 h</sub> = 9.4), and total curcuminoids ( $c_{max} = 1.2$ and AUC <sub>0-12 h</sub> = 9.4), and total curcuminoids ( $c_{max} = 1.2$ and AUC <sub>0-12 h</sub> = 9.4), and total curcuminoids ( $c_{max} = 1.2$ and AUC <sub>0-12 h</sub> = 9.4), and total curcuminoids ( $c_{max} = 1.2$ and AUC <sub>0-12 h</sub> = 9.4), and total curcuminoids ( $c_{max} = 1.2$ and AUC <sub>0-12 h</sub> = 9.4), and total curcuminoids ( $c_{max} = 1.2$ and AUC <sub>0-12 h</sub> = 9.4), and total curcuminoids ( $c_{max} = 1.2$ and AUC <sub>0-12 h</sub> = 9.4), and total curcuminoids ( $c_{max} = 1.2$ and AUC <sub>0-12 h</sub> = 9.4), and total curcuminoids ( $c_{max} = 1.2$ and AUC <sub>0-12 h</sub> = 9.4), and total curcuminoids ( $c_{max} = 1.2$ and AUC <sub>0-1</sub>
(348 mg curcumin, 21.6 mg DMC and	8/ and $AUC_{0-12}h = 389$
2.4 BDMC; 12)	
[190]	
Theracurmin® (Ghatti	gum/glycerin/Lipids/hydroxymethyl, maltose)
CR-033 P (90 mg,	<b>Curcumin</b> ( $C_{max} = 33.78$ and $AUC_{0-12 h} = 177.55$ )
Shizuoka, Japan:	
24)[181]	
	(continued on next page)

Ρ.	Dytrych	et	al
----	---------	----	----

Table 3 (continued)

(Dose Name; Subject)	Physiological curcuminoid concentration
Lit.	Curcuminoid (c <sub>max</sub> /ng/mL and AUC /ng.h/mL)
CR031P (30 mgx3, BIHOLON, Toyama, Japan; 24)x3[181]	<b>Curcumin</b> ( $C_{max} = 30.75 \text{ AUC}_{0-12 \text{ h}} = 148$ )
(182 mg; 9)[185]	<b>Curcumin</b> ( $C_{max} = 231.5$ (2 h) 167 (6 h) and 66.4 (24 h))
Drinkable formulation (30 mg/100 mL; 24 (23–32 years)) [191],	Curcumin (C <sub>max</sub> = 25.5 and AUC <sub>0-8 h</sub> = 121)
Turmipure GOLD	$\label{eq:Free curcumin} (c_{max} = 15.4 \mbox{ and } AUC_{0-24 \mbox{ h}} = 25.7), \mbox{ curcumin sulphate } (c_{max} = 63 \mbox{ and } AUC_{0-24 \mbox{ h}} = 575), \mbox{ curcumin glucuronide } (c_{max} = 42.5 \mbox{ and } AUC_{0-24 \mbox{ h}} = 575), \mbox{ curcumin glucuronide } (c_{max} = 42.5 \mbox{ and } AUC_{0-24 \mbox{ h}} = 575), \mbox{ curcumin glucuronide } (c_{max} = 42.5 \mbox{ and } AUC_{0-24 \mbox{ h}} = 575), \mbox{ curcumin glucuronide } (c_{max} = 42.5 \mbox{ and } AUC_{0-24 \mbox{ h}} = 575), \mbox{ curcumin glucuronide } (c_{max} = 42.5 \mbox{ and } AUC_{0-24 \mbox{ h}} = 575), \mbox{ curcumin glucuronide } (c_{max} = 42.5 \mbox{ and } AUC_{0-24 \mbox{ h}} = 575), \mbox{ curcumin glucuronide } (c_{max} = 42.5 \mbox{ and } AUC_{0-24 \mbox{ h}} = 575), \mbox{ curcumin glucuronide } (c_{max} = 42.5 \mbox{ m}), \mbox{ curcumin glucuronide } (c_{max} = 42$
(90 mg; A dried colloidal	= 226), free DMC ( $c_{max} = 2.7$ and AUC <sub>0-24 h</sub> = 9.4), free BDMC ( $c_{max} = 0.4$ AUC <sub>0-24 h</sub> = 0.6), THC glucuronide ( $c_{max} = 277$ and AUC <sub>0-24 h</sub> = 2030), HHC glucuronide ( $c_{max} = 165$ and AUC <sub>0-24 h</sub> = 1540), THC sulphate ( $c_{max} = 61$ and AUC <sub>0-24 h</sub> = 211), HHC sulphate ( $c_{max} = 197$ and AUC <sub>0-24 h</sub> = 1830), and Cloud AUC <sub>0-24 h</sub> = 1000 ( $c_{max} = 1000$ ( $c_{max} = 1000$ ), THC sulphate ( $c_{max} = 1000$ ( $c_{max} = 1000$ ), and AUC <sub>0-24 h</sub> = 1000 ( $c_{max} = 1000$ ), and AUC <sub>0-24 h</sub> = 1000 ( $c_{max} = 1000$ ), and AUC <sub>0-24 h</sub> = 1000 ( $c_{max} = 1000$ ), and AUC <sub>0-24 h</sub> = 1000 ( $c_{max} = 1000$ ), and AUC <sub>0-24 h</sub> = 1000 ( $c_{max} = 1000$ ), and AUC <sub>0-24 h</sub> = 1000 ( $c_{max} = 1000$ ), and AUC <sub>0-24 h</sub> = 1000 ( $c_{max} = 1000$ ), and AUC <sub>0-24 h</sub> = 1000 ( $c_{max} = 1000$ ), and AUC <sub>0-24 h</sub> = 1000 ( $c_{max} = 1000$ ), and AUC <sub>0-24 h</sub> = 1000 ( $c_{max} = 1000$ ), and AUC <sub>0-24 h</sub> = 1000 ( $c_{max} = 1000$ ), and AUC <sub>0-24 h</sub> = 1000 ( $c_{max} = 1000$ ( $c_{max} = 1000$ ), and AUC <sub>0-24 h</sub> = 1000 ( $c_{max} = 1000$ ), and AUC <sub>0-24 h</sub> = 1000 ( $c_{max} = 1000$ ( $c_{max} = 1000$ ), and AUC <sub>0-24 h</sub> = 1000 ( $c_{max} = 1000$ ( $c_{max} = 1000$ ), and AUC <sub>0-24 h</sub> = 1000 ( $c_{max} = 1000$ ( $c_{max} = $
suspension, IPG quillaja extract, sunflower oil, and acacia gum containing, Naturex; 30)[167]	total curcuminoids ( $c_{max} = 6/8$ and $AUC_{0-24 h} = 6520$ )
Curcuwin Ultra+ Omnia	Active Health Technologies
(50 mg, lot number CU20DNS3–096 (04)/069; 23) [182]	<b>Curcumin</b> ( $c_{max} = 28.36$ and AUC <sub>0-6 h</sub> = 86), <b>DMC</b> ( $c_{max} = 4.33$ and AUC <sub>0-6 h</sub> = 9.35), <b>BDMC</b> ( $c_{max} = 1.05$ and AUC <sub>0-6 h</sub> = 0.49), <b>THC</b> ( $c_{max} = 13.98$ and AUC <sub>0-6 h</sub> = 41), and <b>total curcuminoids</b> ( $c_{max} = 43.83$ and AUC <sub>0-6 h</sub> = 133)
(100 mg, lot number CU20DNS3–096 (04)/073))[182]	Curcumin ( $c_{max} = 54.1$ and AUC <sub>0-6 h</sub> = 162), DMC ( $c_{max} = 7.98$ and AUC <sub>0-6 h</sub> = 20.28), BDMC ( $c_{max} = 1.15$ and AUC <sub>0-6 h</sub> = 0.82), THC ( $c_{max} = 26.24$ and AUC <sub>0-6 h</sub> = 81), and total curcuminoids ( $c_{max} = 86.98$ and AUC <sub>0-6 h</sub> = 274)
Patients	
Micronized powder (Na	tional Cancer Institute); Smoker ( $\geq$ 50 years and $\geq$ 8 rectal ACF)[192]
(2 g/day; 30 days;	<b>Plasma:</b> curcumin ( $c_{max} = 7.3/3.8$ ; pre/post treatment)
20)[192] (4 g/day, 30 days;) [192]	<b>Rectal mucosa: Curcumin</b> ( $c_{max} = ND/4.21$ ; pre/ post treatment) <b>Plasma: curcumin</b> ( $c_{max} = 15.8/78.5$ ng/mL; pre/post treatment), <b>Rectal mucosa: curcumin</b> ( $c_{max} = 3.8/4.5$ ; pre/ post treatment)
Capsule (Laboratorios A	dmira S.L. (Alcantarilla, Murcia, Spain; 65 mg transresveratrol 190 mg turmeric extract 125 mg flaxseed extract and 125 mg red clover extract; 296 mg
phenolic compounds)	; Breast cancer patients)[193]
(3 x capsule (121 mg curcumin, 15 mg BDMC and 43 mg DMC) / day from	Plasma: free curcumin = 5.26 nM, BDMC = 3.89 nM and DMC = 2.45 nmol Normal mammary tissue: free curcumin = 198 pmol/g ~ 1 $\mu$ M, BDMC = 8.78 pmol/g ~ 0.05 $\mu$ M and DMC = 44 pmol/g 0.2 $\mu$ M) Malignant tissue: free curcumin = 109 pmol/g~ 0.2 $\mu$ M, BDMC = 2.73 pmol/g~5 nM and DMC = 13 pmol/g~ 22 nM)
(diagnosis to surgery and 4–6 h before surgery); 39)[193]	

could be comparative with its concentration in the tumour. In this case, it could be assumed that most effective clinically tested oral formulations could achieve a micromolar curcumin concentration in the tumour tissue. It cannot be neglected, that similar concentrations could also be expected in the normal healthy tissue. On the other hand, curcumin toxicity is low, and any side effects are tolerable. In addition, curcumin displayed a protective effect against some disorders associated with carcinogenesis and side effects of used anticancer therapies. It is implied, that its distribution in the normal tissue could be a part of its therapeutic effects. The question is whether if the curcumin intratumoural concentration is reached it may already display antimetastatic effects. As far as we can predict from in vitro studies (e.g., table 1), some suggest that this assumption is possible [69,88-92,94,97,99,102-104,109,111-113,117, 118,120,121]. However, to achieve full clinical potential of curcuminoids, it is sometimes necessary to increase their bioavailability and bioactivity. Therefore, other curcuminoid formulations such as nanoformulation for metastasis suppression are intensively developed.

Although the main goal in the formulation of curcuminoids is their accumulation in the tumour tissue. Some studies suggest, their benefit could also be achieved outside of the tumour tissue. For example, NANOCurc (lipid-polymer nanoparticles;  $d=177.3 \pm 6.2$  nm) significantly increases curcumin efficacy in the reduction of cancer cell adherence and TNF- $\alpha$  activation of endothelial cells [60]. NANOCur display significantly higher cellular uptake after TNF- $\alpha$  activation

against free curcumin.

## 4. Curcuminoid administration: influence of curcumin administration route on its bioavailability and therapeutic effectivity

The curcuminoid levels in the blood, or tumour tissue and their clinical/biological effects are strongly dependent on the method of their administration. Intravenous administration is often used in biological studies, but not as often as in clinical studies. Transferability of results can be strongly limited. On the other hand, in the case of nanodrugs such as liposomal formulations, intravenous application is a common route [200]. It should be mentioned that intravenous formulation of the curcuminoid (Lipocurc<sup>TM</sup>) was also studied in clinical trials [187]. The highest observed concentration of curcumin and tetrahydroxycurcumin in the blood was 2.3 and 0.265 µg/mL, respectively. This result suggests that proportions of metabolically transformed curcumin are sometimes lower than with oral application. It is important to note, that in the case of oral application, curcumin level was significantly lower ( $c_{max} = 0.553$  and AUC<sub>0-inf</sub> = 3050 ng/mL) [177].

Although oral application is more comfortable for the patients, the intravenous route could lead to microsomal concentration of curcumin in the blood (6.6  $\mu$ M) and tumour tissue and thereby higher therapeutic efficiency. Nevertheless, curcumin blood level/therapeutic efficiency

could be significantly increased by its application in the form of metal complexes [201]. In the case of copper curcumin, its highest concentration was 125.8 µg/mL (349 µM). Considering these results obtained in the mice model with metastatic breast cancer to be transferable, application of copper curcumin noticeably slowed down tumour growth, delayed metastasis, and prolonged survival in mice. Nevertheless, it was not enough to fully eradicate the tumour. And it is hard to imagine that intravenously applied curcumin could reach even higher concentrations. Here it is necessary to note curcumin's possible toxicity. For example, Lipocurc was well tolerated when administered intravenously, but at dosages  $\geq$  120 mg/m<sup>2</sup>, transient echinocyte formation in red blood cells was observed with a concomitant increase in mean cell volume [187]. Curcumin is also an inhibitor of the hERG K+ channel [202] and thereby possible cardiotoxicity cannot be neglected [203]. Ranjan et al. reported, that high curcumin doses, such as 6 µM (in the case of liposomal formulation) can decrease its activity by up to 30% [202]. The above suggests that increasing physiological concentration of curcumin can increase its effectivity, but cannot be excluded, that serious side effects, may occur at lower curcumin concentrations than full tumour eradication. And formulation methods and drug delivery systems can strongly increase curcumin bioavalibity and bioactivity, but not unlimited.

It also suggests that curcumin cannot treat cancer by itself. However, numerous high impact studies suggest that correctly applied curcumin could be a promising agent for the enhancement of anticancer treatments. For example, intraparental application of NanoCurc displayed maximal concentration 17 mg/mL (47  $\mu$ M) in the mice model [204]. Nevertheless, its combination with gemcitabine resulted in strong effects against pancreatic carcinoma. Tumour growth was blocked and any micrometastases in lung, lymph nodes or peritoneum were not found. In the case of both single agents, micrometastases were found in the lymph nodes. This effect could be explained by higher intra-tumoural curcumin levels (>2.5  $\mu$ g/g of tissue), which was enough for the potent inhibition of NF-kB binding to DNA after applying both agents. In contrast, only applying NanoCurc<sup>TM</sup> displayed ~4-fold lower intratumoural curcumin levels and both single applications did not display significant inhibition of NF-kB functionality.

The choice of the optimal route is not universal and strongly depends on the type of cancer. More detailed information about methods of curcumin administration (oral, inhalation and intratumoural) are provided in the next subchapters.

## 4.1. Oral administration

Oral administration has been the main method of drug administration for thousands of years, with strong advantages such as safety, good tolerance and compliance, low treatment costs, and convenience to the patients [205]. In this case, the relationship between curcuminoids and gut microflora should be considered. It could be explained by the paradox between curcuminoid limited bioavailability and observed anticancer effects.

After oral administration, curcumin can be metabolized by reduction (catalysed by enzymes of phase 1), conjugation (catalysed by enzymes of phase II), and microbial or non-enzymatic degradation (Fig. 4) [206].

During phase I metabolism, curcumin is mainly converted to tetrahydrocurcumin, hexahydrocurcumin, and octahydrocurcumin [207, 208]. During phase II, their glucuronide and sulfate O-conjugated metabolites are produced [207–209]. Nevertheless curcuminoids, glucuronide and sulfate conjugates are can be to simpler phenolic compounds such as ferulic acid and in the cecum and colon [207,208]. However, microbial transformation of natural curcuminoids can mean a loss of biological activity, but not necessarily [210]. In the case of tetrahydroxycurcumin and octahydrocurcumin, some high impact studies suggest, that their anticancer effectivity could be at least comparable with natural curcumin [211–214]. On the other hand, curcumin, unlike tetrahydrocurcumin, can directly interact with the STAT3 SH2 domain, suppress its dimerization and subsequent nuclear translocalization [215]. In the case of curcumin sulfate and glucuronide curcumin, their possible inhibitory effects against EGFR and NF- $\kappa$ B cannot be excluded [216] Nevertheless, glucuronide curcumin could have significantly lower cellular uptake compared to natural curcumin. Jamil et al. reported, that MDBA-231 human breast cancer cells can metabolize curcumin to curcumin sulphate and subsequently excrete it from the cells [217]. On the other hand, tumours can produce sulfatase [218], which can convert curcumin sulphate and glucuronide to natural curcumin [219]. Besides nanoparticles could protect transported curcuminoids against gut microbiota. Nevertheless, the possibility of crossing the intestinal barrier must be taken into account [220].

On the other hand, curcumin accumulated in the gastrointestinal tract displayed strong effects on intestestinal microflora (microbial richness, diversity and composition) [221-223]. During carcinogenesis, levels of Alistipes, Fusobacteria, Porphyromonadaceae, Staphvlococcaceae, Coriobacteridae, Methanobacteriales and Akkermansia spp. are enhanced, while others are permanently reduced, such as Lactobacillus, Bifidobacterium, Ruminococcus, Roseburia, Faecalibacterium spp. and Treponema [224]. Curcumin application is associated with the increase in Lactobacillus, Bifidobacterium, and butyrate producing bacteria and the decrease in Prevotellaceae, Coriobacterales, Enterobacteria, and Enterococcus [221]. On the molecular level a decrease in production of amino acids and inflammatory factors and an increase in butyrate levels can be expected Dependent on the human condition, intestinal microflora can produce various different metabolites (reduced, demethylated form hydroxylated forms of curcumin). [208] Possible alternative mechanisms are shown in Fig. 5.

However, the question is whether all this is sufficient for effective antimetastatic therapy. Results obtained from the in vitro studies could suggest, that orally applied lipid formulations could be associated with potent antimetastatic effects, especially in combination with other therapeutic modalities. For example, in the case of the mice model with metastatic breast cancer, phospholipid nanoparticles containing curcumin and photothermal dyes reduce pulmonary nodules by more than an order of magnitude [134]. Applied curcumin dose was 50 mg/kg per 2 days ( $\sim$ 3.5 g for human). In the clinical trials, the orally applied Meriva formulation was associated with the decrease in levels of some metastatic factors (e.g., IL-6, TNF- $\alpha$  and TGF- $\beta$ ). [41] Similarly it was reported that administration of BCM-95 (BIO-CURCUMIN®) in dose 500 mg/d can reduce NF-kB activity in PBMCs and blood levels TNF- $\alpha$  of in patients with none-alcoholic fatty liver disease [225]. In clinical trials, maximum micromolar concentration levels were achieved with oral administration of both curcuminoid preparations. Dhillon et al. reported, that anti-inflammatory effects of curcuminoids can be sometimes observed during lower plasma concentration (~tens of nanomoles/l) [226].

In the case of digestive tract cancers, oral curcumin administration could display a significant therapeutic impact [145,146,227]. For example, patients with colorectal cancer after curcumin administration (3.6 g/day, 7 days) displayed a significant decrease in 3-(2-deoxy-beta-di-erythro-pentafuranosyl)-pyr[1,2-alpha]-purin-10(3 H)-one (M (1)G; DNA oxidation product) and COX-2 protein level [146]. Combination of curcumin and triple therapy significantly reduce all scores of active, chronic and endoscopic inflammation in patients with chronic gastritis compared to the baseline and to the triple therapy group [145]. This effect was associated with a decrease in oxidative markers and DNA damage and an increase in the antioxidative capacity of gastric mucosa. Another significant benefit of oral curcumin formulations could be related to the prevention of oncological diseases, for example by reducing DNA damage and oxidative stress [144–146].

It implies, that micromolar blood levels of curcuminoids available in oral formulations could be sufficient to significantly reduce tumour inflammation, decrease cancer risk and inhibit some metastatic mechanisms. Relevant in vivo studies suggest that oral administration of suitably designed curcumin formulations in combination with other cancer agents could lead to a stronger antitumour and antimetastatic

## effects.

#### 4.2. Inhalation administration

The inhalation administration is a promising alternative strategy for the delivery of anticancer agents into lung tumours. Agents could bypass the intracellular and extracellular drug-metabolizing enzyme in the gastrointestinal tract and liver and their therapeutic index would be significantly higher. [228] On the other hand, inhalation administration may be limited by the toxicity of the injected substances to lung tissue, their rapid clearance and lipophilicity. Here it may be interesting to note, that curcumin inhalation formulation displays favourable toxicity to lung cells and tissue. In addition, it was reported that there may be a protective effect against chemo and radiotherapy [229,230].

Extrapulmonary distribution is dependent on the aerodynamic diameter and on the patients breathing pattern [228]. Large particles (d> 5  $\mu$ m) are filtered by inertial impaction in the upper airways (mouth, trachea and main bronchi). Medium-sized particles (d = 1–5  $\mu$ m) are gravitationally distributed to the central and distal tracts. Small particles (d = 0.1 – 1  $\mu$ m) are mostly exhaled. Ultra-small particles are moved (d < 100 nm) by random Brownian motion to alveolar regions, while particles that are 10 nm and smaller stay in the tracheo-bronchial region due to their high diffusion coefficients [231]. Aerodynamic characteristic of curcumin powder (MMAD 3.81  $\mu$ m) implies its absorption into the lung tissue [232]. A problem could be its absorption through the lung epithelial barrier. Drug absorption in the lung epithelia is primarily dependent on their molecular weight and

lipophilicity [228,233]. In general, absorption of small hydrophobic molecules takes minutes, while in the case of large hydrophilic molecules it takes hours. Particles bigger than 200 nm are usually captured by macrophages in the lung tissue. Formulations of curcuminoids can significantly improve distribution and control their uptake into the bloodstream/tumour tissue [232]. Some studies imply that this strategy could lead to a significant improvement in curcumin pharmaceutical efficiency (Table 4). Nevertheless, other clinical trials are requested for validation.

Results obtained from relevant in vivo studies suggest that there is high potential in this strategy [229,237,239,241]. In primary lung cancer models (rats exposed to 3-methylcholanthrene and diethylnitrosamine), the inhalation of curcumin nanoparticles (liposomal curcumin dry powder) sometimes resulted in decreased levels of MDA, TNF- $\alpha$  and VEGF and activation of caspase-3 [239]. Its antitumour effectivity was significantly higher than in the case of curcumin powder and gemcitabine. This approach could also be used for the treatment of lung metastases. Su et al. reported that silica nanoparticles modified curcumin significantly and suppressed the formation of lung metastases in the melanoma bearing mice [237]. This effect was associated with the reduction of Il-6 and IL-8 in the lung tissue. However, the decrease in TNF- $\alpha$  levels was significantly higher than in the case of curcumin powder. It is probably caused by higher phagocytosis of curcumin nanoparticles compared to curcumin powder. In addition, the inhalation administration with a suitable curcumin formulation displayed potent protective effects on the non-tumour tissue lung tissue, for example alleviation of inflammation, or radiation pneumonitis [229].



Fig. 4. Metabolic transformation of curcumin (classical schema).



Fig. 5. Alternative models of curcumin transformation by gut microflora.

### 4.3. Intratumoural therapy

Effective methods of antimetastatic treatment could be based on curcumin intratumoural therapy. The first successful intratumoural therapy was based on bacterial toxins which were administered over 100 years ago, by Dr. William Coley [242,243]. In the present time IT therapy is a intensively studied therapeutic method [244]. One key factor, which controls effectivity of IT therapy is drug distribution in correspondence with the molecular size, charge, and other properties in the tumour and the surrounding tissues. Agents must target the tumour tissue whole and avoid clearance by lymphatic drainage and cell absorption. Compared to proteins, small molecules displayed high potency and improved tissue penetration. Nevertheless, their tumour retention can be significantly improved by suitable formulation (e.g., liposomal carriers, polymeric and hydrogels) [245–247].

Some high impact studies suggest, that suitably designed intratumoural formulations of curcumin could significantly improve its intratumoural concentration and thus its therapeutic efficacy [247–250]. For example, curcumin intratumoural administration (20 mg/kg; mice with H22 tumour) doubles apoptosis rates in the tumour tissue [250]. Nevertheless, in the case of liposomal nanoformulations, curcumin apoptotic effectivity (25–30%) increased fivefold against baseline and blank liposomes. Similarly, these formulations increase Caspase-3 overexpression and decrease VEGF levels. Gao et al. reported, that peritumoural/intratumoural injections of mesoporous silica nanoparticles modified by pegylated lipid bilayer containing curcumin and paclitaxel displayed significantly higher reduction of 4T1 tumours than the corresponding intravenous application in the mice model. [251] Pharmacokinetic study (Sprague-Dawley rats; 18 mg curcumin and 3 mg paclitaxel/kg) showed that the peritumoural route is associated with significantly better curcumin pharmacokinetic parameters such as  $C_{max}$  (6.2 (16.8  $\mu$ M) vs 4.3  $\mu$ g/mL) and AUC<sub>0-8 h</sub> (89 vs 59  $\mu$ g.h/mL) compared to intravenous administration. In connection to this, it should be noted, that intraparental application of NanoCurc displayed maximal plasma concentrations of 17 mg/mL (47  $\mu$ M) in the mice model with pancreatic cancer [204].

Intratumoural application of TriCurin (combination of curcumin, epicatechins, and resveratrol) induced large necrotic areas in the human papillomavirus-positive head and neck squamous cell carcinoma in the mice model [249]. Based on in vitro studies, it can be assumed that the intratumoural levels of used substances can be in the micromolar range. On the other hand, TriCurin application (liposomal formulation of curcumin with resveratrol and epicatechin gallate) is also associated with repolarization of the TAM phenotype (from M2 to M1) and activation of NK cells, resveratrol and epicatechin gallate can also directly kill tumour cells [46], and sublethal TriCurin doses can repolarize the TAM phenotype [252]. However, we cannot expect, that the intratumoural route will always lead to a better therapeutic effect compared to the intravenous application. For example, Chang et al. reported, that intratumoural curcumin application (unformulated and liposomal formulation) displayed a significantly higher decrease in tumour volume compared to the intravenous administration [253]. However, no significant difference was observed in the case of the tumour mass.

An interesting variation of this strategy could be tumour excision and subsequent curcumin application into the tumour site. In the melanoma mice model, curcumin oil nanoemulsions with prolonged retention applied this way resulted in very strong suppression of tumour recurrence and metastatic activity [128]. The effect of natural unmodified curcumin was significantly lower.

The above clearly indicates that the intratumoural route can

## Table 4

Examples of tested inhalation curcumin formulations.

Formulation	Model	d/da	Exp. condition	Effect
Core-shell microparticles[234]	A549	3.8 µm	(0–200 µg/mL; 24–72 h)	Slow cytotoxic activity, strong
Chitosan microspheres with 2-	A549	2.58/3.8 µm	(0–20 µM; 24–72 h; in solvent)	antibacterial activity $IC_{50}$ (11.3, 9.2 and 6.5 $\mu$ M, 24, 48, and 72 h. geographically)
Chitosan microspheres with 2- HP-b-CD and doxorubicin and elastin[235]	A549	4.3/4.9 μm	(0–20 µM; 24–72 h, in solvent)	$IC_{50}$ (8.5, 5.2 and 3.4 $\mu$ M; 24 h, 48 h and 72 h),
Pectin-PVP microparticles[236]	fertilized chicken eggs	D <sub>50/90</sub> 0.99/ 2.74 µm	7 mg; 13 days/ in calf serum	↑cytotoxicity ↑angiogenesis activity
	A549		2.5–250 $\mu g/mL;$ 24 h; in solvent	↑cytotoxicity
Pluronic modified silica particles [237]	mice with lung metastasis of B16F10 melanoma	7.59 nm∕ 1–3 μm	12 mg; 7 d; aerosol inhalation	↓Inflammation (
				JIL-6 and
				$\bigcup$ IL-8; approximately half of cytokines level) and
				↑TNF-α (slightly),
Curcumin in physiological solution[238] Proliposomes with HPβCD[232]	LPS-stimulated A549 cells A549	- 126 and 150 nm	under air-liquid interface (0–100 µM; aresol) and submerged conditions (0–20 M; in the medium) 0–200 mg/mL	↑higher weight Lower IL-6 and IL-8 production; higher curcumin effectivity under air-liquid interface ↑cvtotoxicity.
		/2.10 and 3.18 µm*		↓IL-6,
				↓IL-8 and
	Albino rats		curcumin 1 mg; intratracheally and	↓TNF-α Lung tissue (
			endotracheai; 24 n	$\downarrow$ T <sub>max</sub> (from 2 to 1 h),
				$\uparrow C_{max}$ (from 1.2 to 5.0 and 3.3 µg/mL),
				$\ensuremath{\left }AUC_{0-24}$ (from 6.9 to 12.5 and 9.9 $\mu\text{g}/$ mL),
				$\ensuremath{\bigwedge} AUC_{inf}$ (from 7.2 to 19.8 and 12.3 $\mu g/$ mL),
Liposomal curcumin	A549 and BEAS-2B cells	94 nm/5.81 μm*	0–100 μm; I medium; 24 h	↑MRT (from 4.8 to 7.5 and 7.3 h) ↑cytotoxicity and
	Sprague–Dawley rat with MCA and DEN induced lung cancer		curcumin 1 mg; dry powder inhalation; 4 d	↑cytoselectivity for cancer cells Tumour factor ( ↓VEGF
				↓BCL-2,
				$\ensuremath{\bigcup} TNF\mathcar{-}\alpha)$ and oxidative stress (
				↓MDA), stimulated apoptotic signalling (
				↑Caspase-3) and
Mesoporous polydopamine nanoparticles[229]	BEAS-2B cells Sprague-Dawley rats with single dose of 15 Gy	290 nm/5.17 µm *	0–125 μm curcumin 1 mg; 1xpre-irradiation dose and 1/week post-irradiation dose/4 weeks; sprayed the drug formulations	tumour nodules apoptosis Reduction of haemorrhaging and pulmonary fibrosis, tumour factors (TNF- $\alpha$ , IL-6, IL-1 $\beta$ and TGF- $\beta$ 1) and oxidative stress (MDA) induction of SDD
Milled formulation (curcumin	umin A549 and Calu-3 cells $d_{50} = 2.2 \mu m$		0–100 µM; 72 h; In medium	$IC_{50}$ (18.9 and 22.9 $\mu$ M),
and pacificater 3:1)[230]	Beas-2B	/3.12 μm *		n.d.
				(continued on next page)

### Table 4 (continued)

Formulation	Model	d/da	Exp. condition	Effect
Microrods with TNF-α siRNA [240]	dTHP-1 and A549	2.85 μm	0–100 μg/mL-20% curcumin loading capacity	Slow cytotoxicity, strong reduction of TNF- $\!\alpha$
Curcumin RBP exosomes[241]	LPS-activated RAW264.7 cells	-	5 $\mu$ g; 4 h; in medium	cytokine level ( ↓TNF-α,
				↓IL-6 and
				$\bigcup$ IL-1 $\beta$ , and
	Mice with LPS intratracheal		5 μg; 24 h; intratracheally	↓RAGE downregulation curcumin cellular uptake Lung tissue and BAl fluids (T
				$\sqrt{NF-\alpha}$ and
				<b>I</b> L-1β)

d = mean particle da = aerodynamic diameter; Mass median aerodynamic diameter (MMAD).

significantly improve curcumin bioactivity and prolong its effect and enhance its activity against primary tumours, including its metastatic activity. However, can intratumoular administration of curcumin directly target micro and metastases? Besides the suppression of cell migration and invasiveness of the primary tumour, long-term exposition to curcumin micromolar concentrations could also hack tumour exosomes and induce formation of exosomes with antitumour activity targeted in part at (pre)metastatic sites. Another step could represent direct intratumoral application curcumin.

## 5. Limitations of curcumin applications

Despite very promising results from in vitro and in vivo studies (Tables 1 and 2), curcumin displays significant limitations for its wide clinical use in the anticancer treatment [152]. For example, after oral application of 12 g of curcumin extract, serum level of curcumin was only 57.6 ng/mL. [156] Curcumin most probably had no significant selectivity for tumour tissue [193]. This strongly suggests that oral application of curcumin cannot have a direct effect on tumour tissues/cells and its antitumour effect is mostly caused by modulation of the gut microbiota. A direct effect (cytostatic, even migrastatic) on cancer cells can be observed at least at micromolar concentrations of curcumin. Hence, some limitations of the use of curcumin in the anticancer treatment should be considered (see overview in the Table 5).

Low curcumin bioactivity is mostly caused by its low solubility (456 µg/L) [154,155] and biostability. Currently, pharmaceutically approved excipients (cyclodextrins, oligosaccharides, polymers) can be used, which can significantly improve the solubility of curcumin [159, 254,255]. Lipid nano and microparticles are another available option for improving the efficacy of curcumin. Clinical trials have shown that their applications are associated with significantly increased curcumin blood level versus unformulated curcumin extracts (Table 3). Golden milk can be easily prepared at home and lipidic curcumin formulations such as Meriva formulation are commercially available. In this context, it should be mentioned that nanoparticles with suitable size display significantly higher bioviability than low molecular compounds (Fig. 6) [256].

Recently, some advanced nanoparticles system with excellent anticancer/antimetastatic efficacy have been developed and tested (in biological studies; **table XXX**), such as gold nanoparticles containing curcumin and paclitaxel, or phospholipids nanoparticles for the transport combination of curcumin and IR780 (photothermal dyes) [134, 257]. However, their approval for clinical practice cannot be expected in the near future. Because, therapeutic usability and safety of gold nanoparticles and IR780 will need to be validated be clinical trials.

Higher curcumin efficacy can also be achieved by changing the method of administration. Oral administration of Lipocurc<sup>™</sup> displays

### Table 5

Limitations of the use of curcumin and their possible solution.

Limitation	Importance	Solution	Applicability
Low solubility Low stability	High High	Drug delivery systems/ formulations	Cyclodextrin - available lipidic formulation -available oligosaccharides -available
Insufficient bioactivity against tumour cells	Medium	Combination with other agents/ therapy	Polyphenols - available Piperidine - available
Low distribution into cancer tissue	High	Drug delivery systems	Lyposomes - available Solid nanoparticles -experimental Modified lipidic nanoparticles -experimental
		Alternative route	Intravenous - clinical trial Inhalation -experimental Intratumoural - experimental
Toxicity	Low	Drug delivery system/ nanoparticles	Too little knowledge to make decisions
Loss of therapeutic efficiency	Low	Combination with other agents/ therapy	Too little knowledge to make decisions

value of blood  $c_{max}$  0.553 ng/mL [177]. In this case of an intravenous formulation, the highest concentration observed was 2.3 µg/mL [187]. In a mouse model of lung cancer, inhalation administration significantly increases therapeutic effectivity [237]. Another step could be direct intratumoural application of curcumin. Intraparental administration of nanoformulated curcumin (Sprague-Dawley rats) displays a higher value of  $c_{max}$  than intravenous administration (6.2 vs 4.3 µg/mL) [251]. The results of the studies presented in the Table 1 suggest that archival concentrations could be associated with important antitumour/antimetastatic effects.

On the other hand, an important benefit of curcumin could be to reduce the risk of cancer or its recurrence. Clinical studies have shown that curcumin can significantly decrease the effect of various carcinogens, or alleviate various pathologies associated with carcinogenesis [144–146,192,258,259]. This suggests its application as protective agents for the subjects at risk. For this purpose, oral and inhalation administration are suitable due to the ease of application. However,



Fig. 6. Influence of curcumin nanoformulations on its bioaviability. Bulky particles of curcumin powder are poorly soluble and thus the bioabsorption/bioavability curcumin is severely limited. On the other hand, nanoformulated curcumin can be easily dissolved and effectively absorbed in the gut.

commercially available formulations of curcumin are intended only for the oral administration. be completely excluded.

It follows from the foregoing that even efficacy of curcumin formulation is likely to be limited. A possible solution could be its application with other nature compounds such as piperidine (as constituent of black pepper, which is part of golden milk) or polyphenols. For example, the combination of curcumin with two compounds, resveratrol and epicatechin gallate (food-derived natural polyphenols; 4:1:12.5, molar ratio) exhibited microsomal values of  $IC_{50}$  against head and neck cancer cell lines and potent anticancer effects in mice model of head and neck cancer [249].

Possible limitations of curcumin, especially in the long-term applications, may also be curcumin toxicity or loss of therapy sensitivity. Although curcumin does not show significant selectivity for tumour tissue, it has potent cytoselectivity for tumour cells [197]. In this case, normal cells/tissues display protective effects (decrease of oxidative stress, or protection of tissues from hypoxia) [257,260]. Nevertheless, Storka et al. reported possible toxic effect of curcumin in lipidic formulations against red blood cells (in vitro) [187]. However, no serious problem was not observed in a clinical study (healthy subjects). On the other hand, some patients with locally advanced or metastatic cancer have shown hematological adverse effects after intravenous administration of lyposomal curcumin (300 mg/m<sup>2</sup> over 6 h) [261].

Curcumin is a direct inhibitor of NF- $\kappa$ B signalling and decreases the activity and expression of P-glycoprotein (drug efflux pump) [262]. Curcumin is therefore well known as a promising substance for suppressing of multidrug resistance. However, in this case of a cholangiocarcinoma model (golden hamster exposed by nitrosamine), loss of inhibition activity against NF- $\kappa$ B was observed (six-month application) [142]. This suggests that although the risk of resistance is low, it cannot

## 6. Future direction

Numerous studies strongly imply, that curcumin is a promising antimetastatic agents. High impact studies and clinical trials suggest that curcumin application could enhance clinically used therapies, for example by reducing side effects of the main therapy. Curcumin formulations, especially in combinations with other polyphenols, piperine, represent promising agents for reducing the risk of cancer and its recurrence. Nevertheless, its therapeutic effectivity is severely limited by its bioactivity and biostability.

Nanoparticles are a well-studied system for drug transport, especially for anticancer agents with a high therapeutic impact [263–267]. In the case of curcumin, nanoformulations can significantly increase its stability and physiological concentration and thereby its anticancer effect [158]. However, nanoparticle accumulation in the tumour tissue can significantly improve their modification by tumour selective ligands such as RR-11a (synthetic enzyme inhibitor of Legumain, an asparaginyl endopeptidase) [268]. To increase therapeutic effectivity, nanoparticles together with curcuminoids can transport other anticancer agents such as anticancer peptides [115], photothermal dyes [134], cytostatic drugs (e.g., paclitaxel) [83]. Examples of studied curcumin nanoformulations and other antimetastatic systems are shown in Table 6.

Newly developed curcumin nanoparticle systems displayed high potential in the improvement of curcumin therapeutic effectivity. For example, curcumin formulated with double hydroxide nanoparticles displayed strong antimetastatic activity against A172 cells. [125] The nanoparticles represented submicromolar concentrations of curcumin (0.27  $\mu$ M) and decreased cell migration and invasiveness by

approximately one-third, while the effects of unformulated curcumin were negligible. In combination with used anticancer groups, the nanoparticles can also significantly boost curcumin's anticancer effectivity. Application of nanofibrous microspheres containing curcumin and docetaxel (8 g/kg; 1:1) increase in mean overall survival (OS; 48 days vs 18 days) compared to the control in the mice model with colon cancer [288]. In the case of free drugs, formulated and free docetaxel OS was 42, 39 and 29 days. A similar trend was observed in the reduction of abdominal metastases. A combination of the bortezomib and curcumin (1 and 1.337 mg/kg, respectively) transported by alendronate coated with poly-lactic-co-glycolic acid nanoparticles displayed strong therapeutic potential in the model of breast and bone metastases (mice with bone-implanted MDA-MB-231) [136]. Nanoparticles application sometimes decreases macrophage infiltration in bone metastases. Beside classically used drugs, curcumin formulation could be used for the enhancement of the novel therapeutic modalities such as photothermal therapy. Microemulsifying nanoformulations of curcumin and IR780 (photothermal dyes) decrease the number of metastatic nodules in the lung compared to single agents by more than two orders of magnitude compared to the control [134].

In clinical practice, other routes of administration, such as intravenous, pulmonary, and especially intratumoural administration, can significantly enhance the efficacy of curcumin. The question is whether improving the concentration of curcumin in the tumour could give any therapeutic quality. Beside significantly reducing tumour mass, and cell migration, another promising effect of curcumin could also be that it could target tumour microflora. It is well known that Helicobacter pylori is strongly associated with the induction and development of gastric tumours, which includes metastatic activity [291,292]. It was reported that curcumin displayed potent activity against various strains of Helicobacter pylori (MIC  $\sim$  5–50 µg/mL) [293,294].

In the case of other tumour types, it was also observed, that its specific microflora supports their development [295-298]. Interestingly, metastases have a different proportion of microflora compared to the primary tumour. Curcumin and especially its nanoformulations are effective against certain pathogenic microorganisms such as Staphylococcus aureus, Escherichia coli and Mycoplasma, which have been identified as representatives of the tumour microflora (Table 7) [299].

However, although we consider these readings to be transferable. they are usually too high (approximately hundreds of micromoles) with exceptions for oral and possibly intravenous administration. In comparison, the cytotoxicity of curcumin against tumour cells (represent by IC<sub>50</sub>) is orders of magnitude higher. It could be strongly suggested that curcumin doses, which can effectively target intratumoural microbiota, but most probably tumour cells will die first.

There are three points against these claims. Firstly, the value of curcumin IC<sub>50</sub> for the tumour microflora such as S. mutans (10.2  $\mu$ M) could be more comparable with IC50 than minimal inhibition concentration (MIC; 175 µg/mL) [319]. Besides, in the case of A. baumannii (ATCC 19606) curcumin combination with other agents such as EGCG display significantly antimicrobial effectivity (MIC =  $4 \mu g/mL$ ) [324]. It could be implied, that the therapeutic effect of above discussed Tricurin (curcumin formulation with resveratrol and epicatechin gallate) could also be associated with the reduction of tumour microbiota.

Secondly, significantly lower curcumin concentrations than MIC can still display potent effects against tumour microbiota (e.g., reduction biofilm formation, protease secretion) [303,329].

Thirdly, the above discussed MIC values were mostly obtained by using normal unformulated curcumin. At such concentrations, the curcumin used can be expected to be highly aggregated and its therapeutic ability could be severely reduced. In line with this hypothesis, the MIC value of curcumin quantum dots against P. gingivalis (ATCC 33277)was 1.1 µM [329]. This suggests that curcumin may act against the intratumoural microflora approximately as effectively as against cancer cells, and its antimicrobial toxicity is an integral part of the anticancer effect.

The above clearly shows the therapeutic potential of curcumin in

#### Table 6

Cι

incumin anticancer systems with the o	emphasis on antimetastatic activity.
Model and experimental conditions (route and dose)	Curcumin effect Assay (value of tested effect in the curcumin presence/ original value of tested effect <sup>a</sup> ; dose / time
Solid core system	
Double hydroxide (d = 119 nm)[125] A172[125]	Cell (
	↓PI3K,
	AKT and
Gold nanoparticles (d= 101/152; water/s MDA-MB 231 an 4T1	$ \begin{tabular}{lllllllllllllllllllllllllllllllllll$
	↑E.Cadherin
	CTATO
	VEGF), Proliferation (2/3 and ½; 10 μg/
Mice with 4T1; IP, 50 mg/kg/d for 21	mL/24 n), <b>Sraten</b> (1/2 and ½; 10 µg/mL/ 36 h), and <b>Migration</b> (1/5 and 1/3; 10 mg/mL/24 h) <b>Tumour</b> (
aayo	↓STAT3,
	MMP2, 3 and 9,
	↓cyclin D,
	VEGF),
Gold nanoparticles (d= 128/166 nm; wate curcumin and paclitaxel; 1:1, mass ratio MDA-MB 231 and 4T1	tumour volume and zero neoplasticity in both mammary and hepatic tissues er/serum containing medium) modified by [83] Cell (
	∱E-Cadherin,
	↓STAT3,
Mice with 4T1; IP 25 and 25 mg	VEGF), <b>Proliferation</b> (½ and ½; 10/ 24 h), <b>Sratch</b> (1/3 and 1/3/24; 5 mg/mL/ 24 h), and <b>Migration</b> (1/10 and 1/6; 5 mg/mL/24 h) <b>Tumour</b> (Paclitaxel induced
(curcumin and paclitaxel)/kg /d for 21 days	↓STAT3,
	$\downarrow$ MMP 2 and 3,
	↓cyclin D,
	VEGF), tumour volume and zero neoplasticity in mammary and hepatic tissues
Graphene Oxide (d =123 nm)[269] A549 and MDA-MB-231	Cell (
	↑ROS and
Graphene Oxide (d = 139 nm) curcumin a	$\Delta \Psi m$ ) and <b>Proliferation</b> (IC <sub>50</sub> ; 17.8 and 49 µg/mL/48 h) nd paclitaxel[269]

(continued on next page)

Model and experimental conditions	Curcumin effect	Model and experimental conditions	Curcumin effect
(route and dose)	Assay (value of tested effect in the curcumin presence/ original value of tested effect <sup>a</sup> ; dose / time	(route and dose)	Assay (value of tested effect in the curcumin presence/ original value of tested effect <sup>a</sup> ; dose / time
A549 and MDA-MB-231	Cell (	Lipid nanoparticles Lipid nanoparticles (d=20 nm) modified	with PEG and $\alpha\text{-NTP}$ (synthetic inhibitor of
	ROS and	5–8 F cells	<b>Proliferation</b> (41%; 50 $\mu$ M/24 h)
Superparamagnetic iron oxide nanoparti	$\downarrow$ ΔΨm) icles modified with β-cyclodextrin and	Mice with 5–8 F; IV, 125 nmol 13th, 15th, 17th, 19th, and 21th day	tumour volume (sometimes), delayed
Panc-1 and HPAF-II	Panc-1 (CXCL12 activated (		tos
	CXCR4, NF-κB	Lipid nanoparticles (d=20 nm) modified Hydrazinocurcumin (curcumin synthe	l with PEG and $\alpha$ -NTP-(d=150 nm[273]); tic analogue)[268]
	$\downarrow$ , SHH) gemcitabine activated $\alpha$ -SMA Cell (	Coculture 4T1 with RAW264.7	<b>4T1 Apoptosis</b> (0; RAW264.7/12 h; Pretreatment 18 $\mu$ M/12 h), <b>Macrophage</b> (IL-10 <sup>high</sup> , IL-12 <sup>low</sup> and TGF- $\beta$ <sup>high</sup> ;
	RRM1 and		STAT3
	RRM2,		MMP9,
	↑DCK,		MMP2,
	↑hCNT		VEGF and <b>0</b> CD206) <b>4T1</b> (
	miR-21 and		STAT3
Mice with HPAF-II; IP, Curcumin (100 μg) and geneticabine (300 μg)	miR-200a), <b>Proliferation</b> (>1/10; 15 +0.2 µM/48 h)		↓ MMP9,
	secondary tumourspheres.		MMP2 and
	cell migration)		VEGF), <b>4T1 Migration</b> (1/3;
	Tumour (		RAW264.7/12 h; Pretreatment 18 $\mu$ M/ 12 h), and <b>4T1 Invasion</b> (1/4; RAW264.7/12 h; Pretreatment 18 $\mu$ M/
twice a week, / weeks	Sino, E-cadherin,	Mice with 4T1 and E-RAW264.7; IV,	12 h) Metastase (
	↑Suga	(1 mM) 10th, 13th, 16th, 19th, 22th, 25th day	↓STAT3,
	Gli-1 and 2		Ki-67
	α-Sma SHH)		↓.
	Tumour mass.		CD31-positive microvessel density)
	felasticity tumour tissue		Tumour volume and
	Intestine, Liver, Lungs and Brain	Phospholinid nanonarticles (38 nm) con	↑OS taing curcumin and IB780 (photothermal
Zeolitic imidazolate framework-8 nanop	metastasis articles modified with hyaluronic acid (d $=$	dyes)[134] 4T1	Cell (
184/217; solid/hydrodynamic)[2/1] 4T1 tumour-bearing BALB/c mice, 25 mg/tg (whole papapartialse) /2	Tumour mass and		^ROS), <b>Proliferation</b> (~50%;9–12 μg/
days for 15 days	lung metastasis		mL curcumin/12 h), Invasion (49%; 6 and 3 $\mu g/mL$ (curcumin and IR780)/24 h
<pre>Pegylated nanodiamant (d = 19/95/128 irinotecan)[272] in LSL-KrasG12D/+ ;</pre>	nm; water/DMEM/RPMI); (curcumin and		12 h pretreatment), <b>Migration</b> (12.1%; <i>θ</i> and 3 μg/mL (curcumin and IR780)/24 h 12 h pretreatment)
Trp53loxP/loxP mice with Ca5Cre adenovirus; IP,15 and/ or 5 mg/kg	Ki67, IL-10 and TAM M2 markers (	4T1 under photothermal therapy	<b>Proliferation</b> (~50%; /6–9 μg/mL (curcumin and IR780; 2:1, mass ratio)
(curcumin and/ or irinotecan)	Ym1 and		/12 h), Invasion (6.1%; 6 and 3 $\mu g/mL$ (curcumin and IR780)/24 h; 12 h
	Ly6G)), and <b>serum</b> (		pretreatment), and <b>Migration</b> (12.1%; 6 and 3 μg/mL (curcumin and IR780)/24 h
	↓IL-9,	Mice with 4T1 under photothermal	12 h pretreatment) ↓Tumour volume,
	↓IL-10,	(curcumin and IR780)/ kg /2 days for 16 days	pulmonary metastatic nodules (more
		-	uian an order of magnitude)

adle 6 (continued)		Table 6 (continued)		
Model and experimental conditions (route and dose)	Curcumin effect Assay (value of tested effect in the curcumin presence/ original value of tested effect <sup>a</sup> ; dose / time	Model and experimental conditions (route and dose)	<b>Curcumin effect</b> Assay (value of tested effect in the curcumin presence/ original value of tested effect <sup>a</sup> ; dose / time	
yposome decored TN (fusion peptide of essential modulator -binding domain p acid (d = 127 nm); curcumin and cele 4T1, RAW264.7 and HUVEC	cell-penetrating peptide and cell NF-κB eptide; inhibitor of NF-κB) and hyaluronic coxib, ~1:1 molar ratio[274] Cell (	Pegylated liposomes (d=190 nm) modified	↓lung metastasis l with RGDK-lipopeptide (α5β1 integrin	
,	NTED	HUVEC and B16F10	Cell (	
	↓NF-KB,		TGFβ1,	
	IL-6 and		TGfBR2.	
	UTNF-α), Proliferation (IC <sub>50</sub> ; 12.3, 9.2 and 270 $\mu$ M/24 h), Macrophage migration (<5; 20 $\mu$ M/6 h; 4T1		↓Smad 2,	
	induced), and <b>Mamosphera volume</b> (50%: 20 µM/9 d)		Smad 3,	
ice with 4T1; IV, 20 mg (curcumin	Tumour (		Smad 4 and	
18th, 21th day	∱ІКВ-α,		<sup>↑</sup> Smad 7), <b>Proliferation</b> (~40% and	
	↓TNF-α,		40%; 20/24), <b>Migration</b> (1/2 and ½; 12 μM /24 h) and <b>Invasion</b> (1/2 and (1/2 and (1/2)))	
	<b>↓</b> IL-6),	C57BL/6 J mice with B16F10[278]; IV, 12 mg 14th 16th 19th 21th 24th	$\downarrow$ OS,	
	↑OS,	day	tumour volume (more than an order of	
	↓tumour volume,		magnitude),	
posome decored TN (fusion peptide of essential modulator -binding domain p	Uung and bone metastasis cell-penetrating peptide and cell NF-κB eptide; inhibitor of NF-κB) and hyaluronic	Pegylated liposomes (d = 208 nm) modifie receptor-targeting); curcumin and doxor B16E10 and HUVEC	Microvessel densities around tumours ed with RGDK-lipopeptide (α5β1 integrin ubicin; 6:1, molar ratio[278]	
acid (d = 127 nm); curcumin and celec lyposomes (d = 110 nm)[275] Mice with 471; IV, 21 (curcumin and celecoxib; 1:1, mass ratio) and 2.1 mg (doxorubicin)/ ke 9th 15th 21th	coxib, ~1:1 molar ratio and doxorubicin		↓TGFβ1,	
	MDSC infiltration in the lung tissue		↓TGfBR2,	
27th day posomes modified with glycyrrhetinic	acid and galactose ( $d = 139 \text{ nm}$ ); capsaicin		Smad 2,	
and curcumin[276] ppG2 and HepG2/LX-2 cells	Cell (		↓Smad 3,	
	↓P-gp,		Smad 4 and	
	Vimentin		<sup>^</sup> Smad 7), <b>Proliferation</b> (~30% and 30%: 1 µM (10 h) <b>Migration</b> (1/5 and 1	
	$\ensuremath{\left< E\text{-Cadherin} \right)}$ , Proliferation (IC $_{50;}$ 7.25 $\mu\text{M}$ and 11.36 $\mu\text{M}$ /24 h), and		10; 7 $\mu$ M /24 h; 8 h pretreatment), Invasion (1/4 and 1/6; 7 $\mu$ M /24 h; 8 h pretreatment)	
	Cell migration		↓OS,	
tee with H22 and H22/HSC[276]; IV, 5 mg/kg/2 days for 2 weeks	HSC (		tumour volume (half against double dose	
	↓α-SMA,		single agents),	
	↓CD31),	Lyposomes (d=168 nm) Zn-Curcumin[201	↓Microvessel densities around tumours ]	
	Tumour volume,	4T1	<b>Proliferation</b> (IC <sub>50</sub> ; 5.5 $\mu$ g /48 h),	
posomes modified with glycyrrhetinic	unumber of pulmonary nodules acid and galactose (d = 154; curcumin and $\frac{1}{2}$	Mice with 4T1; IV, 20 mg/kg (Zn-	↓Cell invasion and migration <b>Tumour</b> (apoptotic,	
combretastatin A-4 phosphate (CA4P) EL7402 and BEL7402/HUVEC)	277] BEL7402 (		<b>↓</b> Ki67,	
	↓VEGF)		↓CD31),	
	Cell adhesivity and		↑os	
	Cell migration		tumour volume,	
ALB/c mice with H22; IV 5 mg/kg	VEGF,		number of lung metastatic foci	
CA4P)/ 2 days for 2 weeks	VEGFR2 and	Lyposomes (d=168 nm) Cu-curcumin[201	]	
			(continued on next page	

Model and experimental conditions	Curcumin effect	Model and experimental conditions	Curcumin effect
(route and dose)	Assay (value of tested effect in the curcumin presence/ original value of tested effect <sup>a</sup> ; dose / time	(route and dose)	Assay (value of tested effect in the curcumin presence/ original value of tested effect <sup>a</sup> ; dose / time
4T1	<b>Proliferation</b> (IC <sub>50</sub> ; 5.6 µg/48 h),	Poly(2-ethyl-2-oxazoline)-poly(D,L-lactide	e) conjugate with curcumin and doxorubicin
Mice with 4T1; IV, 20 mg/kg (Cu- Curcumin). 2th. 5th. 8th. 11th day	Cell invasion and migration <b>Tumour</b> (apoptotic,	(d=104 nm); 5:1, mass ratio[281] MDA-MB-231	<b>Proliferation</b> (IC <sub>50</sub> ; 0.66 and 0.13 μg/mL (curcumin and doxorubicin) /48 h), <b>Cell</b> <b>adhesion assay</b> (60%; 0.21 and 0.04
	↓Ki67,		13 µg/mL (curcumin and doxorubicin)/ 48 h), Sratch (52%; 0.86 and 0.17 µg/mL
	↓CD31),		(curcumin and doxorubicin)/ 48 h), <b>Transendothelial migration</b> (100%; 0.86 and 0.17 µg/mL (curcumin and
	↑os		doxorubicin) /48 h) and <b>Invasion</b> (60%; 0.86 and 0.1713 ug/mL (curcumin and
	tumour volume,	BALB/c nude mice with MDA-MB-231:	doxorubicin)/ 48 h), Lung tissue (0MMP-9, 0E-Cadherin).
Lipid nanoparticle with polymeric core (p	unumber of lung metastatic foci,	Intravenously 4 and 20 mg/g doxorubicin and curumin/2, 14	reduction of lung metastasis.
PEG (d= 171/177 nm, water/ PBS)[60]		Polymeric nanoparticle (50 nm)[282]	¥ · · · · · · · · · · · · · · · · · · ·
MDA-MB-231	Proliferation (IC <sub>50</sub> ; 30 $\mu$ M /72 h),	BXPC-3, MiaPaCa and PBMC	Cancer cells (
	HUVEC (		↓NF-κB) PBMC (
	ICAM-1) and <b>Vascular adhesion</b> (30%;		LPS induced IL-6, IL-8 and TNF- $\alpha$
Oil nanoemulsion (d $\approx 200 \text{ nm}$ )[128]	10 µm / 24 II, to TNF-0 activated HOVEC)		production)
B16F10 cells	Proliferation (IC_{50}; 46 $\mu\text{M}/\text{24}$ h), Cell (	Huh7	Cumin (NFC) and free drug sorareni0[283] Cell (
	ROS), Migration (1/2; 12.5 μM/24 h) and Invasion (1/5; 12.5 μM/24 h)		↓ ERK 1/2,
C57BL/6 mice with B16F10, were administered after excising the	Tumour recurrence and metastasis		<b>↓</b> MMP-9,
tumour and before suturing the wound 1500 µM; 14 days	(more than an order of magnitude)		p65 (nukleus),
Micelles (d = 53.5 nm),[279] A549	Cell (		CD133 and
	VM channels		<sup>TIMP-1</sup> ), <b>Proliferation</b> (IC <sub>50</sub> ; 35 $\mu$ M/ 48 h), <b>Invasion</b> (1/4; 40 $\mu$ M/24 h) and
	MMP-2 and		Invasion (1/5; 40 $\mu M$ /24 h; in the combination of 10 $\mu M$ sorafenib)
	HIF-1α), <b>Proliferation</b> (89.9%; 62.5/24)	MHCCLM3	Cell (
	and <b>Cell adhesion assessments</b> (67;8/12)		ERK1/2,
Nanomicellar-curcumin (10 nm, Exir Nano The B16F10 cell line	o Sina Company (Tehran, Iran)[280] Proliferation (1/8; 20/24)		<b>↓</b> MMP-9,
C57BL/6 mice with B16; IP, 20 mg/kg, 4X/week, 3 weeks	↑os,		p65 (nukleus),
	ung metastasis, <b>lung</b> (		CD133 and
	↓Treg and		TIMP-1), <b>Proliferation</b> (IC <sub>50</sub> ; 5 μM
	$\uparrow$ activated T cells + serum (		/48 h), <b>Invasion</b> (6/10; 40 µM/24 h), and <b>Invasion</b> (1/3; 40 µM/24 h; in the
	↑CXCL10 and	Mice with MHCCLM3;NFC (1.56 g/kg/	combination of 10 µM sorafenib) ↓CD133,
Polymeric system	ÎFN-γ)	kg/ daily, oral)	tumour mass,
Monomethyl poly(ethylene glycol)-poly(& micelles (28.2)[135]	-caprolactone) copolymer (MPEG-PCL)		Pulmonary metastasis
4T1	Proliferation (~50; 20 µg/48 h)	Alexandrometer	synergy
BALB/c mice with 4T1, IV, 30 mg/kg /day for 10 days	↑OS (42 vs 28 days),	Alendronate coated poly-lactic-co-glycolic [136]	acid 235 nm curcumin and bortezomib
	tumour volume and mass (1/4)	231; IV,1 and 1.337 mg (bortezomib	<b>Bone</b> : (Sometimes reduction in macrophage infiltration of tumour mass)
	$\uparrow$ apoptosis index (4x) and	PLGA-curcumin NPs (188 nm)[284]	<b>Proliferation</b> (73%: 2.5 µg/mJ /24)
	↓angiogenesis (microvehicles per field 1/ 3),	107-701	<b>Scratch</b> (80%; 2.5 µg/mL /24), <b>Invasion</b> (80% and 56%; 7.5 µg/mL/24 h and 48 h)
	unumber and mass of lung metastases (1/4)	NanoCurc[204]	(continued on next page)

#### Table 6 (continued)

Model and experimental conditions (route and dose)	Curcumin effect Assay (value of tested effect in the curcumin presence/ original value of tested effect <sup>a</sup> ; dose / time	
Mice with Pa03C; IP, NanoCurc	Tumour mass,	
(25 mg/kg) and or gemcitabine (20 mg/kg) twice, daily, 2weaks	Micrometastasis in lungs, lymph nodes, and peritoneum -strong synergic effect tumour (	
	↓NF-KB,	
	cyclin D1 and	
Dextran Nanobubles (dextran sulfate-shell	$\downarrow$ MMP-9) and perfluoropentane core, d = 348 nm)	
[285] PC-3 and DU-145	<b>Proliferation</b> (86% and 91%; 5 $\mu$ M/24 h) <b>adhesion</b> (1/3 and ½; 5 $\mu$ M /18 h; TNF- $\alpha$ activated, 30 min pretreatment) and migration (1/3 and 1/3; 5 $\mu$ M /18 h; TNF- $\alpha$ activated and normal)	
B16F10	articles 189 nm[129]	
Mice with B16F10; orally 3 and 6 mg/	Lung metastases (formation tumour	
BSA nanoparticles 130 nm, curcumin and	doxorubicin[286]	
Mice with B16F10	Lung metastases (weight 1/4)	
Hyaluronic acid (HA)-functionalized regenerated silk fibroin-based nanoparticles		
4T1	<b>Proliferation</b> (IC <sub>50</sub> ; 37 and 1.2 μM /24 and 48 h) and <b>Scrath</b> (37% (total drug); 16 mg/24)	
BALB/c nude mice with 4T1; IV, 5 mg/	Tumour mass,	
kg total day 27 day	$\uparrow$ OS, tumour tissue (	
	Ki67), apoptosis cell,	
Nanofibrous microspheres (PLA-PEO-PPO	↓numbers of lung metastasis nodules PEO-PLA); doxorubicin and curcumin, 1:1	
CT26 mice, IP; 8 mg/kg	Tumour tissue (	
	<b>↓</b> Ki-67),	
	↑os,	
	abdominal metastases (1/4),	
Cell and cell derived systems	↓microvesell density (1/5)	
B16F10 derived nanovesicle[289] B16F10/spleen T cells	$\uparrow$ CD8 + and	
	CD4 + T cells	
Biotin chitosan naporticles (hydrodynamic cells and streptavidin[290]	size = 377 nm), biotin mesenchymal stem	
Mice with mice melanoma B16F10; IV	lung metastases	

α-SMA, Alpha smooth muscle actin; ΔΨm, Mitochondrial membrane potential; Akt, Protein kinase B; CXCL10, C-X-C motif chemokine ligand 10; CXCL12 C-X-C motif chemokine ligand 12; CXCR-4, C-X-C chemokine receptor type 4; DCK, Deoxycytidine kinase, PI3K, Fosfatidylinositol-3-kináza; hCNTs, Homology Modeling of Human Concentrative Nucleoside Transporters; HIF-1α, Hypoxiainducible factor 1-alpha; GLI1, GLI Family Zinc Finger 1; GLI1, GLI2 Family Zinc Finger 2; ICAM-1, Intercellular Adhesion Molecule 1; IkB-α, Nuclear factor of kappa light polypeptide gene enhancer in B-cells inhibitor, alpha; IL-6, Interleukin 6I; L-9, Interleukin 9; IL-10, Interleukin 10;; IL-12, Interleukin 12; INF-γ, Interferon gama; mTOR, Ly6G, Lymphocyte antigen 6 complex locus G6D; Mammalian target of rapamycin; MMP-2, Matrix metalloproteinase-2; MMP3, Matrix metalloproteinase-3; MMP-9, Matrix metalloproteinase-9; NF-κB, Nuclear factor kappa-light-chain-enhancer of activated B cells; Pgp, P-glycoprotein 1; ROS, Reactive oxygen species; RRM1, Ribonucleoside-diphosphate reductase large subunit; RRM2, Ribonucleotide reductase small subunit; SHH, Sonic Hedgehog; SMO, Smoothened, Seven Transmembrane Spanning Receptor; Sufu, Suppressor of fused homolog; STAT3, Signal transducer and activator of transcription 3; TGF- $\beta$ , Transforming growth factor  $\beta$ ; TGF- $\beta$ 1, Transforming growth factor  $\beta$ 1; TGfBR2, Transforming growth factor beta receptor II; TIMP1, TIMP Metallopeptidase Inhibitor 1; TNF- $\alpha$  Tumour necrosis factor alpha; VEGF, Vascular endothelial growth factor; VEGFR-2, Vascular endothelial growth factor receptor 2; vascular mimicry (VM); Ym1, known as chitinase-like protein 3;

5–8 F, Human nasopharyngeal cancer; A172; Human glioblastoma A549, Human lung adenocarcinoma; B16, Murine melanoma cell line; B16F10, Highly metastatic murine melanoma cell line; BEL7402, Human hepatocellular carcinoma; BXPC-3, Human pancreatic adenocarcinoma; CT26, Mouse colon adenocarcinoma; DU-145, Human prostatic carcinoma; H22, Mouse hepatocellular carcinoma; HPAF-II, Human pancreatic adenocarcinoma; HepG2, Human liver cancer cell line; HSC, Mouse hepatic stellate cell; Huh7, Human hepatoma; HUVEC, Human umbilical vein endothelial cells; LX-2, Human hepatostellate cell; MDA-MB 231, Triple negative human breast adenocarcinoma; MiaPaCa, Human pancreatic ductal adenocarcinoma; MHCCLM3, Human hepatocellular carcinoma; PBMC, Peripheral blood mononuclear cells; Pa03C, Human pancreatic cancer; RAW264.7, Mice macrophage cell line; 4T1, Mice mammary tumour ↑ = curcumin activation/induction/enhance; ↓ = curcumin suppression/inhi-

| = curcular activation induction character,  $\downarrow =$  curcular suppression in bition;  $\mathbf{0} =$  without change.

<sup>a</sup> If a fraction was stated, the value was not explicitly stated in the article, but was subtracted from the presented graph.

anticancer therapy, especially as an antimetastatic. Nevertheless, it cannot be ruled out that in some cases it may not be sufficient. In medical research, curcumin analogues and synthetic derivatives are intensively studied in anti-cancer therapy. Possible synthetic approaches to the modification of curcumin chemical structures are shown in the Fig. 7.

The most used strategy is the elimination of  $\beta$ -diketone moiety from the curcuminoid structure motif. The enolic proton of the curcumin also plays a critical role in its aggregation [333]. Breaking of intermolecular hydrogen bonds between the formed layers primarily through the acidic enolic proton exchange can lead to significantly increase curcumin solubility [333]. On the other hand,  $\alpha$ , $\beta$ -unsaturated diketone moiety can represent a Michael reaction acceptor, which belongs to the major class of phase-II enzyme inducers [334]. It can be possible, that this structure motif may be responsible for inducing HO-1 and STAT3 inhibition [215,335]. In addition, some studies suggest that synthetic curcuminoids with a  $\beta$ -diketone moiety are promising anticancer agents [336]. Examples of curcumin derivatives are shown on Fig. 8.

For example, dimethylcurcumin (ASC-J9) induces degradation of the androgen receptors, including AR-V7 and AR-F876L, which are strongly associated with lacking sensitivity to therapy and metastasis during pancreatic cancer [337]. Its application sometimes decreases invasiveness of pancreatic cancer cells (Du145 and C4–2 cells), whereas this effect was significantly higher for AR-F876L positive cells [338]. Similarly, non-toxic doses of curcumin reduced AR expression and activity in LNCaP Similarly, non-toxic doses of curcumin reduced AR expression and activity in LNCaP Similarly, non-toxic doses of curcumin reduced AR expression and activity in LNCaP Similarly, non-toxic doses of curcumin reduced AR expression and activity in LNCaP (prostate cancer cells) [339].

Various polar groups such as fluorine have also been studied to improve the solubility and biological efficacy of curcuminoids. For example, EF24 (1  $\mu$ M) completely suppressed the migration of melanoma cells (A375 and Lu1205) [340]. Similarly, CDF (0.125  $\mu$ M) displayed strong cytotoxicity (~ 50% viability), the decrease in CSC self-renewal capacity and potent antimetastatic effects (decrease in the cell invasiveness, migration, and angiogenesis) against pancreatic cancer cells (AsPC-1 and MiaPaCa-2) in hypoxia conditions [341]. This effect was associated with the decrease in IL-6 and VEGF protein levels.

An interesting strategy is the combination of curcumin and piperidine structure motif [342-345]. For example, Das et al. reported, that IC<sub>50</sub> of 3,5-bis(arylidene)- 4-piperidone dimers for colon cancer cell

#### Table 7

Curcumin effect on the intratumoural microbiota.

Microflora	Cancer type	Variant	Toxicity
Pseudomonas	Breast	P. aeruginosa	50–2500 μg/mL (MIC)[197.301–306]
Porphyromonas	Breast, Head	P. gingivalis (ATCC	62.5 μg/mL (MIC)
[300,307]	and Neck	33277)	[308]
Proteus[300]	Breast	P. mirabilis	192–300 μg/mL
			(MIC)[197,301]
Enterotoxigenic	Colorecteral	B. fragilis	> 128, 32, 64 and
Bacterolaes			128 µg/mL (MIC)
[309]			DMC tetrabudrovy
			respectively[310]
Bacteroides[311]	Colorectal	B. dorei	> 128, 128, 128 and >
			128 μg/mL (MIC)
			curcumin BDMC and
			DMC tetrahydroxy,
			respectively[310]
Prevotella[311]	Colorectal	P. intermedia	10 μg/mL (MIC)[312]
Fusobacterium	Breast,	F. nucleatum	10 μg/mL (MIC)
[313-315]	Colorectal		(27 µM)[312]
Esterichia[316]	Colorectal	E. coli	$192-1500 \ \mu g/mL$
			(IMIC)[197,301,302, 317]
Streptococcus	Lung	S. pyogenes	31.25 µg/mL (MIC)
[318]	8	er FJ -8	[197]
		S. mutans	128–175 µM (MIC)
			10.2 (IC50)[319,320]
Aggregatibacter	Pancreatic	A. actinomycetem	0.2 μg/mL (MIC)
[321]			(0.54 μM)[322]
Acinetobacter	Ovarian	A. lwoffii	250 μg/mL[197]
[323]		A. baumannii	> 5000 µg/mL (MIC)
			[197]
		A. baumannii (ATCC	> 256 µg/mL (MIC),
		19606; MDR)	Synergy EGCG[324]
Mycoplasma	Ovarian	M. hominis,	50 μg/mL (MIC)[326]
[325]		capricolum,	
		genitalium,	
		M mycoides subsp	100 ug/mL (MIC)
		capri	[326]
Klebsiella[325]	Ovarian	K. pneumonia	216–2000 µg/mL
		1	(MIC)[197,302,317]
Staphylococcus	Non-	S. aureus	187–600 mg/mL
[327]	melanoma		(MIC)
	skin cancer	S. aureus (MSSA)	219 μg/mL (MIC)
			[302]
		5. uureus (MKSA)	217 μg/mL (MIC) [302]
		S. epidermidis	100 μg/mL (MIC)
		4	[197]
		S. epidermidis	46.9 µg/mL (MIC)
		(ATCC 14990)	[328]

MIC = minimal inhibition concentration.

lines (HCT116 and HT29) can vary in the range  $0.01-0.1 \mu M$  [342]. In the case of 5-floururacil, these values were approximately hundred times higher. In addition to higher cytotoxicity in vitro, these compounds could exhibit higher bioactivity in vivo compared to curcumin.

PAC (Fig. 8 3,5-Bis (4-hydroxy-3-methoxybenzylidene)-N-methyl-4piperidone) intraperitoneal application (100 mg/kg) showed  $c_{max}$ 35 µg/mL (~320 µM) in the mice plasma, nevertheless in this curcumin application  $c_{max}$  was only 10.3 µg/mL [344]. Since PAC sometimes exhibited cytotoxicity against MDA-MB-231 cells (55% vs ~5%; 10 µM, 72 h), PAC could represent a suitable structural motif for the treatment of triple negative breast cancer compared to single curcumin [345]. PAC (10 µM, 24 h) also displayed strong antimetastatic effects against MDA-MB-231 (reduction of cell migration and invasiveness by less than 5%) [344]. In the mice model with the MDA-MB231 tumour, PAC (100 mg/kg) sometimes decreased the expression of NF- $\kappa$ B, c-Myc and Cyclin D1 in the tumour tissue [344,345].

In the case of xenografts of colon and pancreatic cancer cells, lower

levels of NF- $\kappa$ B, HIF-1 $\alpha$ , VEGF and TGF- $\beta$  proteins were sometimes observed after treatment with synthetic curcuminoids (EF31 and especially UBS109) compared to control and natural curcumin [346,347]. In accordance with UBS109 (15 mg/kg/weak, i.p.) strongly suppress metastatic activity MDA-MB-231 in the mice lung [348]. This effect was associated with high cytotoxic activity against MDA-MB231 (IC<sub>50</sub> =1.33  $\mu$ M) and plasma concentration (618 ng/mL). In the case of oral application (50 mg/kg), UBS109 suppression of breast cancer induced bone destruction and bone metastasis. This effect was probably caused by the strong increase in bone mineralization and the decrease of osteoclast formation [349].

Since curcumin and other polyphenols are strong chelators, their use with highly soluble metal ions (e.g.,  $Zn^{2+}$ ,  $Cu^{2+}$ ,  $Mg^{2+}$  and  $Se^{2+}$ ) leads to significant improvements of solubility and stability in aqueous systems [350]. Because labile  $\beta$ -diketone moiety of curcuminoids significantly participated in their instability [155], chelated metal ions can protect the curcumin moiety and improve its lifetime [350]. Metal-curcumin complexes display potent effectivity against various oncological diseases [351]. These biological properties are associated with the targeting of inflammatory and transcription factors, protein kinases, antiapoptotic proteins, and antioxidant enzymes. In addition, some of them (e.g.,  $Cu^{2+}$ , VO, La) display sometimes higher light toxicity than original curcumin, comparable with Photofrin. [352] In combination with structure motif of the curcumin and ruthenium complexes, this could lead to an increase in curcumin therapeutic effectivity [353-356]. In this case, IC<sub>50</sub> value for Ru(cym)(bdcurc)(PTA)]PF<sub>6</sub> (Fig. 8) against A2780 was 0.14 µM (24 h), value for cisplatin was 1.5 µM. [354] Srivastava et al., reported that derivatives of  $[Ru(NN)_2(cur)](PF_6)$  [NN = bpy (1), phen (2)] display strong activity (MIC =  $1 \mu g/mL$ ) against methicillin and vancomycin-resistant S. aureus strains (intratumoural microbiota) [357]. Some of them are potent inhibitors of NF-KB [339], which might suggest their usefulness in the treatment of metastatic and drug-resistant tumours.

As implied above, suitable curcumin designed synthetic analogues can sometimes display higher in vitro and in vivo therapeutic potential compared to natural curcumin. For example, UBS109 (Fig. 8) displays higher plasma  $c_{max}$  than IC<sub>50</sub> for MDA-BA231. Improving pharmaceutical properties of synthetic curcuminoids by finding the most suitable method of administration and formulation could lead to the preparation of highly effective antimetastatic agents. Nevertheless, numerous preclinical and clinical trials are requested for the final validation of this hypothesis and for the design of suitable therapeutic regiments.

### 7. Conclusion

Curcumin represents a promising structural motif for the independently multitargeting of various metastatic mechanisms. Nevertheless, its therapeutic abilities are strongly limited by its solubility and bioactivity. This review described various curcumin micro and nanoformulations used and studied for the improvement of therapeutic effectivity of curcumin. Also, the influence of various routes of application of curcumin on its therapeutic effectivity was considered. Finally, curcumin's possible effect on the tumour microbiota and some promising curcumin synthetic derivatives were also presented and discussed.

## Funding

This work was supported by the project of Charles University in Prague (SVV260521; UNCE 204064) and the Ministry of Education, Youth, and Sports of Czech Republic, grant no. LM2023053 (EATRIS-CZ). We were also supported by the Cooperation Program of Charles University, as well as the project of Charles University in Prague Progress Q26–38/LF1 and Prague Progress Q27/LF1. The research study was also supported by the Technology Agency of the Czech Republic within projects nos. TN01000013, TN02000109 and FW02020128. We also acknowledge the Operational Programme Research, Development, and



**Fig. 7.** Various approaches in the synthetic modification of curcumin more detailed described in the cited works [330–332], A, B - 2-methoxy-2-hydroxyphenyl moiety: substitution at hydroxy group, other substituents at phenyl core/heterocycle; C - ethylene moiety: reduced form (CH<sub>2</sub>CH<sub>2</sub>), omitting or extending by ethylene unit; D - substitution at C-4: substitution by alkyl, aryl or arylidene; E – diketo or keto-enol group: keto-enol tautomerism, heterocycle (especially pyrazole and isoxazole derivatives), Schiff bases and oximes, substitution for keto group (acetone or cyclic ketone derivatives).



Fig. 8. Examples of tested curcumin derivatives.

Education within the project Center for Tumor Ecology—Research of the Cancer Microenvironment Supporting Cancer Growth and Spread (reg. no. CZ.02.1.01/0.0/0.0/16\_019/0000785). This work was also supported by the Ministry of Health of Czech Republic, grant nos. RVO-VFN64165, NU21–08–00407 and NU22–08–00160. We also thank the project National Institute for Cancer Research (Programme EXCELES, ID project no. LX22NPO5102) funded by the European Union—Next Generation EU. This work was also supported by the National Institute for Neurological Research (Programme EXCELES, ID Project No. LX22NPO5107) funded by the European Union—Next Generation EU.

## CRediT authorship contribution statement

Petr Dytrych participate in writing and supervision of manuscript, he also integrates individual author contributions. Zdeněk Kejík participate in writing and supervision of manuscript, design of figures and table preparations. Jan Hajduch participate in writing and supervision of manuscript. Robert Kaplánek participate in supervision of manuscript and design of figures. Kateřina Veselá and Kateřina Kučnirová participate in writing manuscript and table preparation. Markéta Skaličková and Anna Venhauerová participate design of figures and preparing tables. David Hoskovec participate in writing and supervision of manuscript. Milan Jakubek and Pavel Martásek design concept of manuscript and participate in manuscript supervision.

## Conflict of interest statement

Petr Dytrych did not have any conflict of interest. Zdeněk Kejík did not have any conflict of interest. Jan Hajduch did not have any conflict of interest. Robert Kaplánek did not have any conflict of interest. Kateřina Veselá did not have any conflict of interest. Kateřina Kučnirová did not have any conflict of interest. Markéta Skaličková did not have any conflict of interest. Anna Venhauerová did not have any conflict of interest. David Hoskovec did not have any conflict of interest. Pavel Martásek did not have any conflict of interest. Milan Jakubek did not have any conflict of interest.

## Data availability

No data was used for the research described in the article.

#### References

- H. Sung, et al., Global Cancer Statistics 2020: GLOBOCAN estimates of incidence and mortality worldwide for 36 cancers in 185 countries, CA Cancer J. Clin. 71 (3) (2021) 209–249.
- [2] M.E. Menyailo, et al., Metastasis prevention: focus on metastatic circulating tumor cells, Mol. Diagn. Ther. 25 (5) (2021) 549–562.
- [3] A. Gandalovičová, et al., Migrastatics-anti-metastatic and anti-invasion drugs: promises and challenges, Trends Cancer 3 (6) (2017) 391–406.
- [4] D. Rosel, et al., Migrastatics: Redirecting R&D in Solid Cancer Towards Metastasis? Trends Cancer 5 (12) (2019) 755–756.
- [5] D.Y. Chung, J.S. Ha, K.S. Cho, Novel treatment strategy using second-generation androgen receptor inhibitors for non-metastatic castration-resistant prostate cancer, Biomedicines 9 (6) (2021).
- [6] J. Brábek, et al., Interleukin-6: Molecule in the Intersection of Cancer, Ageing and COVID-19, Int J. Mol. Sci. 21 (21) (2020).
- [7] J. He, et al., The new opportunities in medicinal chemistry of fourth-generation EGFR inhibitors to overcome C797S mutation, Eur. J. Med Chem. 210 (2021), 112995.
- [8] G. Sun, et al., Treatment of patients with cancer using PD-1/PD-L1 antibodies: Adverse effects and management strategies (Review), Int J. Oncol. 60 (6) (2022).
- [9] M. Rašková, et al., The Role of IL-6 in Cancer Cell Invasiveness and Metastasis-Overview and Therapeutic Opportunities, Cells 11 (22) (2022).
- [10] S. Gkountela, et al., Circulating Tumor Cell Clustering Shapes DNA Methylation to Enable Metastasis Seeding, Cell 176 (1–2) (2019) 98–112.e14.
- [11] E. Schuster, et al., Better together: circulating tumor cell clustering in metastatic cancer, Trends Cancer 7 (11) (2021) 1020–1032.
- [12] A. Ring, et al., Biology, vulnerabilities and clinical applications of circulating tumour cells, Nat. Rev. Cancer 23 (2) (2023) 95–111.
- [13] J. Petrik, et al., Circulating tumor cells in colorectal cancer: detection systems and clinical utility, Int J. Mol. Sci. 23 (21) (2022).
- [14] I. Fridrichova, L. Kalinkova, S. Ciernikova, Clinical relevancy of circulating tumor cells in breast cancer: epithelial or mesenchymal characteristics, single cells or clusters? Int J. Mol. Sci. 23 (20) (2022).
- [15] M. Raudenská, et al., Engine shutdown: migrastatic strategies and prevention of metastases, Trends Cancer (2023).
- [16] I. Dagogo-Jack, A.T. Shaw, Tumour heterogeneity and resistance to cancer therapies, Nat. Rev. Clin. Oncol. 15 (2) (2018) 81–94.
- [17] Y. Yamamoto, et al., The heterogeneity of cancer-associated fibroblast subpopulations: Their origins, biomarkers, and roles in the tumor microenvironment, Cancer Sci. 114 (1) (2023) 16–24.
- [18] M. Jakubek, et al., Strategy for improved therapeutic efficiency of curcumin in the treatment of gastric cancer, Biomed. Pharm. 118 (2019), 109278.
- [19] Z. Kejík, et al., Circulating Tumour Cells (CTCs) in NSCLC: From Prognosis to Therapy Design, Pharmaceutics 13 (11) (2021).
- [20] W. Fan, et al., Nanotechnology for multimodal synergistic cancer therapy, Chem. Rev. 117 (22) (2017) 13566–13638.
- [21] A. Aras, et al., Targeting cancer with nano-bullets: curcumin, EGCG, resveratrol and quercetin on flying carpets, Asian Pac. J. Cancer Prev. 15 (9) (2014) 3865–3871.
- [22] G.T. Bolger, et al., Relationship between the in vitro efficacy, pharmacokinetics and in vivo efficacy of curcumin, Biochem Pharm. 205 (2022), 115251.
- [23] B.E. Bachmeier, P.H. Killian, D. Melchart, The role of curcumin in prevention and management of metastatic disease, Int J. Mol. Sci. 19 (6) (2018).
- [24] Y. DENG, E. VERRON, R. ROHANIZADEH, Molecular mechanisms of antimetastatic activity of curcumin, Anticancer Res. 36 (11) (2016) 5639–5647.
- [25] U. Sharma, et al., Role of Hedgehog and Hippo signaling pathways in cancer: A special focus on non-coding RNAs, Pharmacol. Res. 186 (2022), 106523.
- [26] A.A. Hasan, V. Tatarskiy, E. Kalinina, Synthetic pathways and the therapeutic potential of quercetin and curcumin, Int. J. Mol. Sci. 23 (22) (2022) 14413.
   [27] M. Mohamadian, et al., Review on the therapeutic potential of curcumin and its
- [27] M. Mohamadian, et al., Review on the therapeutic potential of curcumin and its derivatives on glioma biology, Neurochem Res 47 (10) (2022) 2936–2953.
- [28] M. Chu, et al., Targeting cancer stem cells by nutraceuticals for cancer therapy, Semin Cancer Biol. 85 (2022) 234–245.
- [29] J.Y. Zou, et al., Curcumin converts Foxp3+ regulatory T cells to T helper 1 cells in patients with lung cancer, J. Cell Biochem 119 (2) (2018) 1420–1428.
- [30] B. Xu, L. Yu, L.Z. Zhao, Curcumin up regulates T helper 1 cells in patients with colon cancer, Am. J. Transl. Res 9 (4) (2017) 1866–1875.
- [31] D. Xue, et al., Role of regulatory T cells and CD8(+) T lymphocytes in the dissemination of circulating tumor cells in primary invasive breast cancer, Oncol. Lett. 16 (3) (2018) 3045–3053.
- [32] M. Mego, et al., Circulating Tumor Cells (CTC) Are Associated with Defects in Adaptive Immunity in Patients with Inflammatory Breast Cancer, J. Cancer 7 (9) (2016) 1095–1104.
- [33] K. Ried, P. Eng, A. Sali, Screening for Circulating Tumour Cells Allows Early Detection of Cancer and Monitoring of Treatment Effectiveness: An Observational Study, Asian Pac. J. Cancer Prev.: APJCP 18 (8) (2017) 2275–2285.
- [34] M.M. Khan, V.P. Torchilin, Recent Trends in Nanomedicine-Based Strategies to Overcome Multidrug Resistance in Tumors, Cancers (Basel) 14 (17) (2022).
- [35] T. Xu, et al., Application of curcumin and its derivatives in tumor multidrug resistance, Phytother. Res 34 (10) (2020) 2438–2458.
- [36] A. Alausa, et al., Checkpoints and immunity in cancers: Role of GNG12, Pharm. Res 180 (2022), 106242.
- [37] S. Mirzaei, et al., NF-κB as a regulator of cancer metastasis and therapy response: A focus on epithelial-mesenchymal transition, J. Cell Physiol. 237 (7) (2022) 2770–2795.

- [38] S. Mirzaei, et al., Regulation of Nuclear Factor-KappaB (NF-kB) signaling pathway by non-coding RNAs in cancer: Inhibiting or promoting carcinogenesis? Cancer Lett. 509 (2021) 63–80.
- [39] J. Piette, Signalling pathway activation by photodynamic therapy: NF-κB at the crossroad between oncology and immunology, Photochem. Photobio. Sci. 14 (8) (2015) 1510–1517.
- [40] M. Younes, et al., The Synergistic Effects of Curcumin and Chemotherapeutic Drugs in Inhibiting Metastatic, Invasive and Proliferative Pathways, Plants (Basel) 11 (16) (2022).
- [41] Y. Panahi, et al., Adjuvant therapy with bioavailability-boosted curcuminoids suppresses systemic inflammation and improves quality of life in patients with solid tumors: a randomized double-blind placebo-controlled trial, Phytother. Res 28 (10) (2014) 1461–1467.
- [42] A. Wells, C. Yates, C.R. Shepard, E-cadherin as an indicator of mesenchymal to epithelial reverting transitions during the metastatic seeding of disseminated carcinomas, Clin. Exp. Metastas-.- 25 (6) (2008) 621–628.
- [43] Q. Dong, et al., Pre-metastatic Niche Formation in Different Organs Induced by Tumor Extracellular Vesicles, Front. Cell Dev. Biol. 9 (2021).
- [44] M. Colombo, G. Raposo, C. Théry, Biogenesis, secretion, and intercellular interactions of exosomes and other extracellular vesicles, Annu Rev. Cell Dev. Biol. 30 (2014) 255–289.
- [45] S. Mukherjee, et al., Phytosomal curcumin causes natural killer cell-dependent repolarization of glioblastoma (GBM) tumor-associated microglia/macrophages and elimination of GBM and GBM stem cells, J. Exp. Clin. Cancer Res 37 (1) (2018) 168.
- [46] S. Mukherjee, et al., TriCurin, a synergistic formulation of curcumin, resveratrol, and epicatechin gallate, repolarizes tumor-associated macrophages and triggers an immune response to cause suppression of HPV+ tumors, Cancer Immunol. Immunother. 67 (5) (2018) 761–774.
- [47] Q. Wang, et al., Curcumin suppresses epithelial-to-mesenchymal transition and metastasis of pancreatic cancer cells by inhibiting cancer-associated fibroblasts, Am. J. Cancer Res 7 (1) (2017) 125–133.
- [48] G. Jin, et al., Combination curcumin and (-)-epigallocatechin-3-gallate inhibits colorectal carcinoma microenvironment-induced angiogenesis by JAK/STAT3/IL-8 pathway, Oncogenesis 6 (10) (2017), e384.
- [49] S. Bhattacharyya, et al., Curcumin reverses T cell-mediated adaptive immune dysfunctions in tumor-bearing hosts, Cell Mol. Immunol. 7 (4) (2010) 306–315.
- [50] S. Mirza, et al., Curcumin Targets Circulating Cancer Stem Cells by Inhibiting Self-Renewal Efficacy in Non-Small Cell Lung Carcinoma, Anticancer Agents Med Chem. 17 (6) (2017) 859–864.
- [51] A. Coker-Gurkan, et al., Proinflammatory cytokine profile is critical in autocrine GH-triggered curcumin resistance engulf by atiprimod correatment in MCF-7 and MDA-MB-231 breast cancer cells, Mol. Biol. Rep. 47 (11) (2020) 8797–8808.
- [52] D.E. Iglesias, et al., Curcumin Mitigates TNFα-Induced Caco-2 Cell Monolayer Permeabilization Through Modulation of NF-κB, ERK1/2, and JNK Pathways, Mol. Nutr. Food Res 66 (21) (2022), e2101033.
- [53] I.H. Ham, et al., Curcumin inhibits the cancer-associated fibroblast-derived chemoresistance of gastric cancer through the suppression of the JAK/STAT3 signaling pathway, Int J. Oncol. 61 (1) (2022).
- [54] G. Hamilton, et al., Small cell lung cancer: Recruitment of macrophages by circulating tumor cells, Oncoimmunology 5 (3) (2016), e1093277.
- [55] T. Liu, et al., Self-seeding circulating tumor cells promote the proliferation and metastasis of human osteosarcoma by upregulating interleukin-8, Cell Death Dis. 10 (8) (2019) 575.
- [56] J.P. Arnoletti, et al., Pancreatic Ductal Adenocarcinoma (PDAC) circulating tumor cells influence myeloid cell differentiation to support their survival and immunoresistance in portal vein circulation, PLoS One 17 (3) (2022), e0265725.
- [57] G. Theil, et al., Association of circulating tumor cells with inflammatory and biomarkers in the blood of patients with metastatic castration-resistant prostate cancer, Life (Basel) 11 (7) (2021).
- [58] A. König, et al., Determination of Interleukin-4, -5, -6, -8 and -13 in Serum of Patients with Breast Cancer Before Treatment and its Correlation to Circulating Tumor Cells, Anticancer Res 36 (6) (2016) 3123–3130.
- [59] A.E. Lohmann, et al., Association of metabolic, inflammatory, and tumor markers with circulating tumor cells in metastatic breast cancer, JNCI Cancer Spectr. 2 (2) (2018) pky028.
- [60] A.L. Palange, et al., Lipid-polymer nanoparticles encapsulating curcumin for modulating the vascular deposition of breast cancer cells, Nanomedicine 10 (5) (2014) 991–1002.
- [61] F.G. Giancotti, Mechanisms governing metastatic dormancy and reactivation, Cell 155 (4) (2013) 750–764.
- [62] A. Ferrer, et al., Hypoxia-mediated changes in bone marrow microenvironment in breast cancer dormancy, Cancer Lett. 488 (2020) 9–17.
- [63] L. Cao, et al., Curcumin inhibits hypoxia-induced epithelial-mesenchymal transition in pancreatic cancer cells via suppression of the hedgehog signaling pathway, Oncol. Rep. 35 (6) (2016) 3728–3734.
- [64] W. Li, et al., Curcumin inhibits pancreatic cancer cell invasion and EMT by interfering with tumor-stromal crosstalk under hypoxic conditions via the IL-6/ ERK/NF-κB axis, Oncol. Rep. 44 (1) (2020) 382–392.
- [65] A. Coker-Gurkan, et al., Curcumin prevented human autocrine growth hormone (GH) signaling mediated NF-κB activation and miR-183-96-182 cluster stimulated epithelial mesenchymal transition in T47D breast cancer cells, Mol. Biol. Rep. 46 (1) (2019) 355–369.
- [66] A. Coker-Gurkan, et al., Curcumin inhibits autocrine growth hormone-mediated invasion and metastasis by targeting NF-kB signaling and polyamine metabolism in breast cancer cells, Amino Acids 50 (8) (2018) 1045–1069.

- [67] C. Hu, et al., Anti-metastasis activity of curcumin against breast cancer via the inhibition of stem cell-like properties and EMT, Phytomedicine 58 (2018).
- [68] P.C. Leow, et al., Antitumor activity of natural compounds, curcumin and PKF118-310, as Wnt/β-catenin antagonists against human osteosarcoma cells, Invest N. Drugs 28 (6) (2010) 766–782.
- [69] J. Shi, et al., Curcumin inhibits bladder cancer progression via regulation of β-catenin expression, Tumour Biol. 39 (7) (2017), p. 1010428317702548.
- [70] K. Hyoudou, et al., SOD derivatives prevent metastatic tumor growth aggravated by tumor removal, Clin. Exp. Metastas-.-. 25 (5) (2008) 531–536.
- [71] Y. Yoshioka, et al., Anti-Cancer Effects of Dietary Polyphenols via ROS-Mediated Pathway with Their Modulation of MicroRNAs, Molecules 27 (12) (2022).
  [72] H. Wu, et al., Curcumin increases exosomal TCF21 thus suppressing exosome-
- induced lung cancer, Oncotarget 7 (52) (2016) 87081–87090. [73] S. Taverna, et al., Curcumin modulates chronic myelogenous leukemia exosomes
- (73) Stravena, et al., curcum modulates enounc injerogenous reaching exosonies composition and affects angiogenic phenotype via exosonal miR-21, Oncotarget 7 (21) (2016) 30420–30439.
- [74] H.G. Zhang, et al., Curcumin reverses breast tumor exosomes mediated immune suppression of NK cell tumor cytotoxicity, Biochim Biophys. Acta 1773 (7) (2007) 1116–1123.
- [75] Y. Wang, et al., Evaluating cell migration in vitro by the method based on cell patterning within microfluidic channels, Electrophoresis 33 (5) (2012) 773–779.
- [76] H. Zong, et al., Curcumin inhibits metastatic progression of breast cancer cell through suppression of urokinase-type plasminogen activator by NF-kappa B signaling pathways, Mol. Biol. Rep. 39 (4) (2012) 4803–4808.
- [77] M. Gallardo, G.M. Calaf, Curcumin inhibits invasive capabilities through epithelial mesenchymal transition in breast cancer cell lines, Int J. Oncol. 49 (3) (2016) 1019–1027.
- [78] M. Boonrao, et al., The inhibitory effect of turmeric curcuminoids on matrix metalloproteinase-3 secretion in human invasive breast carcinoma cells, Arch. Pharm. Res 33 (7) (2010) 989–998.
- [79] M. Li, et al., Curcumin inhibits the invasion and metastasis of triple negative breast cancer via Hedgehog/Gli1 signaling pathway, J. Ethnopharmacol. 283 (2022), 114689.
- [80] L.E. Wright, et al., Curcuminoids block TGF-β signaling in human breast cancer cells and limit osteolysis in a murine model of breast cancer bone metastasis, J. Nat. Prod. 76 (3) (2013) 316–321.
- [81] S.N.A. Harun, et al., The Molecular Targets and Anti-Invasive Effects of 2,6-bis-(4hydroxyl-3methoxybenzylidine) cyclohexanone or BHMC in MDA-MB-231 Human Breast Cancer Cells, Molecules 23 (4) (2018).
- [82] J.M. Kim, et al., Curcumin suppresses the TPA-induced invasion through inhibition of PKCα-dependent MMP-expression in MCF-7 human breast cancer cells, Phytomedicine 19 (12) (2012) 1085–1092.
- [83] S.K. Vemuri, et al., Modulatory Effects of Biosynthesized Gold Nanoparticles Conjugated with Curcumin and Paclitaxel on Tumorigenesis and Metastatic Pathways-In Vitro and In Vivo Studies, Int J. Mol. Sci. 23 (4) (2022).
- [84] K. Sun, et al., Curcumin inhibits LPA-induced invasion by attenuating RhoA/ ROCK/MMPs pathway in MCF7 breast cancer cells, Clin. Exp. Med 16 (1) (2016) 37–47.
- [85] T. Huang, Z. Chen, L. Fang, Curcumin inhibits LPS-induced EMT through downregulation of NF-κB-Snail signaling in breast cancer cells, Oncol. Rep. 29 (1) (2013) 117–124.
- [86] S. Mukherjee, et al., Curcumin inhibits breast cancer stem cell migration by amplifying the E-cadherin/β-catenin negative feedback loop, Stem Cell Res Ther. 5 (5) (2014) 116.
- [87] M.S. Charpentier, et al., Curcumin targets breast cancer stem-like cells with microtentacles that persist in mammospheres and promote reattachment, Cancer Res 74 (4) (2014) 1250–1260.
- [88] Q.Y. Chen, et al., Expression analysis of Cdc42 in lung cancer and modulation of its expression by curcumin in lung cancer cell lines, Int J. Oncol. 40 (5) (2012) 1561–1568.
- [89] L. Zhang, et al., Curcumin inhibits cell proliferation and migration in NSCLC through a synergistic effect on the TLR4/MyD88 and EGFR pathways, Oncol. Rep. 42 (5) (2019) 1843–1855.
- [90] S.S. Lin, et al., Curcumin inhibits the migration and invasion of human A549 lung cancer cells through the inhibition of matrix metalloproteinase-2 and -9 and Vascular Endothelial Growth Factor (VEGF), Cancer Lett. 285 (2) (2009) 127–133.
- [91] J.W. Zhan, et al., Integrated microRNA and gene expression profiling reveals the crucial miRNAs in curcumin anti-lung cancer cell invasion, Thorac. Cancer 8 (5) (2017) 461–470.
- [92] J.H. Kang, et al., Curcumin sensitizes human lung cancer cells to apoptosis and metastasis synergistically combined with carboplatin, Exp. Biol. Med (Maywood) 240 (11) (2015) 1416–1425.
- [93] P. Baharuddin, et al., Curcumin improves the efficacy of cisplatin by targeting cancer stem-like cells through p21 and cyclin D1-mediated tumour cell inhibition in non-small cell lung cancer cell lines, Oncol. Rep. 35 (1) (2016) 13–25.
- [94] J.H. Xu, et al., Role of Wnt Inhibitory Factor-1 in Inhibition of Bisdemethoxycurcumin Mediated Epithelial-to-Mesenchymal Transition in Highly Metastatic Lung Cancer 95D Cells, Chin. Med J. (Engl. ) 128 (10) (2015) 1376–1383.
- [95] J. Yang, et al., Curcumin inhibits the survival and metastasis of prostate cancer cells via the Notch-1 signaling pathway, Apmis 125 (2) (2017) 134–140.
- [96] M. He, et al., Re-purposing of curcumin as an anti-metastatic agent for the treatment of epithelial ovarian cancer: in vitro model using cancer stem cell enriched ovarian cancer spheroids, Oncotarget 7 (52) (2016) 86374–86387.

- [97] M. Kim, et al., The inhibitory effect of curcumin via fascin suppression through JAK/STAT3 pathway on metastasis and recurrence of ovary cancer cells, BMC Women'S. Health 20 (2020).
- [98] G. Calibasi-Kocal, et al., Curcumin effects on cell proliferation, angiogenesis and metastasis in colorectal cancer, J. B. U. .: Off. J. Balk. Union Oncol. 24 (2019) 1482–1487.
- [99] L. Xiang, et al., Antitumor effects of curcumin on the proliferation, migration and apoptosis of human colorectal carcinoma HCT-116 cells, Oncol. Rep. 44 (5) (2020) 1997–2008.
- [100] M. Shakibaei, et al., Curcumin potentiates antitumor activity of 5-fluorouracil in a 3D alginate tumor microenvironment of colorectal cancer, BMC Cancer 15 (1) (2015) 250.
- [101] S.A. Hosseini, H. Zand, M. Cheraghpour, The Influence of Curcumin on the Downregulation of MYC, Insulin and IGF-1 Receptors: A possible Mechanism Underlying the Anti-Growth and Anti-Migration in Chemoresistant Colorectal Cancer Cells, Med. (Kaunas.) 55 (4) (2019).
- [102] X. Wang, et al., Curcumin inhibits neurotensin-mediated interleukin-8 production and migration of HCT116 human colon cancer cells, Clin. Cancer Res 12 (18) (2006) 5346–5355.
- [103] G. Mudduluru, et al., Curcumin regulates miR-21 expression and inhibits invasion and metastasis in colorectal cancer, Biosci. Rep. 31 (3) (2011) 185–197.
- [104] C.L. Liao, et al., Bisdemethoxycurcumin Suppresses Migration and Invasion of Human Cervical Cancer HeLa Cells via Inhibition of NF-κB, MMP-2 and -9 Pathways, Anticancer Res 38 (7) (2018) 3989–3997.
- [105] L. Zhang, et al., Curcumin inhibits metastasis in human papillary thyroid carcinoma BCPAP cells via down-regulation of the TGF-β/Smad2/3 signaling pathway, Exp. Cell Res. 341 (2) (2016) 157–165.
- [106] C. Tan, et al., Curcumin inhibits hypoxia-induced migration in K1 papillary thyroid cancer cells, Exp. Biol. Med (Maywood) 240 (7) (2015) 925–935.
- [107] C.Y. Zhang, et al., Curcumin inhibits invasion and metastasis in K1 papillary thyroid cancer cells, Food Chem. 139 (1–4) (2013) 1021–1028.
- [108] C.Y. Zhang, et al., Curcumin inhibits the metastasis of K1 papillary thyroid cancer cells via modulating E-cadherin and matrix metalloproteinase-9 expression, Biotechnol. Lett. 35 (7) (2013) 995–1000.
- [109] L. Cao, et al., Curcumin inhibits H2O2-induced invasion and migration of human pancreatic cancer via suppression of the ERK/NF-kB pathway, Oncol. Rep. 36 (4) (2016) 2245–2251.
- [110] X.-D. Sun, X.-E. Liu, D.-S. Huang, Curcumin reverses the epithelial-mesenchymal transition of pancreatic cancer cells by inhibiting the Hedgehog signaling pathway, Oncol. Rep. 29 (6) (2013) 2401–2407.
- [111] M. Shao, et al., Curcumin and wikstroflavone B, a new biflavonoid isolated from Wikstroemia indica, synergistically suppress the proliferation and metastasis of nasopharyngeal carcinoma cells via blocking FAK/STAT3 signaling pathway, Phytomedicine 79 (2020), 153341.
- [112] P.S. de Campos, et al., Low doses of curcuma longa modulates cell migration and cell-cell adhesion, Phytother. Res 31 (9) (2017) 1433–1440.
- [113] A.Y. Lee, et al., Curcumin Inhibits Invasiveness and Epithelial-Mesenchymal Transition in Oral Squamous Cell Carcinoma Through Reducing Matrix Metalloproteinase 2, 9 and Modulating p53-E-Cadherin Pathway, Integr. Cancer Ther. 14 (5) (2015) 484–490.
- [114] L. Zhen, et al., Curcumin inhibits oral squamous cell carcinoma proliferation and invasion via EGFR signaling pathways, Int J. Clin. Exp. Pathol. 7 (10) (2014) 6438–6446.
- [115] H. Luo, et al., Curcumin loaded sub-30 nm targeting therapeutic lipid nanoparticles for synergistically blocking nasopharyngeal cancer growth and metastasis, J. Nanobiotechnol. 19 (1) (2021) 224.
- [116] B. Ren, et al., Curcumin inhibits liver cancer by inhibiting DAMP molecule HSP70 and TLR4 signaling, Oncol. Rep. 40 (2) (2018) 895–901.
- [117] S. Wang, et al., Curcumin inhibits the migration and invasion of mouse hepatoma Hca-F cells through down-regulating caveolin-1 expression and epidermal growth factor receptor signaling, IUBMB Life 63 (9) (2011) 775–782.
- [118] C.F. Tsai, et al., Curcumin Suppresses Phthalate-Induced Metastasis and the Proportion of Cancer Stem Cell (CSC)-like Cells via the Inhibition of AhR/ERK/ SK1 Signaling in Hepatocellular Carcinoma, J. Agric. Food Chem. 63 (48) (2015) 10388–10398.
- [119] H.J. Kim, et al., Curcumin suppresses migration and proliferation of Hep3B hepatocarcinoma cells through inhibition of the Wnt signaling pathway, Mol. Med Rep. 8 (1) (2013) 282–286.
- [120] Y. Ohashi, et al., Prevention of intrahepatic metastasis by curcumin in an orthotopic implantation model, Oncology 65 (3) (2003) 250–258.
- [121] S.Y. Wu, et al., Curcumin-induced heme oxygenase-1 expression plays a negative role for its anti-cancer effect in bladder cancers, Food Chem. Toxicol. 50 (10) (2012) 3530–3536.
- [122] J.H. Hong, et al., The effects of curcumin on the invasiveness of prostate cancer in vitro and in vivo, Prostate Cancer Prostatic Dis. 9 (2) (2006) 147–152.
- [123] J.G. Herman, H.L. Stadelman, C.E. Roselli, Curcumin blocks CCL2-induced adhesion, motility and invasion, in part, through down-regulation of CCL2 expression and proteolytic activity, Int J. Oncol. 34 (5) (2009) 1319–1327.
- [124] H.J. Lin, et al., Curcumin blocks migration and invasion of mouse-rat hybrid retina ganglion cells (N18) through the inhibition of MMP-2, -9, FAK, Rho A and Rock-1 gene expression, Oncol. Rep. 23 (3) (2010) 665–670.
- [125] H. Zhang, et al., Curcumin-Loaded Layered Double Hydroxide Nanoparticles-Induced Autophagy for Reducing Glioma Cell Migration and Invasion, J. Biomed. Nanotechnol. 12 (11) (2016) 2051–2062.

- [126] S. Philip, A. Bulbule, G.C. Kundu, Matrix metalloproteinase-2: mechanism and regulation of NF-kappaB-mediated activation and its role in cell motility and ECM-invasion, Glycoconj. J. 21 (8–9) (2004) 429–441.
- [127] L.G. Menon, R. Kuttan, G. Kuttan, Anti-metastatic activity of curcumin and catechin, Cancer Lett. 141 (1–2) (1999) 159–165.
- [128] S. Guerrero, et al., Curcumin-loaded nanoemulsion: a new safe and effective formulation to prevent tumor reincidence and metastasis, Nanoscale 10 (47) (2018) 22612–22622.
- [129] G. Loch-Neckel, et al., Orally Administered Chitosan-Coated Polycaprolactone Nanoparticles Containing Curcumin Attenuate Metastatic Melanoma in the Lungs, J. Pharm. Sci. 104 (10) (2015) 3524–3534.
- [130] G. Sharma, et al., Chemokines network in bone metastasis: Vital regulators of seeding and soiling, Semin. Cancer Biol. 86 (2022) 457–472.
- [131] S. Anupriya, A. Chakraborty, S. Patnaik, Clonal evolution and expansion associated with therapy resistance and relapse of Colorectal Cancer, Mutat. Res Rev. Mutat. Res (2022), 108445.
- [132] L.G. Menon, R. Kuttan, G. Kuttan, Inhibition of lung metastasis in mice induced by B16F10 melanoma cells by polyphenolic compounds, Cancer Lett. 95 (1–2) (1995) 221–225.
- [133] B.B. Aggarwal, et al., Curcumin suppresses the paclitaxel-induced nuclear factorkappaB pathway in breast cancer cells and inhibits lung metastasis of human breast cancer in nude mice, Clin. Cancer Res 11 (20) (2005) 7490–7498.
- [134] Y. Liu, et al., Hybrid curcumin-phospholipid complex-near-infrared dye oral drug delivery system to inhibit lung metastasis of breast cancer, Int J. Nanomed. 14 (2019) 3311–3330.
- [135] L. Liu, et al., Curcumin loaded polymeric micelles inhibit breast tumor growth and spontaneous pulmonary metastasis, Int. J. Pharm. 443 (1) (2013) 175–182.
- [136] S.I. Thamake, et al., Alendronate coated poly-lactic-co-glycolic acid (PLGA) nanoparticles for active targeting of metastatic breast cancer, Biomaterials 33 (29) (2012) 7164–7173.
- [137] L. Yan, Dietary supplementation with curcumin enhances metastatic growth of Lewis lung carcinoma in mice, Int J. Cancer 132 (2) (2013) 269–275.
- [138] W. Da, et al., Curcumin suppresses lymphatic vessel density in an in vivo human gastric cancer model, Tumour Biol. 36 (7) (2015) 5215–5223.
- [139] G. Zhu, et al., Curcumin inhibited the growth and invasion of human monocytic leukaemia SHI-1 cells in vivo by altering MAPK and MMP signalling, Pharm. Biol. 58 (1) (2020) 25–34.
- [140] N. Frank, et al., No prevention of liver and kidney tumors in Long-Evans Cinnamon rats by dietary curcumin, but inhibition at other sites and of metastases, Mutat. Res 523–524 (2003) 127–135.
- [141] M. Si, et al., Reversion effects of curcumin on multidrug resistance of MNNG/HOS human osteosarcoma cells in vitro and in vivo through regulation of Pglycoprotein, Chin. Med J. (Engl.) 126 (21) (2013) 4116–4123.
- [142] S. Prakobwong, et al., Curcumin decreases cholangiocarcinogenesis in hamsters by suppressing inflammation-mediated molecular events related to multistep carcinogenesis, Int J. Cancer 129 (1) (2011) 88–100.
- [143] Z. Liang, et al., Curcumin reverses tobacco smoke-induced epithelial-mesenchymal transition by suppressing the MAPK pathway in the lungs of mice, Mol. Med Rep. 17 (1) (2018) 2019–2025.
- [144] J. Biswas, et al., Curcumin protects DNA damage in a chronically arsenic-exposed population of West Bengal, Hum. Exp. Toxicol. 29 (6) (2010) 513–524.
- [145] A. Judaki, et al., Curcumin in combination with triple therapy regimes ameliorates oxidative stress and histopathologic changes in chronic gastritisassociated helicobacter pylori infection, Arq. Gastroenterol. 54 (3) (2017) 177–182.
- [146] G. Garcea, et al., Consumption of the putative chemopreventive agent curcumin by cancer patients: assessment of curcumin levels in the colorectum and their pharmacodynamic consequences, Cancer Epidemiol. Biomark. Prev. 14 (1) (2005) 120–125.
- [147] D. Liu, et al., Inhibition of curcumin on myeloid-derived suppressor cells is requisite for controlling lung cancer, Int Immunopharmacol. 39 (2016) 265–272.
- [148] S. Fan, et al., Opposite angiogenic outcome of curcumin against ischemia and Lewis lung cancer models: in silico, in vitro and in vivo studies, Biochim Biophys. Acta 1842 (9) (2014) 1742–1754.
- [149] L. Yan, J.A. Yee, J. Cao, Curcumin reduces trabecular and cortical bone in naive and lewis lung carcinoma-bearing mice, Anticancer Res 33 (8) (2013) 3153–3161.
- [150] Y. Xu, et al., Curcumin inhibits tumor proliferation induced by neutrophil elastase through the upregulation of  $\alpha$ 1-antitrypsin in lung cancer, Mol. Oncol. 6 (4) (2012) 405–417.
- [151] D. Yan, M.E. Geusz, R.J. Jamasbi, Properties of lewis lung carcinoma cells surviving curcumin toxicity, J. Cancer 3 (2012) 32–41.
- [152] K. Hassanzadeh, et al., Obstacles against the Marketing of Curcumin as a Drug, Int J. Mol. Sci. 21 (18) (2020).
- [153] J. Passildas-Jahanmohan, et al., Multicenter randomized phase II study comparing docetaxel plus curcumin versus docetaxel plus placebo in first-line treatment of metastatic castration-resistant prostate cancer, Cancer Med. 10 (7) (2021) 2332–2340.
- [154] B.T. Kurien, et al., Improving the solubility and pharmacological efficacy of curcumin by heat treatment, Assay. Drug Dev. Technol. 5 (4) (2007) 567–576.
- [155] G. Liang, et al., Exploration and synthesis of curcumin analogues with improved structural stability both in vitro and in vivo as cytotoxic agents, Bioorg. Med Chem. 17 (6) (2009) 2623–2631.
- [156] C.D. Lao, et al., Dose escalation of a curcuminoid formulation, BMC Complement Alter. Med 6 (2006) 10.
- [157] A.K. Singh, Engineered Nanoparticles: Structure, Properties and Mechanisms of Toxicity, Academic Press, 2015.

- [158] S. Hafez Ghoran, et al., Curcumin-based nanoformulations: a promising adjuvant towards cancer treatment, Molecules 27 (16) (2022).
- [159] N. Mor, N. Raghav, Alginate hydrogels: Sustained release system to analyze the effect of traditional excipients on curcumin availability, Bioorg. Chem. 107 (2021), 104513.
- [160] B. Zheng, et al., Loading natural emulsions with nutraceuticals using the pHdriven method: Formation & stability of curcumin-loaded soybean oils bodies, Food Funct. 10 (2019).
- [161] R. Vecchione, et al., Curcumin bioavailability from oil in water nano-emulsions: In vitro and in vivo study on the dimensional, compositional and interactional dependence, J. Control Release 233 (2016) 88–100.
- [162] R.K. Bhardwaj, et al., Piperine, a major constituent of black pepper, inhibits human P-glycoprotein and CYP3A4, J. Pharm. Exp. Ther. 302 (2) (2002) 645–650.
- [163] X. Zeng, et al., Selective reduction in the expression of UGTs and SULTs, a novel mechanism by which piperine enhances the bioavailability of curcumin in rat, Biopharm. Drug Dispos. 38 (1) (2017) 3–19.
- [164] V.M. Patil, S. Das, K. Balasubramanian, Quantum chemical and docking insights into bioavailability enhancement of curcumin by piperine in pepper, J. Phys. Chem. A 120 (20) (2016) 3643–3653.
- [165] L. Slika, et al., Preparation of curcumin-poly (allyl amine) hydrochloride based nanocapsules: Piperine in nanocapsules accelerates encapsulation and release of curcumin and effectiveness against colon cancer cells, Mater. Sci. Eng.: C. 109 (2020), 110550.
- [166] G. Shoba, et al., Influence of piperine on the pharmacokinetics of curcumin in animals and human volunteers, Planta Med 64 (4) (1998) 353–356.
- [167] P. Fança-Berthon, et al., Pharmacokinetics of a Single Dose of Turmeric Curcuminoids Depends on Formulation: Results of a Human Crossover Study, J. Nutr. 151 (7) (2021) 1802–1816.
- [168] C.H. Facina, et al., Protective effect of the association of curcumin with piperine on prostatic lesions: New perspectives on BPA-induced carcinogenesis, Food Chem. Toxicol. 158 (2021), 112700.
- [169] V. Patial, et al., Synergistic effect of curcumin and piperine in suppression of DENA-induced hepatocellular carcinoma in rats, Environ. Toxicol. Pharm. 40 (2) (2015) 445–452.
- [170] A. Sehgal, et al., Piperine as an adjuvant increases the efficacy of curcumin in mitigating benzo(a)pyrene toxicity, Hum. Exp. Toxicol. 31 (5) (2012) 473–482.
  [171] A. Sehgal, et al., Combined effects of curcumin and piperine in ameliorating
- [171] A. Sehgal, et al., Combined effects of curcumin and piperine in ameliorating benzo(a)pyrene induced DNA damage, Food Chem. Toxicol. 49 (11) (2011) 3002–3006.
- [172] T. Toyoda, et al., Anti-Inflammatory Effects of Capsaicin and Piperine on Helicobacter pylori-Induced Chronic Gastritis in Mongolian Gerbils, Helicobacter 21 (2) (2016) 131–142.
- [173] T. Eteraf-Oskouei, M. Najafi, Uses of Natural Honey in Cancer: An Updated Review, Adv. Pharm. Bull. 12 (2021).
- [174] M.A. Kamal, P. Klein, Determination of sugars in honey by liquid chromatography, Saudi J. Biol. Sci. 18 (1) (2011) 17–21.
- [175] J. Cuomo, et al., Comparative absorption of a standardized curcuminoid mixture and its lecithin formulation, J. Nat. Prod. 74 (4) (2011) 664–669.
- [176] A. Rajeshwari, et al., Spectroscopic studies on the interaction of bovine serum albumin with Al2O3 nanoparticles, J. Lumin. 145 (2014) 859–865.
- [177] B. Antony, et al., A Pilot Cross-Over Study to Evaluate Human Oral Bioavailability of BCM-95CG (Biocurcumax), A Novel Bioenhanced Preparation of Curcumin, Indian J. Pharm. Sci. 70 (4) (2008) 445–449.
- [178] R. Jäger, et al., Comparative absorption of curcumin formulations, Nutr. J. 13 (2014) 11.
- [179] S. Thanawala, et al., Comparative bioavailability of curcuminoids from a waterdispersible high curcuminoid turmeric extract against a generic turmeric extract: a randomized, cross-over, comparative, pharmacokinetic study, J. Pharm. Pharm. 73 (6) (2021) 816–823.
- [180] Gopi, S., et al. A Pilot Cross-Over Study to Assess theHuman Bio Availability of "Cureit" A BioAvailable Curcumin in Complete NaturalMatrix. 2015.
- [181] H. Chung, et al., Comparative pharmacokinetics of Theracurmin, a highly bioavailable curcumin, in healthy adult subjects, Int J. Clin. Pharm. Ther. 59 (10) (2021) 684–690.
- [182] S. Kothaplly, et al., Superior bioavailability of a novel curcumin formulation in healthy humans under fasting conditions, Adv. Ther. 39 (5) (2022) 2128–2138.
- [183] T.L.N. Gomes, et al., Evaluation of the impact of different doses of curcuma longa l. on antioxidant capacity: a randomized, double-blind, crossover pilot trial, Biomed. Res Int 2021 (2021) 3532864.
- [184] C. Schiborr, et al., The oral bioavailability of curcumin from micronized powder and liquid micelles is significantly increased in healthy humans and differs between sexes, Mol. Nutr. Food Res 58 (3) (2014) 516–527.
- [185] Y. Sunagawa, et al., Colloidal submicron-particle curcumin exhibits high absorption efficiency-a double-blind, 3-way crossover study, J. Nutr. Sci. Vitam. (Tokyo) 61 (1) (2015) 37–44.
- [186] S. Gopi, et al., Comparative Oral Absorption of Curcumin in a Natural Turmeric Matrix with Two Other Curcumin Formulations: An Open-label Parallel-arm Study, Phytother. Res 31 (12) (2017) 1883–1891.
- [187] A. Storka, et al., Safety, tolerability and pharmacokinetics of liposomal curcumin in healthy humans, Int J. Clin. Pharm. Ther. 53 (1) (2015) 54–65.
- [188] S.J. Stohs, et al., A Comparative Pharmacokinetic Assessment of a Novel Highly Bioavailable Curcumin Formulation with 95% Curcumin: A Randomized, Double-Blind, Crossover Study, J. Am. Coll. Nutr. 37 (1) (2018) 51–59.
- [189] D. Briskey, et al., Increased bioavailability of curcumin using a novel dispersion technology system (LipiSperse®), Eur. J. Nutr. 58 (5) (2019) 2087–2097.

- [190] M. Purpura, et al., Analysis of different innovative formulations of curcumin for improved relative oral bioavailability in human subjects, Eur. J. Nutr. 57 (3) (2018) 929–938.
- [191] T. Morimoto, et al., Drinkable preparation of Theracurmin exhibits high absorption efficiency–a single-dose, double-blind, 4-way crossover study, Biol. Pharm. Bull. 36 (11) (2013) 1708–1714.
- [192] R.E. Carroll, et al., Phase IIa clinical trial of curcumin for the prevention of colorectal neoplasia, Cancer Prev. Res (Philos.) 4 (3) (2011) 354–364.
- [193] M. Ávila-Gálvez, et al., Disposition of Dietary Polyphenols in Breast Cancer Patients' Tumors, and Their Associated Anticancer Activity: The Particular Case of Curcumin, Mol. Nutr. Food Res 65 (12) (2021), e2100163.
- [194] X. Duan, Y. Li, Physicochemical characteristics of nanoparticles affect circulation, biodistribution, cellular internalization, and trafficking, Small 9 (9–10) (2013) 1521–1532.
- [195] S.Y. Fam, et al., Stealth coating of nanoparticles in drug-delivery systems, Nanomater. (Basel) 10 (4) (2020).
- [196] S. Dützmann, et al., Intratumoral concentrations and effects of orally administered micellar curcuminoids in glioblastoma patients, Nutr. Cancer 68 (6) (2016) 943–948.
- [197] A. Adamczak, M. Ożarowski, T.M. Karpiński, Curcumin, a natural antimicrobial agent with strain-specific activity, Pharm. (Basel) 13 (7) (2020).
- [198] D.H. Zhou, et al., Combination of low concentration of (-)-epigallocatechin gallate (EGCG) and curcumin strongly suppresses the growth of non-small cell lung cancer in vitro and in vivo through causing cell cycle arrest, Int J. Mol. Sci. 14 (6) (2013) 12023–12036.
- [199] D. Choudhury, et al., Apigenin shows synergistic anticancer activity with curcumin by binding at different sites of tubulin, Biochimie 95 (6) (2013) 1297–1309.
- [200] S. Bhattacharjee, D.J. Brayden, Addressing the challenges to increase the efficiency of translating nanomedicine formulations to patients, Expert Opin. Drug Disco 16 (3) (2021) 235–254.
- [201] S. Zhou, et al., Unique flower-like Cur-metal complexes loaded liposomes for primary and metastatic breast cancer therapy, Mater. Sci. Eng. C. Mater. Biol. Appl. 121 (2021), 111835.
- [202] A.P. Ranjan, et al., Mitigating prolonged QT interval in cancer nanodrug development for accelerated clinical translation, J. Nanobiotechnology 11 (2013) 40.
- [203] A. Garrido, et al., hERG toxicity assessment: Useful guidelines for drug design, Eur. J. Med Chem. 195 (2020), 112290.
- [204] S. Bisht, et al., Systemic administration of polymeric nanoparticle-encapsulated curcumin (NanoCurc) blocks tumor growth and metastases in preclinical models of pancreatic cancer, Mol. Cancer Ther. 9 (8) (2010) 2255–2264.
- [205] A.S. Hoffman, The origins and evolution of "controlled" drug delivery systems, J. Control. Release 132 (3) (2008) 153–163.
- [206] R.R. Kotha, D.L. Luthria, Curcumin: Biological, Pharmaceutical, Nutraceutical, and Analytical Aspects, Molecules 24 (16) (2019).
- [207] Z. Li, et al., Gut microbiota dictate metabolic Fate of Curcumin in the colon, FASEB J. 31 (S1) (2017), 646.12-646.12.
- [208] Y. Lou, et al., Application of ultra-performance liquid chromatography coupled with quadrupole time-of-flight mass spectrometry to identify curcumin metabolites produced by human intestinal bacteria, J. Chromatogr. B Anal. Technol. Biomed. Life Sci. 985 (2015) 38–47.
- [209] A. Asai, T. Miyazawa, Occurrence of orally administered curcuminoid as glucuronide and glucuronide/sulfate conjugates in rat plasma, Life Sci. 67 (23) (2000) 2785–2793.
- [210] A. Pandey, et al., Reductive metabolites of curcumin and their therapeutic effects, Heliyon 6 (11) (2020), e05469.
- [211] W. Liu, et al., Tetrahydrocurcumin is more effective than curcumin in inducing the apoptosis of H22 cells via regulation of a mitochondrial apoptosis pathway in ascites tumor-bearing mice, Food Funct. 8 (9) (2017) 3120–3129.
- [212] C.S. Lai, et al., Tetrahydrocurcumin is more effective than curcumin in preventing azoxymethane-induced colon carcinogenesis, Mol. Nutr. Food Res 55 (12) (2011) 1819–1828.
- [213] Z.B. Zhang, et al., Curcumin's Metabolites, Tetrahydrocurcumin and Octahydrocurcumin, Possess Superior Anti-inflammatory Effects in vivo Through Suppression of TAK1-NF-kB Pathway, Front Pharm. 9 (2018) 1181.
- [214] B.B. Aggarwal, L. Deb, S. Prasad, Curcumin differs from tetrahydrocurcumin for molecular targets, signaling pathways and cellular responses, Molecules 20 (1) (2014) 185–205.
- [215] Y.I. Hahn, et al., Curcumin interacts directly with the Cysteine 259 residue of STAT3 and induces apoptosis in H-Ras transformed human mammary epithelial cells, Sci. Rep. 8 (1) (2018) 6409.
- [216] M.E.M. Saeed, et al., In Silico and In Vitro Screening of 50 Curcumin Compounds as EGFR and NF-κB Inhibitors, Int J. Mol. Sci. 23 (7) (2022).
- [217] Q.U.A. Jamil, et al., Metabolism of curcumin in human breast cancer cells: impact of sulfation on cytotoxicity, Planta Med 83 (12–13) (2017) 1028–1034.
- [218] H.S. Anbar, Z Isa, J.J. Elounais, M.A. Jameel, J.H. Zib, A.M. Samer, A.F. Jawad, M.I. El-Gamal, Steroid sulfatase inhibitors: the current landscape, Expert Opin Ther Pat 31 (6) (2021) 453–472.
- [219] P.B. Luis, et al., Incomplete hydrolysis of curcumin conjugates by β-glucuronidase: detection of complex conjugates in plasma, Mol. Nutr. Food Res 64 (6) (2020), e1901037.
- [220] P. Subramanian, Lipid-based nanocarrier system for the effective delivery of nutraceuticals, Molecules 26 (18) (2021).
- [221] W. Zam, Gut microbiota as a prospective therapeutic target for curcumin: a review of mutual influence, J. Nutr. Metab. 2018 (2018) 1367984.

- [222] R. Pluta, S. Januszewski, M. Ułamek-Kozioł, Mutual two-way interactions of curcumin and gut microbiota, Int. J. Mol. Sci. 21 (3) (2020).
- [223] L. Shen, L. Liu, H.F. Ji, Regulative effects of curcumin spice administration on gut microbiota and its pharmacological implications, Food Nutr. Res 61 (1) (2017) 1361780.
- [224] M. Borges-Canha, et al., Role of colonic microbiota in colorectal carcinogenesis: a systematic review, Rev. Esp. Enferm. Dig. 107 (11) (2015) 659–671.
- [225] S. Saadati, et al., Curcumin and inflammation in non-alcoholic fatty liver disease: a randomized, placebo controlled clinical trial, BMC Gastroenterol. 19 (1) (2019) 133.
- [226] N. Dhillon, et al., Phase II trial of curcumin in patients with advanced pancreatic cancer, Clin. Cancer Res 14 (14) (2008) 4491–4499.
- [227] R.A. Sharma, et al., Phase I clinical trial of oral curcumin: biomarkers of systemic activity and compliance, Clin. Cancer Res 10 (20) (2004) 6847–6854.
- [228] C. Loira-Pastoriza, J. Todoroff, R. Vanbever, Delivery strategies for sustained drug release in the lungs, Adv. Drug Deliv. Rev. 75 (2014) 81–91.
- [229] T. Chen, et al., Inhaled curcumin mesoporous polydopamine nanoparticles against radiation pneumonitis, Acta Pharm. Sin. B 12 (5) (2022) 2522–2532.
   [230] W.H. Lee, et al., Development and evaluation of paclitaxel and curcumin dry
- powder for inhalation lung cancer treatment, Pharmaceutics 13 (1) (2020).
- [231] D.A. Edwards, et al., Large porous particles for pulmonary drug delivery, Science 276 (5320) (1997) 1868–1872.
- [232] I.M. Adel, et al., Design and characterization of spray-dried proliposomes for the pulmonary delivery of curcumin, Int J. Nanomed. 16 (2021) 2667–2687.
- [233] J.S. Patton, C.S. Fishburn, J.G. Weers, The lungs as a portal of entry for systemic drug delivery, Proc. Am. Thorac. Soc. 1 (4) (2004) 338–344.
- [234] D.M. Silva, et al., Design of polymeric core-shell carriers for combination therapies, J. Colloid Interface Sci. 587 (2021) 499–509.
- [235] K. Jyoti, et al., Inhalable bioresponsive chitosan microspheres of doxorubicin and soluble curcumin augmented drug delivery in lung cancer cells, Int J. Biol. Macromol. 98 (2017) 50–58.
- [236] D. Gaikwad, et al., Enhancement in in vitro anti-angiogenesis activity and cytotoxicity in lung cancer cell by pectin-PVP based curcumin particulates, Int J. Biol. Macromol. 104 (Pt A) (2017) 656–664.
- [237] W. Su, et al., Treatment of metastatic lung cancer via inhalation administration of curcumin composite particles based on mesoporous silica, Eur. J. Pharm. Sci. 134 (2019) 246–255.
- [238] Y. Hu, et al., Comparative anti-inflammatory effect of curcumin at air-liquid interface and submerged conditions using lipopolysaccharide stimulated human lung epithelial A549 cells, Pulm. Pharm. Ther. 63 (2020), 101939.
- [239] T. Zhang, et al., Inhalation treatment of primary lung cancer using liposomal curcumin dry powder inhalers, Acta Pharm. Sin. B 8 (3) (2018) 440–448.
- [240] T. Fischer, et al., Cylindrical Microparticles Composed of Mesoporous Silica Nanoparticles for the Targeted Delivery of a Small Molecule and a Macromolecular Drug to the Lungs: Exemplified with Curcumin and siRNA, Pharmaceutics 13 (2021) 844.
- [241] G. Kim, et al., Engineering exosomes for pulmonary delivery of peptides and drugs to inflammatory lung cells by inhalation, J. Control. Release 330 (2021) 684–695.
- [242] W.B. COLEY, The treatment of inoperable sarcoma with the mixed toxins of erysipelas and bacillus prodigiosus.: immediate and final results in one hundred and forty cases, J. Am. Med. Assoc. XXXI (8) (1898) 389–395.
- [243] W.B. Coley, The treatment of inoperable sarcoma by bacterial toxins (the mixed toxins of the Streptococcus erysipelas and the Bacillus prodigiosus), Proc. R. Soc. Med. 3 (Surg Sect) (1910) 1–48.
- [244] A. Huang, et al., Human intratumoral therapy: Linking drug properties and tumor transport of drugs in clinical trials, J. Control Release 326 (2020) 203–221.
- [245] M. Gao, et al., Preparation and characterization of curcumin thermosensitive hydrogels for intratumoral injection treatment, Drug Dev. Ind. Pharm. 40 (11) (2014) 1557–1564.
- [246] W. Zhang, et al., Improving Anti-Tumor Activity of Curcumin by Polymeric Micelles in Thermosensitive Hydrogel System in Colorectal Peritoneal Carcinomatosis Model, J. Biomed. Nanotechnol. 11 (7) (2015) 1173–1182.
- [247] H. Cao, et al., Dual-loaded, long-term sustained drug releasing and thixotropic hydrogel for localized chemotherapy of cancer, Biomater. Sci. 7 (7) (2019) 2975–2985.
- [248] M. Orunoğlu, et al., Effects of curcumin-loaded PLGA nanoparticles on the RG2 rat glioma model, Mater. Sci. Eng. C. Mater. Biol. Appl. 78 (2017) 32–38.
- [249] L. Piao, et al., TriCurin, a novel formulation of curcumin, epicatechin gallate, and resveratrol, inhibits the tumorigenicity of human papillomavirus-positive head and neck squamous cell carcinoma, Oncotarget 8 (36) (2017) 60025–60035.
- [250] M. Chang, M. Wu, H. Li, Antitumor Effects of Curcumin and Glycyrrhetinic Acid-Modified Curcumin-Loaded Cationic Liposome by Intratumoral Administration, Evid. Based Complement Altern. Med 2020 (2020) 4504936.
- [251] J. Gao, et al., PEGylated lipid bilayer coated mesoporous silica nanoparticles codelivery of paclitaxel and curcumin leads to increased tumor site drug accumulation and reduced tumor burden, Eur. J. Pharm. Sci. 140 (2019), 105070.
- [252] S. Mukherjee, et al., Liposomal TriCurin, A Synergistic Combination of Curcumin, Epicatechin Gallate and Resveratrol, Repolarizes Tumor-Associated Microglia/ Macrophages, and Eliminates Glioblastoma (GBM) and GBM Stem Cells, Molecules 23 (1) (2018).
- [253] M. Chang, M. Wu, H. Li, Antitumor activities of novel glycyrrhetinic acidmodified curcumin-loaded cationic liposomes in vitro and in H22 tumor-bearing mice, Drug Deliv. 25 (1) (2018) 1984–1995.
- [254] M. Shahriari, et al., Anticancer potential of curcumin-cyclodextrin complexes and their pharmacokinetic properties, Int J. Pharm. 631 (2023), 122474.

- [255] K.Z. Sanidad, et al., Curcumin: recent advances in the development of strategies to improve oral bioavailability, Annu Rev. Food Sci. Technol. 10 (2019) 597–617.
- [256] E.M. Merisko-Liversidge, G.G. Liversidge, Drug nanoparticles: formulating poorly water-soluble compounds, Toxicol. Pathol. 36 (1) (2008) 43–48.
- [257] P. Zhu, et al., Curcumin attenuates hypoxia/reoxygenation-induced cardiomyocyte injury by downregulating Notch signaling, Mol. Med Rep. 20 (2) (2019) 1541–1550.
- [258] M. Cruz-Correa, et al., Combination treatment with curcumin and quercetin of adenomas in familial adenomatous polyposis, Clin. Gastroenterol. Hepatol. 4 (8) (2006) 1035–1038.
- [259] M. Saberi-Karimian, et al., Effects of curcuminoids on inflammatory status in patients with non-alcoholic fatty liver disease: A randomized controlled trial, Complement Ther. Med 49 (2020), 102322.
- [260] F.F. Cox, et al., Protective effects of curcumin in cardiovascular diseases-impact on oxidative stress and mitochondria, Cells 11 (3) (2022).
- [261] R. Greil, et al., A phase 1 dose-escalation study on the safety, tolerability and activity of liposomal curcumin (Lipocurc(<sup>TM</sup>)) in patients with locally advanced or metastatic cancer, Cancer Chemother. Pharm. 82 (4) (2018) 695–706.
- [262] V. Lopes-Rodrigues, E. Sousa, M.H. Vasconcelos, Curcumin as a Modulator of P-Glycoprotein in Cancer: Challenges and Perspectives, Pharm. (Basel) 9 (4) (2016).
- [263] S.H. Ahmad Shariff, et al., Poly(caprolactone)-b-poly(ethylene glycol)-Based Polymeric Micelles as Drug Carriers for Efficient Breast Cancer Therapy: A Systematic Review, Polym. (Basel) 14 (22) (2022).
- [264] A.A. Attama, et al., Nanogels as target drug delivery systems in cancer therapy: A review of the last decade, Front Pharm. 13 (2022), 874510.
- [265] V. Kral, et al., Nanomedicine Current status and perspectives: A big potential or just a catchword? Chem. Listy 100 (1) (2006) 4–9.
- [266] A. Karthic, et al., Cyclodextrin nanoparticles for diagnosis and potential cancer therapy: A systematic review, Front Cell Dev. Biol. 10 (2022), 984311.
- [267] J.Q. Zhu, et al., Responsive hydrogels based on triggered click reactions for liver cancer, Adv. Mater. 34 (38) (2022), e2201651.
- [268] X. Zhang, et al., Hydrazinocurcumin Encapsuled nanoparticles "re-educate" tumor-associated macrophages and exhibit anti-tumor effects on breast cancer following STAT3 suppression, PLoS One 8 (6) (2013), e65896.
- [269] K. Muthoosamy, et al., Exceedingly Higher co-loading of Curcumin and Paclitaxel onto Polymer-functionalized Reduced Graphene Oxide for Highly Potent Synergistic Anticancer Treatment, Sci. Rep. 6 (1) (2016) 32808.
- [270] S. Khan, et al., Superparamagnetic iron oxide nanoparticles of curcumin enhance gemcitabine therapeutic response in pancreatic cancer, Biomaterials 208 (2019) 83–97.
- [271] S. Yu, et al., Hyaluronic acid coating on the surface of curcumin-loaded ZIF-8 nanoparticles for improved breast cancer therapy: An in vitro and in vivo study, Colloids Surf. B Biointerfaces 203 (2021), 111759.
- [272] V.S. Madamsetty, et al., Development of multi-drug loaded PEGylated nanodiamonds to inhibit tumor growth and metastasis in genetically engineered mouse models of pancreatic cancer, Nanoscale 11 (45) (2019) 22006–22018.
- [273] D. Liao, et al., Synthetic enzyme inhibitor: a novel targeting ligand for nanotherapeutic drug delivery inhibiting tumor growth without systemic toxicity, Nanomed.: Nanotechnol., Biol. Med. 7 (6) (2011) 665–673.
- [274] Y. Sun, et al., Cell Permeable NBD Peptide-Modified Liposomes by Hyaluronic Acid Coating for the Synergistic Targeted Therapy of Metastatic Inflammatory Breast Cancer, Mol. Pharm. 16 (3) (2019) 1140–1155.
- [275] J. Liu, et al., Efficiency of Different Treatment Regimens Combining Anti-tumor and Anti-inflammatory Liposomes for Metastatic Breast Cancer, AAPS PharmSciTech 21 (7) (2020) 259.
- [276] C. Qi, et al., Co-Delivery of Curcumin and Capsaicin by Dual-Targeting Liposomes for Inhibition of aHSC-Induced Drug Resistance and Metastasis, ACS Appl. Mater. Interfaces 13 (14) (2021) 16019–16035.
- [277] F. Wang, et al., Dual-Ligand-Modified Liposomes Co-Loaded with Anti-Angiogenic and Chemotherapeutic Drugs for Inhibiting Tumor Angiogenesis and Metastasis, Int J. Nanomed. 16 (2021) 4001–4016.
- [278] S. Barui, et al., Simultaneous delivery of doxorubicin and curcumin encapsulated in liposomes of pegylated RGDK-lipopeptide to tumor vasculature, Biomaterials 35 (2013).
- [279] Q. An, et al., Development and characterization of octreotide-modified curcumin plus docetaxel micelles for potential treatment of non-small-cell lung cancer, Pharm. Dev. Technol. 24 (9) (2019) 1164–1174.
- [280] R. Mardani, et al., Nanomicellar-curcumin exerts its therapeutic effects via affecting angiogenesis, apoptosis, and T cells in a mouse model of melanoma lung metastasis, Pathol. Res Pr. 216 (9) (2020), 153082.
- [281] Y. Zhou, et al., Multi pH-sensitive polymer-drug conjugate mixed micelles for efficient co-delivery of doxorubicin and curcumin to synergistically suppress tumor metastasis, Biomater. Sci. 8 (18) (2020) 5029–5046.
- [282] S. Bisht, et al., Polymeric nanoparticle-encapsulated curcumin ("nanocurcumin"): a novel strategy for human cancer therapy, J. Nanobiotechnol. 5 (1) (2007) 3.
- [283] B. Hu, et al., A polymeric nanoparticle formulation of curcumin in combination with sorafenib synergistically inhibits tumor growth and metastasis in an orthotopic model of human hepatocellular carcinoma, Biochem Biophys. Res Commun. 468 (4) (2015) 525–532.
- [284] A. Sharma, et al., Effects of curcumin-loaded poly(lactic-co-glycolic acid) nanoparticles in MDA-MB231 human breast cancer cells, Nanomed. (Lond. ) 16 (20) (2021) 1763–1773.
- [285] F. Bessone, et al., Low-dose curcuminoid-loaded in dextran nanobubbles can prevent metastatic spreading in prostate cancer cells, Nanotechnology 30 (21) (2019), 214004.

- [286] B. Kim, et al., Albumin nanoparticles with synergistic antitumor efficacy against metastatic lung cancers, Colloids Surf. B Biointerfaces 158 (2017) 157–166.
- [287] Y. Huang, et al., Quadruple-responsive nanoparticle-mediated targeted combination chemotherapy for metastatic breast cancer, Nanoscale 13 (11) (2021) 5765–5779.
- [288] R. Fan, et al., Dual drug loaded biodegradable nanofibrous microsphere for improving anti-colon cancer activity, Sci. Rep. 6 (1) (2016) 28373.
- [289] Y.Y. Lin, et al., Cell-derived artificial nanovesicle as a drug delivery system for malignant melanoma treatment, Biomed. Pharm. 147 (2022), 112586.
- [290] M. Xu, et al., Mesenchymal stem cells-curcumin loaded chitosan nanoparticles hybrid vectors for tumor-tropic therapy, Int J. Biol. Macromol. 134 (2019) 1002–1012.
- [291] R. Deng, et al., Effects of helicobacter pylori on tumor microenvironment and immunotherapy responses, Front Immunol. 13 (2022), 923477.
- [292] J.G. Navashenaq, et al., The interaction of Helicobacter pylori with cancer immunomodulatory stromal cells: New insight into gastric cancer pathogenesis, Semin. Cancer Biol. 86 (2022) 951–959.
- [293] R. De, et al., Antimicrobial activity of curcumin against helicobacter pylori isolates from india and during infections in mice, Antimicrob. Agents Chemother. 53 (4) (2009) 1592–1597.
- [294] G.B. Mahady, et al., Turmeric (Curcuma longa) and curcumin inhibit the growth of Helicobacter pylori, a group 1 carcinogen, Anticancer Res. 22 (6C) (2002) 4179–4181.
- [295] Y. Xie, et al., Microbiota in Tumors: From Understanding to Application, Adv. Sci. (2022) 2200470 (n/a(n/a)).
- [296] A. Gagnaire, et al., Collateral damage: insights into bacterial mechanisms that predispose host cells to cancer, Nat. Rev. Microbiol. 15 (2) (2017) 109–128.
- [297] P. Łaniewski, Z.E. Ilhan, M.M. Herbst-Kralovetz, The microbiome and gynaecological cancer development, prevention and therapy, Nat. Rev. Urol. 17 (4) (2020) 232–250.
- [298] D.F. Squarzanti, et al., Non-Melanoma Skin Cancer: news from microbiota research, Crit. Rev. Microbiol 46 (4) (2020) 433–449.
- [299] Y. Hussain, et al., Antimicrobial potential of curcumin: therapeutic potential and challenges to clinical applications, Antibiot. (Basel) 11 (3) (2022).
- [300] A. Tzeng, et al., Human breast microbiome correlates with prognostic features and immunological signatures in breast cancer, Genome Med 13 (1) (2021) 60.
- [301] I.A. Packiavathy, et al., Inhibition of biofilm development of uropathogens by curcumin - an anti-quorum sensing agent from Curcuma longa, Food Chem. 148 (2014) 453–460.
- [302] H. Gunes, et al., Antibacterial effects of curcumin: An in vitro minimum inhibitory concentration study, Toxicol. Ind. Health 32 (2) (2016) 246–250.
- [303] T. Rudrappa, H.P. Bais, Curcumin, a known phenolic from Curcuma longa, attenuates the virulence of Pseudomonas aeruginosa PAO1 in whole plant and animal pathogenicity models, J. Agric. Food Chem. 56 (6) (2008) 1955–1962.
- [304] N. Negi, et al., Possible Role of Curcumin as an Efflux Pump Inhibitor in Multi Drug Resistant Clinical Isolates of Pseudomonas aeruginosa, J. Clin. Diagn. Res 8 (10) (2014) Dc04–Dc07.
- [305] A. Kali, et al., Antibacterial synergy of curcumin with antibiotics against biofilm producing clinical bacterial isolates, J. Basic Clin. Pharm. 7 (3) (2016) 93–96.
- [306] S. Bahari, et al., Inhibition of Pseudomonas aeruginosa quorum sensing by subinhibitory concentrations of curcumin with gentamicin and azithromycin, J. Glob. Antimicrob. Resist 10 (2017) 21–28.
- [307] S. Gao, et al., Presence of Porphyromonas gingivalis in esophagus and its association with the clinicopathological characteristics and survival in patients with esophageal cancer, Infect. Agent Cancer 11 (2016) 3.
- [308] V.M. Kumbar, et al., Effect of curcumin on growth, biofilm formation and virulence factor gene expression of Porphyromonas gingivalis, Odontology 109 (1) (2021) 18–28.
- [309] S. Parida, et al., A Procarcinogenic Colon Microbe Promotes Breast Tumorigenesis and Metastatic Progression and Concomitantly Activates Notch and β-Catenin Axes, Cancer Disco 11 (5) (2021) 1138–1157.
- [310] D. Mody, A.I.M. Athamneh, M.N. Seleem, Curcumin: A natural derivative with antibacterial activity against Clostridium difficile, J. Glob. Antimicrob. Resist 21 (2020) 154–161.
- [311] W. Liu, et al., Microbial community heterogeneity within colorectal neoplasia and its correlation with colorectal carcinogenesis, Gastroenterology 160 (7) (2021) 2395–2408.
- [312] S. Izui, et al., Antibacterial activity of curcumin against periodontopathic bacteria, J. Periodo 87 (1) (2016) 83–90.
- [313] L. Parhi, et al., Breast cancer colonization by Fusobacterium nucleatum accelerates tumor growth and metastatic progression, Nat. Commun. 11 (1) (2020) 3259.
- [314] A.D. Kostic, et al., Fusobacterium nucleatum potentiates intestinal tumorigenesis and modulates the tumor-immune microenvironment, Cell Host Microbe 14 (2) (2013) 207–215.
- [315] S. Yachida, et al., Metagenomic and metabolomic analyses reveal distinct stagespecific phenotypes of the gut microbiota in colorectal cancer, Nat. Med 25 (6) (2019) 968–976.
- [316] A. Bertocchi, et al., Gut vascular barrier impairment leads to intestinal bacteria dissemination and colorectal cancer metastasis to liver, Cancer Cell 39 (5) (2021) 708–724, e11.
- [317] O.S. Adeyemi, et al., Altered redox status, DNA damage and modulation of Ltryptophan metabolism contribute to antimicrobial action of curcumin, Heliyon 6 (3) (2020), e03495.
- [318] J.J. Tsay, et al., Airway microbiota is associated with upregulation of the PI3K pathway in lung cancer, Am. J. Respir. Crit. Care Med 198 (9) (2018) 1188–1198.

- [319] P. Hu, P. Huang, M.W. Chen, Curcumin reduces Streptococcus mutans biofilm formation by inhibiting sortase A activity, Arch. Oral. Biol. 58 (10) (2013) 1343–1348.
- [320] J. Song, et al., Curcumin suppresses Streptococcus mutans adherence to human tooth surfaces and extracellular matrix proteins, Eur. J. Clin. Microbiol Infect. Dis. 31 (7) (2012) 1347–1352.
- [321] X. Fan, et al., Human oral microbiome and prospective risk for pancreatic cancer: a population-based nested case-control study, Gut 67 (1) (2018) 120–127.
- [322] S. Mallikarjun, C. Dawra, K. Bhat, An in vitro Evaluation of Antibacterial Activity of Curcumin against Aggregatibacter actinomycetemcomitans, World J. Dent. 6 (2015) 16–19.
- [323] B. Zhou, et al., The biodiversity Composition of Microbiome in Ovarian Carcinoma Patients, Sci. Rep. 9 (1) (2019) 1691.
- [324] J.W. Betts, D.W. Wareham, In vitro activity of curcumin in combination with epigallocatechin gallate (EGCG) versus multidrug-resistant Acinetobacter baumannii, BMC Microbiol 14 (2014) 172.
- [325] S. Banerjee, et al., The ovarian cancer oncobiome, Oncotarget 8 (22) (2017) 36225–36245.
- [326] A. Boeder, et al., Anti-mycoplasma activity of Curcuma longa extracts and your isolated compound, the curcumin, Rev. Fitos 12 (2018).
- [327] J. Kullander, O. Forslund, J. Dillner, Staphylococcus aureus and squamous cell carcinoma of the skin, Cancer Epidemiol. Biomark. Prev. 18 (2) (2009) 472–478.
- [328] S. Tajbakhsh, et al., Antibacterial activity of indium curcumin and indium diacetylcurcumin, Afr. J. Biotechnol. 7 (2008) 3832–3835.
- [329] A.K. Singh, et al., Quantum curcumin mediated inhibition of gingipains and mixed-biofilm of Porphyromonas gingivalis causing chronic periodontitis, RSC Adv. 8 (70) (2018) 40426–40445.
- [330] A. Adeluola, et al., Perspectives for synthetic curcumins in chemoprevention and treatment of cancer: An update with promising analogues, Eur. J. Pharm. 906 (2021), 174266.
- [331] F.C. Rodrigues, N.A. Kumar, G. Thakur, The potency of heterocyclic curcumin analogues: An evidence-based review, Pharm. Res 166 (2021), 105489.
- [332] J. Moreira, et al., Diarylpentanoids with antitumor activity: A critical review of structure-activity relationship studies, Eur. J. Med Chem. 192 (2020), 112177.
- [333] R. Jagannathan, P.M. Abraham, P. Poddar, Temperature-Dependent Spectroscopic Evidences of Curcumin in Aqueous Medium: A Mechanistic Study of Its Solubility and Stability, J. Phys. Chem. B 116 (50) (2012) 14533–14540.
- [334] T. Prestera, et al., Parallel induction of heme oxygenase-1 and chemoprotective phase 2 enzymes by electrophiles and antioxidants: regulation by upstream antioxidant-responsive elements (ARE), Mol. Med 1 (7) (1995) 827–837.
- [335] G.S. Jeong, et al., Comparative effects of curcuminoids on endothelial heme oxygenase-1 expression: ortho-methoxy groups are essential to enhance heme oxygenase activity and protection, Exp. Mol. Med 38 (4) (2006) 393–400.
- [336] H. Hu, H. Zhou, D. Xu, A review of the effects and molecular mechanisms of dimethylcurcumin (ASC-J9) on androgen receptor-related diseases, Chem. Biol. Drug Des. 97 (4) (2021) 821–835.
- [337] H. Zhang, et al., Androgen Metabolism and Response in Prostate Cancer Anti-Androgen Therapy Resistance, Int J. Mol. Sci. 23 (21) (2022).
- [338] R. Wang, et al., ASC-J9® suppresses castration resistant prostate cancer progression via degrading the enzalutamide-induced androgen receptor mutant AR-F876L, Cancer Lett. 379 (1) (2016) 154–160.
- [339] K. Nakamura, et al., Curcumin down-regulates AR gene expression and activation in prostate cancer cell lines, Int J. Oncol. 21 (4) (2002) 825–830.

- [340] P. Zhang, et al., MicroRNA-33b, upregulated by EF24, a curcumin analog, suppresses the epithelial-to-mesenchymal transition (EMT) and migratory potential of melanoma cells by targeting HMGA2, Toxicol. Lett. 234 (3) (2015) 151–161.
- [341] B. Bao, et al., Hypoxia-induced aggressiveness of pancreatic cancer cells is due to increased expression of VEGF, IL-6 and miR-21, which can be attenuated by CDF treatment, PLoS One 7 (12) (2012), e50165.
- [342] S. Das, et al., Novel 3,5-bis(arylidene)-4-piperidone dimers: potent cytotoxins against colon cancer cells, Eur. J. Med Chem. 64 (2013) 321–328.
- [343] A. Jha, et al., Curcumin-inspired cytotoxic 3,5-bis(arylmethylene)-1-(N-(orthosubstituted aryl)maleamoyl)-4-piperidones: A novel group of topoisomerase II alpha inhibitors, Bioorg. Med. Chem. 23 (2015).
- [344] H.A. Al-Howail, et al., PAC down-regulates estrogen receptor alpha and suppresses epithelial-to-mesenchymal transition in breast cancer cells, BMC Cancer 16 (1) (2016) 540.
- [345] E.M. Al-Hujaily, et al., PAC, a novel curcumin analogue, has anti-breast cancer properties with higher efficiency on ER-negative cells, Breast Cancer Res Treat. 128 (1) (2011) 97–107.
- [346] G.P. Nagaraju, et al., Antiangiogenic effects of a novel synthetic curcumin analogue in pancreatic cancer, Cancer Lett. 357 (2) (2015) 557–565.
- [347] B. Rajitha, et al., Inhibition of NF-kB translocation by curcumin analogs induces G0/G1 arrest and downregulates thymidylate synthase in colorectal cancer, Cancer Lett. 373 (2) (2016) 227–233.
- [348] M. Shoji, et al., Inhibition of breast cancer metastasis to the lungs with UBS109, Oncotarget 9 (90) (2018) 36102–36109.
- [349] M. Yamaguchi, et al., Curcumin analogue UBS109 prevents bone loss in breast cancer bone metastasis mouse model: involvement in osteoblastogenesis and osteoclastogenesis, Cell Tissue Res 357 (1) (2014) 245–252.
- [350] B. Zebib, Z. Mouloungui, V. Noirot, Stabilization of Curcumin by Complexation with Divalent Cations in Glycerol/Water System, Bioinorg. Chem. Appl. 2010 (2010), 292760.
- [351] S. Prasad, et al., Metal–Curcumin Complexes in Therapeutics: An Approach to Enhance Pharmacological Effects of Curcumin, Int. J. Mol. Sci. 22 (2021), https:// doi.org/10.3390/ijms22137094.
- [352] S. Banerjee, A.R. Chakravarty, Metal complexes of curcumin for cellular imaging, targeting, and photoinduced anticancer activity, Acc. Chem. Res 48 (7) (2015) 2075–2083.
- [353] N. Pagliaricci, et al., Potent and selective anticancer activity of half-sandwich ruthenium and osmium complexes with modified curcuminoid ligands, Dalton Trans. 51 (2022).
- [354] R. Pettinari, et al., Ruthenium(II)–Arene RAPTA type complexes containing curcumin and bisdemethoxycurcumin display potent and selective anticancer activity, Organometallics 33 (14) (2014) 3709–3715.
- [355] B. Mohan, et al., Design, Synthesis, and Anticancer Studies of a p-Cymene-Ru(II)-Curcumin Organometallic Conjugate Based on a Fluorescent 4-Amino-1,8-naphthalimide Tröger's Base Scaffold, Inorg. Chem. 61 (30) (2022) 11592–11599.
- [356] J. Leskovská, et al., Antiproliferative Ruthenium Complexes Containing Curcuminoid Ligands Tested In Vitro on Human Ovarian Tumor Cell Line A2780, towards Their Capability to Modulate the NF-kBTranscription Factor, FGF-2 Growth Factor, and MMP-9 Pathway, Molecules 27 (2022), https://doi.org/ 10.3390/molecules27144565.
- [357] P. Srivastava, et al., Rationally designed curcumin based ruthenium(ii) antimicrobials effective against drug-resistant Staphylococcus aureus, Dalton Trans. 48 (31) (2019) 11822–11828.